Cosserat elasticity, spectral theory of first order systems, and the massless Dirac operator

A thesis submitted for the degree of Doctor of Philosophy

by

Robert James Downes

Department of Mathematics UCL January 2014

Declaration

I, Robert James Downes, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Abstract

This thesis is concerned with the study of the massless Dirac operator in dimension three and is, in part, based upon [12, 22, 21, 26, 25].

An introduction is given in Chapter 1.

In Chapter 2 we study a special version of Cosserat elasticity, with deformations induced by rotations only, and no displacements. We prove that for a particular choice of elastic moduli and in the stationary setting (harmonic dependence on time) our mathematical model reduces to the massless Dirac equation.

Chapter 3 contains a description of the progress recently made in the spectral theory of first order systems, with a particular focus on dimension three presented in Chapter 4.

We prove in Chapter 5 that the second asymptotic coefficient of the counting function of a first order system has the geometric meaning of the massless Dirac action.

Finally, in Chapter 6 we examine the spectral asymmetry of the massless Dirac operator. We work on a 3-torus equipped, initially, with a Euclidean metric and consider the behaviour of the spectrum under a perturbation of the metric. We derive an explicit asymptotic formula for the eigenvalue closest to zero.

Acknowledgements

Thanks must go, first and foremost, to my supervisor Dima Vassiliev for the direction, support, and innumerable cups of coffee he has provided over these past few years. Suffice it to say this thesis would not be here without his superb guidance.

Christian Böhmer, Olga Chervova, and Michael Levitin, friends and co-authors alike, have had a decisive influence on me and my work. I am extremely grateful for all their help.

There is no doubt in my mind that I would not have got this far without the boundless encouragement (and mockery) provided by my family. I hope there are sufficient sums in the sequel to show this was time well spent.

My doctoral studies have been generously funded by UCL and the Department of Mathematics, the latter by a J. J. Sylvester Scholarship and a John Hawkes Scholarship and the former by an Impact Studentship. I am exceptionally appreciative of this support.

Finally, I would like to extend my heartfelt thanks to Caitlin Dempsey for her friendship and for the confidence she has had in me throughout this endeavour.

Contents

Declaration	1
Abstract	2
Acknowledgements	3

1	Intr	oduction	7
	1.1	The massless Dirac operator: an overview	7
	1.2	Cosserat elasticity and the massless Dirac equation $\ldots \ldots \ldots$	12
	1.3	Microlocal analysis of first order systems I	16
	1.4	Microlocal analysis of first order systems II	19
	1.5	The massless Dirac action as the second coefficient in the asymp-	
		totics of the counting function	22
	1.6	Spectral asymmetry and the massless Dirac operator	25
2	Rot	ational elasticity	29
	2.1	Introduction	29
	2.2	Setting the playing field	31
	2.3	Kinetic energy	35
	2.4	Potential energy	35
		2.4.1 Measuring rotational deformations	35
		2.4.2 Irreducible decomposition of rotational deformations \ldots	39
		2.4.3 Formula for potential energy	41
		2.4.4 Simplifying the formula for potential energy	41
	2.5	Lagrangian of Rotational elasticity	42
	2.6	Reformulating the problem in the language of spinors \ldots \ldots \ldots	43
	2.7	Euler–Lagrange equation	46
	2.8	Plane wave solutions	47
	2.9	The massless Dirac equation	54

	2.10	The linearised problem	56
	2.A	Notation	58
	2.B	Spinor representation of torsion	59
	$2.\mathrm{C}$	Separation of variables	61
	2.D	Conservation of mass	64
	2.E	Matrix representation of the Lagrangian of Rotational elasticity $% \mathcal{A}(\mathcal{A})$.	65
3	Mic	rolocal analysis of first order systems I	67
	3.1	Key ideas in microlocal analysis	68
		3.1.1 Hamiltonian trajectories	68
		3.1.2 Oscillatory integrals	69
		3.1.3 Pseudodifferential operators	71
		3.1.4 Time-dependent oscillatory integrals	73
	3.2	Main results	77
	3.3	Overview of the algorithm for the construction of the propagator .	85
	3.4	U(1) connection	91
	3.5	Spectral asymmetry	95
	3.6	Bibliographic review	96
4	Mic	rolocal analysis of first order systems II	98
	4.1	$Main results \ldots \ldots$	98
	4.2	Reduction from the general setting $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	105
	4.3	Teleparallel connection	107
	4.4	Relation between curvature of the $U(1)$ connection and torsion of	
		the Teleparallel connection	112
	4.5	Integration of the curvature term	117
	4.6	The subprincipal symbol of the massless Dirac operator	118
	4.7	Proof of Theorem 4.1.2	120
	4.8	Explicit formula for axial torsion	121
	4.A	The massless Dirac operator	122
	4.B	The spectrum for the torus and the sphere	129
5	\mathbf{Spe}	ctral theoretic characterization of the massless Dirac action	133
	5.1	Main result	133
	5.2	Spinor representation of the principal symbol	140
	5.3	Proof of Theorem 5.1.1	144
	5.4	Spin structure	146
	5.5	Conformal invariance	148
	5.6	SU(2) invariance	150
	5.7	Invariance under rigid rotations	151
	5 A	The massless Dirac action	153

	5.B	Example	153
6	Spee	ctral asymmetry of the massless Dirac operator	157
	6.1	Introduction	157
	6.2	Main result	159
	6.3	Perturbation process I: preliminaries	162
	6.4	Perturbation process II: formal procedure	167
	6.5	Perturbation process III: justification	170
	6.6	Proof of Theorem 6.2.1	172
	6.7	Axisymmetric case	178
	6.8	Example of quadratic dependence on ϵ	180
	6.9	Example of quartic dependence on ϵ	182
	6.10	The eta invariant \ldots	185

Bibliography

190

Chapter 1

Introduction

1.1 The massless Dirac operator: an overview

This thesis is concerned with the massless Dirac operator in dimension three. This is directly related to the accepted mathematical model for a massless neutrino or antineutrino.

The massless Dirac operator is a particular self-adjoint complex two-by-two matrix partial differential operator acting on two-columns of complex-valued scalar fields. In theoretical physics and differential geometry this pair of complex-valued scalars is referred to as a *spinor*. When we say that the components of a spinor are scalars we mean that they are anholonomic objects, i.e. they do not depend on the choice of local coordinates.

Although reasonably familiar when defined on \mathbb{R}^3 , one can work with a more general operator defined on a Riemannian 3-manifold (that is, a 3-manifold equipped with a Riemannian metric). For simplicity, in this thesis we will always work on either \mathbb{R}^3 or an oriented compact Riemannian 3-manifold, M, unless otherwise stated.

We give two explicit formulae for the massless Dirac operator, W, both of which are utilised in the thesis. In matrix form we have

$$W := -i\sigma^{\alpha} \left[\frac{\partial}{\partial x^{\alpha}} + \frac{1}{4}\sigma_{\beta} \left(\frac{\partial\sigma^{\beta}}{\partial x^{\alpha}} + \left\{ \begin{matrix} \beta \\ \alpha\gamma \end{matrix} \right\} \sigma^{\gamma} \right) \right].$$
(1.1.1)

Here summation is carried out over repeated tensor indices $\alpha = 1, 2, 3$ and $\sigma^{\beta} = \sigma^{\beta}(x), \beta = 1, 2, 3$, are a set of Pauli matrices. These matrices span the 3-dimensional real vector space of trace-free two-by-two Hermitian matrices and satisfy the defining relation

$$\sigma^{\alpha}\sigma^{\beta} + \sigma^{\beta}\sigma^{\alpha} = 2g^{\alpha\beta}I_{2\times 2},\tag{1.1.2}$$

where $g^{\alpha\beta}$ is the metric tensor and $I_{2\times 2}$ the two-by-two identity matrix. In general, the Pauli matrices depend upon local coordinates $x^{\beta} \in M$, $\beta = 1, 2, 3$. In addition, in (1.1.1) $\left\{ {}^{\alpha}_{\beta\gamma} \right\}$ are the Christoffel symbols, the Levi-Civita connection coefficients, given by

$$\begin{cases} \alpha \\ \beta \gamma \end{cases} = \frac{1}{2} g^{\alpha \delta} \left(\frac{\partial g_{\delta \beta}}{\partial x^{\gamma}} + \frac{\partial g_{\gamma \delta}}{\partial x^{\beta}} - \frac{\partial g_{\beta \gamma}}{\partial x^{\delta}} \right).$$
 (1.1.3)

An alternative formulation of the massless Dirac operator, utilising so-called spinor notation, is

$$W := -i\sigma^{\alpha}{}_{\dot{a}b}\nabla_{\alpha}. \tag{1.1.4}$$

Aside from the appearance of the covariant derivative ∇ with respect to the Levi-Civita connection, see Section I.A in [81], this formula differs from (1.1.1) in its use of spinor indices $\dot{a} = \dot{1}, \dot{2}, b = 1, 2$.

The appearance of a dotted index in formula (1.1.4) is a matter of tradition. We could have written all formulae containing spinor indices without using dots. The necessity to distinguish dotted and undotted indices arises when working in the more general 4-dimensional Lorentzian setting. In dimension four the spin group, which determines transformation properties of spinors under changes of the corresponding anholonomic basis, is reducible, having two distinct irreducible subgroups. This necessitates the introduction of two types of index (dotted/undotted) indicating the so-called left- or right-handedness of spinors, i.e. distinguishing two possible types of spinor determined by behaviour under changes of anholonomic basis.

In three dimensions this is not a critical issue: the spin group is irreducible and, as such, all spinors transform in the same manner under changes of the anholonomic basis. We need not, therefore, use dotted indices at all. However, to preserve accord between the formulae appearing in this thesis and those in the literature, we maintain the dotted indices.

The use or spinor indices, or lack thereof, in formulae (1.1.1) and (1.1.4) is made clear by the formulation of the massless Dirac equation, the accepted mathematical model for a massless neutrino or antineutrino. Forming the massless Dirac equation is equivalent to the introduction of time into our problem, that is, considering the dynamic equation

$$(D_t + W) w = 0. (1.1.5)$$

Here, $D_t := -i\partial/\partial t$ and w(t, x) is a column-function on $\mathbb{R} \times M$. Rewriting this equation in spinor form, for example, we arrive at the standard formulation of the massless Dirac equation

$$-i\left(-\sigma^{0}{}_{\dot{a}b}\partial_{t}+\sigma^{\alpha}{}_{\dot{a}b}\nabla_{\alpha}\right)\zeta^{b}=0.$$
(1.1.6)

Here ζ denotes a two-component complex-valued spinor field on $\mathbb{R} \times M$, summation is carried out over repeated spinor indices b = 1, 2, and $\sigma^0{}_{ab}$ is the negative identity matrix (this is again a relic of the 4-dimensional setting), see Appendix 2.A. In addition, the subscript attached to the partial derivative ∂ indicates the variable with respect to which partial differentiation occurs (e.g. $\partial_1 := \partial/\partial x^1$).

We see that the massless Dirac equation is a system of two $(\dot{a} = 1, 2)$ complex linear partial differential equations on the 4-manifold $\mathbb{R} \times M$ for the unknown spinor field w. Note that the mapping $\sigma^0{}_{ab} \mapsto -\sigma^0{}_{ab}$ is equivalent to switching our object of study from a neutrino to an antineutrino.

Of course, (1.1.5) and (1.1.6) are two equivalent ways of writing the same hyperbolic system of equations. In particular, the elements of the 2-column w are the components of the spinor ζ . We use different notation in (1.1.5) and (1.1.6) as a matter of tradition.

For clarity, we state the explicit form of the massless Dirac equation when the massless Dirac operator is defined on \mathbb{R}^3 . In this instance the formulae are particularly simple and will allow us to demonstrate various properties of the massless Dirac operator with ease throughout this thesis. On \mathbb{R}^3 the Pauli matrices σ^{α} are constant and so, taking the canonical choice, see (2.A.4) in Appendix 2.A, we are able to write the massless Dirac equation as

$$-i \begin{pmatrix} \partial_t + \partial_3 & \partial_1 - i\partial_2 \\ \partial_1 + i\partial_2 & \partial_t - \partial_3 \end{pmatrix} \begin{pmatrix} \zeta^1 \\ \zeta^2 \end{pmatrix} = 0.$$
(1.1.7)

In this thesis we eventually consider a spectral problem arising from the separation of variables in equation (1.1.5). In particular, we seek a separable solution $w(t, x) = e^{-it\lambda}v(x)$ and consider the corresponding spectral problem

$$Wv = \lambda v. \tag{1.1.8}$$

In fact, we study this problem for a more general operator than W in arbitrary dimension (greater than, or equal to, 2), focusing on the massless Dirac operator at a later stage, see Chapter 3 and Chapter 4.

We study the *spectral* and *counting functions*, defined respectively as

$$e(\lambda, x, x) := \sum_{0 < \lambda_k < \lambda} \|v_k(x)\|^2,$$
 (1.1.9)

$$N(\lambda) := \sum_{0 < \lambda_k < \lambda} 1 = \int_M e(\lambda, x, x) \sqrt{\det g_{\alpha\beta}} \, dx, \qquad (1.1.10)$$

where λ_k is an eigenvalue of the operator W determined from (1.1.8) and $v_k(x)$ a corresponding normalised eigenfunction evaluated at the point $x \in M$, $\lambda \in \mathbb{R}$ is a positive parameter, $\|\cdot\|^2$ is the square of the Euclidean norm, and $dx = dx^1 dx^2 dx^3$. In other words, the counting function $N(\lambda)$ is the number of eigenvalues λ_k between zero and λ , and $e(\lambda, x, x)$ its local (pointwise) version. Again, we begin by studying (1.1.9) and (1.1.10) for more general operators before eventually considering the special case of the massless Dirac operator.

The purpose of this thesis is to consider the massless Dirac operator from a new geometric point of view. The formulae (1.1.1)-(1.1.10) are our basic objects of study in this endeavour.

We start in Chapter 2 by developing a novel theory of elasticity centred around rotations of material points. This is based somewhat on the work of the Cosserat brothers and is linked to the study of Teleparallel gravity, an alternative field theory developed by A. Einstein and É. Cartan. Although seemingly disconnected from the study of the massless Dirac operator, we are able to give a theorem connecting our theory with plane wave solutions of the massless Dirac equation, an unusual geometric link.

We go on in Chapter 3 to present recent developments in the theory of general first order matrix (pseudo)differential operators. Reducing this to the case of the massless Dirac operator in three dimensions we then compute two-term spectral asymptotics for the spectral and counting functions: this work is found in Chapter 4.

For a specific class of operators (those featured in Chapter 4 under the additional assumption of vanishing subprincipal symbol), the second asymptotic term is shown to be related to the massless Dirac action. A comprehensive discussion concerning this relationship is given in Chapter 5.

Finally, in Chapter 6 we consider the issue of spectral asymmetry of the massless Dirac operator. We show that, by perturbing the metric on a 3-torus, we are able to achieve spectral asymmetry. We develop a perturbation theory taking into account some of the unique properties of the massless Dirac operator in dimension three and present several families of Riemannian metrics for which we can compute explicitly the shift of the zero eigenvalue.

Note that, as this thesis draws on a number of different areas of mathematics, it has not been possible to achieve complete consistency in the use of notation. For example, in Chapter 2 we denote by ξ a nonvanishing spinor field, whereas ξ in Chapters 3–6 stands for the dual variable (momentum). Within each particular chapter the notation is consistent.

In addition, a number of lengthy calculations that would add significantly to the overall length of this thesis have been omitted, although, of course, details are always provided to assist the interested reader in the derivation. The author welcomes requests for further details regarding such calculations, and the omission of a detailed calculation is always noted in the text of this thesis.

For greater clarity, a more detailed introduction to each chapter of this thesis is given below.

1.2 Cosserat elasticity and the massless Dirac equation

In Chapter 2 we develop a new theory of elastic media which we name *Rotational elasticity*. Working in 3-dimensional Euclidean space, which we view as an elastic continuum, we develop a theory of elasticity in which deformations of our medium correspond only to rotations of material points.

While this may seem exotic, it is a natural limit case of a more general theory of elastic media developed by the Cosserat brothers in the early twentieth century. A general deformation in this case corresponds to both a displacement and rotation of material points. It is interesting to note that classical linear elasticity, wherein deformations are induced only by displacements of material points, is a wellstudied limit case of Cosserat elasticity.

Rotational elasticity is an interesting topic of study for several reasons. Unlike its more famous cousin, linear elasticity, Rotational elasticity has received very little attention from the academic community. It is therefore a beneficial exercise to contrast these two theories. In addition, as we show, there are unexpected links both with gravitation and the massless Dirac operator.

The essential geometric object in this construction is the coframe, the field of orthonormal covector fields e^j , j = 1, 2, 3, defined over \mathbb{R}^3 . The orthogonality condition is expressed as a kinematic constraint on the coframe:

$$g = \delta = \delta_{jk} e^j \otimes e^k \tag{1.2.1}$$

or, equivalently, component-wise

$$g_{\alpha\beta} = \delta_{\alpha\beta} = \delta_{jk} e^j{}_{\alpha} e^k{}_{\beta}, \quad \alpha, \beta = 1, 2, 3.$$
(1.2.2)

Here, for simplicity, we assume the metric to be Euclidean (i.e. $g_{\alpha\beta} = \delta_{\alpha\beta}$ where δ is the Kronecker delta). Note that we select coframes satisfying (1.2.1) which leaves us with three real degrees of freedom at each point in \mathbb{R}^3 . This justifies our description of (1.2.1) as a kinematic constraint.

We take as the dynamical variables of our theory the coframe e^{j} and a real positive density ρ . Following the same logic as in classical elasticity, we define an appropriate measure of deformation of our medium which, in our case, is the torsion tensor of the connection generated by our coframe. More specifically, the connection generated by the coframe is the *Teleparallel connection*, a metric compatible affine connection with vanishing curvature but non-vanishing torsion. The connection coefficients can be expressed explicitly in terms of the coframe and its partial derivatives, and are given by

$$\Gamma^{\alpha}{}_{\beta\gamma} = e_k{}^{\alpha} \frac{\partial e^k{}_{\gamma}}{\partial x^{\beta}}, \quad \alpha, \beta, \gamma = 1, 2, 3, \tag{1.2.3}$$

where

$$e_k{}^{\alpha} := \delta_{kj} g^{\alpha\beta} e^j{}_{\beta} \tag{1.2.4}$$

is the "inverse" of the coframe: see Section 4.3 for further discussion. The components of the torsion tensor are defined as the difference

$$T^{\alpha}{}_{\beta\gamma} := \Gamma^{\alpha}{}_{\beta\gamma} - \Gamma^{\alpha}{}_{\gamma\beta}. \tag{1.2.5}$$

This is an important observation that links Chapter 2 with Chapter 3 and Chapter 4.

We are then able to define both a potential and kinetic energy in terms of the torsion tensor, coframe, and density which allows us to construct a Lagrangian (energy functional) from which we can determine appropriate equations of motion for waves propagating in our elastic continuum.

However, achieving this aim while maintaining the kinematic constraint (1.2.1) is extremely difficult. To ameliorate this problem, we reformulate our model in terms of spinors. It is known that, in three dimensions, a coframe and positive density e^{j} , ρ are equivalent to a spinor field ζ^{b} (modulo the sign of the field). Taking advantage of this fact, we are able to rewrite our Lagrangian in a more user-friendly manner, avoiding the constraint (1.2.1), and immediately seek plane wave solutions.

The main results presented in Chapter 2 are the following theorems.

Firstly, we present a theorem concerning plane wave solutions, i.e. spinors satisfying the field equations of Rotational elasticity of the form

$$\zeta(t, x^1, x^2, x^3) = \begin{pmatrix} 1\\ 0 \end{pmatrix} e^{i(p_0 t + p_1 x^1 + p_2 x^2 + p_3 x^3)}, \qquad (1.2.6)$$

where $p_i \in \mathbb{R}$, i = 0, 1, 2, 3, and $p_0 \neq 0$. Note that we can always map one of the components of our plane wave spinor field to zero by performing a rotation of the coordinate system. This justifies our choice of spinor field in (1.2.6).

We can then state the following theorem in which v_1 and v_2 are real constants expressed via the elastic moduli of the underlying elastic continuum, see formula (2.8.22).

Theorem 1.2.1. Plane wave solutions of Rotational elasticity can, up to rescaling and rotation, be explicitly written down in the form (1.2.6) with arbitrary nonzero p_0 and $p = (p_1, p_2, p_3)$ determined as follows.

• If $v_1 > 0$ and $v_2 > 0$ and $v_1 \neq v_2$ then we have two possibilities:

 $- p = \left(0, 0, \pm \frac{p_0}{v_1}\right) \text{ (type 1 wave), or}$ $- p = \left(\frac{|p_0|}{v_2}\cos\varphi, \frac{|p_0|}{v_2}\sin\varphi, 0\right) \text{ where } \varphi \in \mathbb{R} \text{ is arbitrary (type 2 wave).}$

• If $v_1 > 0$ and $v_2 > 0$ and $v_1 = v_2$ then p is an arbitrary 3-vector satisfying $||p|| = \frac{|p_0|}{v_1}$.

• If
$$v_1 > 0$$
 and $v_2 = 0$ then $p = \left(0, 0, \pm \frac{p_0}{v_1}\right)$.

• If $v_1 = 0$ and $v_2 > 0$ then $p = \left(\frac{|p_0|}{v_2}\cos\varphi, \frac{|p_0|}{v_2}\sin\varphi, 0\right)$ where $\varphi \in \mathbb{R}$ is arbitrary.

This theorem classifies all plane wave solutions of Rotational elasticity. This allows us to compare our model directly with linear elasticity, in particular the existence of longitudinal and transverse waves and their modes of propagation.

The existence of plane wave solutions is highly nontrivial due to the nonlinear nature of the system of partial differential equations governing our model of elasticity. This nonlinearity persists even when we simplify matters by writing our equations in terms of spinors. As such, this theorem is a surprising result. While we are not sure exactly why this holds, we suspect it is related to underlying group-theoretic properties of our model.

Our next result forges a link between Rotational elasticity and the massless Dirac equation for the case when we choose to deal with an axial material, that is, a material for which the two constants are specified as follows:

$$v_1 = 1, \quad v_2 = 0.$$
 (1.2.7)

Under this condition we have the following theorem.

Theorem 1.2.2. In the case of a purely axial material a plane wave spinor field is a solution of Rotational elasticity if and only if it is a solution of one of the two (\pm) massless Dirac equations (1.1.6)

$$-i \begin{pmatrix} \pm \partial_t + \partial_3 & \partial_1 - i\partial_2 \\ \partial_1 + i\partial_2 & \pm \partial_t - \partial_3 \end{pmatrix} \begin{pmatrix} \zeta^1 \\ \zeta^2 \end{pmatrix} = 0.$$
(1.2.8)

This somewhat surprising result motivated us to investigate the structure of the massless Dirac operator in more detail with a particular focus on the role of geometry. To achieve this aim we first had to develop an understanding of more general first order systems of partial differential equations. An overview of this work is provided in the following section.

1.3 Microlocal analysis of first order systems I

In order to develop a deeper understanding of the massless Dirac operator we first consider the more general case of an arbitrary first order system of partial differential equations under certain assumptions. The material appearing in Chapter 3 contains an abridged version of [22]. Certain aspects of this material are also investigated in [20].

Initially, we consider a classical first order self-adjoint pseudodifferential operator A acting on m-columns of complex-valued half-densities over a connected compact n-dimensional manifold without boundary, M, where both $m, n \geq 2$.

Note that the dynamical variable in this case is not a spinor field as in Section 1.1 and Section 1.2. We work instead with half-densities. A half-density $v \in C^{\infty}(M)$ changes in a specific manner under changes of local coordinates. We say that, given two choices of local coordinates $x, \tilde{x} \in M, v(x)$ is a half-density if $v(x) = J^{1/2}(x)\tilde{v}(\tilde{x}(x))$ where \tilde{v} is the representation of v in the coordinates \tilde{x} and $J(x) = |\det \partial \tilde{x}/\partial x|$ is the Jacobian determinant.

Half-densities are useful to work with as one is able to invariantly define an inner product without any additional assumptions or constructions. In particular, define the inner product for half-densities $v, w \in C^{\infty}(M)$

$$\langle v, w \rangle := \int_M w^* v \, dx. \tag{1.3.1}$$

Here the * indicates complex conjugation. Note that (1.3.1) as defined here extends to the case when v, w are columns of half-densities: in this case the * must be understood as indicating both complex conjugation *and* transposition, i.e. as Hermitian conjugation.

As we work with matrix operators throughout this thesis this more general notion of an inner product will be useful. Under changes of local coordinates the inner product (1.3.1) is invariant. This gives an invariantly defined notion of formal self-adjointness for our operator A:

$$\langle Av, w \rangle = \int_{M} w^*(Av) \, dx = \int_{M} (Aw)^* v \, dx = \langle v, Aw \rangle. \tag{1.3.2}$$

Again, (1.3.2) extends to the case when v, w are columns of half-densities and A is an appropriate matrix operator.

Our aim is to construct the propagator, i.e. the one-parameter family of operators defined as

$$U(t) := e^{-itA} = \sum_{k} e^{-it\lambda_{k}} v_{k}(x) \int_{M} [v_{k}(y)]^{*}(\cdot) dy, \qquad (1.3.3)$$

 $t \in \mathbb{R}$. The propagator provides a solution of the Cauchy problem

$$(D_t + A)w = 0, \quad w|_{t=0} = v, \tag{1.3.4}$$

cf. (1.1.5). Of course, the solution of the Cauchy problem (1.3.4) can be written in terms of the propagator (1.3.3) as w = U(t)v. We construct the propagator (1.3.3) explicitly in terms of oscillatory integrals, modulo an integral operator with an infinitely smooth kernel, see Chapter 3 for technical details.

This allows us to derive two-term asymptotics for the spectral function (1.1.9) of the operator A in the form

$$e(\lambda, x, x) = a(x)\lambda^n + b(x)\lambda^{n-1} + o(\lambda^{n-1})$$
(1.3.5)

as $\lambda \to +\infty$. Formal integration of (1.3.5) over the manifold M gives an asymptotic formula for the counting function (1.1.10) of the operator A. Section 8 of [22] gives the details of the assumptions under which this argument holds, in particular Theorems 8.3 and 8.4 are required to justify this statement.

Our major result is being able to state explicit formulae for the coefficients a(x)and b(x) appearing in (1.3.5). This is significant as previous publications on the subject have either stopped short of providing such formulae or have given incorrect expressions, see Section 3.6 in Chapter 3. We show that

$$a(x) = \sum_{j=1}^{m^+} \int_{h^{(j)} < 1} d\xi, \qquad (1.3.6)$$

$$b(x) = -n \sum_{j=1}^{m^+} \int_{h^{(j)} < 1} \left[[v^{(j)}]^* A_{\text{sub}} v^{(j)} - \frac{i}{2} \{ [v^{(j)}]^*, A_1 - h^{(j)}, v^{(j)} \} + \frac{i}{n-1} h^{(j)} \{ [v^{(j)}]^*, v^{(j)} \} \right] d\xi, \quad (1.3.7)$$

where m^+ is the number of positive eigenvalues $h^{(j)} = h^{(j)}(x,\xi)$ of the principal symbol A_1 of the operator A, $v^{(j)}(x)$ are the corresponding eigenvectors of the principal symbol, A_{sub} is the subprincipal symbol of A, $\{\cdot, \cdot\}$ and $\{\cdot, \cdot, \cdot\}$ are the Poisson bracket on matrix-functions and its further generalisation, and $d\xi =$ $(2\pi)^{-n}d\xi = (2\pi)^{-n}d\xi_1 \dots d\xi_n$, see Chapter 3 for further details and complete definitions.

1.4 Microlocal analysis of first order systems II

In Chapter 4 we reduce the general setting described in Section 1.3 to the particular case wherein:

Assumption 1: The dimension of our manifold is three, n = 3.

Assumption 2: We are dealing with a 2×2 matrix operator, m = 2.

Assumption 3: Our operator is differential as opposed to pseudodifferential.

Assumption 4: The principal symbol of our operator is trace-free.

This reduction is motivated by our interest in the massless Dirac operator. However, the massless Dirac operator defined in Section 1.1 is not an operator of the type discussed above: it acts on spinors, not half-densities. Fortunately, on a parallelizable manifold one can identify the components of a spinor with halfdensities (and, as we show in Chapter 4, the existence of the principal symbol of the operator A implies the 3-manifold M is parallelizable). When we perform this identification for the massless Dirac operator (1.1.1) we refer to the resulting operator as the massless Dirac operator on half-densities, see (4.A.3) in Appendix 4.A. Using this fact we will, in due course, be able to apply the tools developed in Chapter 3 to the massless Dirac operator on half-densities. The results can then be mapped to the case of the massless Dirac operator (1.1.1)

Under Assumptions 1–4, the principal symbol of our operator A admits a complete geometric description. In particular, a number of geometric objects are encoded in the principal symbol: • *The metric.* We observe that the determinant of the principal symbol is a negative definite quadratic form

$$\det A_1(x,\xi) = -g^{\alpha\beta}(x)\xi_\alpha\xi_\beta \tag{1.4.1}$$

and the coefficients $g^{\alpha\beta}(x)$, $\alpha, \beta = 1, 2, 3$, in (1.4.1) can be interpreted as the components of a symmetric contravariant Riemannian metric. In particular, this implies that the two eigenvalues of the (trace-free) principal symbol are of the form

$$\pm \sqrt{g^{\alpha\beta}(x)\xi_{\alpha}\xi_{\beta}}.$$
 (1.4.2)

• The Teleparallel connection. This is the affine connection defined in an operator-theoretic manner as follows. Given a point $x \in M$ and a covector $\xi \in T'_x M$ ($T'_x M = T^*_x M \setminus \{\xi = 0\}$, i.e. the cotangent fibre with zero removed) we can construct a parallel (co)vector $\tilde{\xi} \in T'_x M$ at the point $\tilde{x} \in M$ by solving the system of linear equations

$$A_1(x,\xi) = A_1(\tilde{x},\tilde{\xi}).$$
(1.4.3)

Equation (1.4.3) is equivalent to a real algebraic system of three linear equations for three unknowns, the components of $\tilde{\xi}$. This system has a unique solution and, furthermore, preserves the Riemannian norm of (co)vectors, i.e. the connection is metric compatible. As the parallel transport defined by (1.4.3) is independent of the curve along which we transport the (co)vector, the connection has zero curvature. In other words, (1.4.3) defines a Teleparallel connection.

• The topological charge. As is mentioned above, the existence of a principal symbol implies our 3-manifold M is parallelizable and, hence, orientable. Specifying a given orientation, we only allow changes of local coordinates $x \in M$ which preserve orientation. We then define the topological charge as

$$\mathbf{c} := -\frac{i}{2}\sqrt{\det g_{\alpha\beta}} \operatorname{tr}\left[(A_1)_{\xi_1}(A_1)_{\xi_2}(A_1)_{\xi_3}\right].$$
(1.4.4)

As we demonstrate in Chapter 4, the quantity \mathbf{c} takes one of only two values, ± 1 , and expresses the orientation of the principal symbol relative to the specified orientation of local coordinates.

Note that the Teleparallel connection appears in Chapter 2 as (1.2.3), although (1.4.3) is expressed in an operator-theoretic manner. In addition, the topological charge (1.4.4) is implicitly present in Chapter 2 and takes the value +1 by definition.

We are able to express the asymptotic coefficients of the spectral function (1.3.6), (1.3.7) for operators obeying Assumptions 1–4 in terms of the three geometric objects given above. Indeed, the following theorem is one of the main results of Chapter 4.

Theorem 1.4.1. The coefficients of the two-term asymptotics of the spectral function (1.3.6), (1.3.7) for operators under Assumptions 1–4 are given by the formulae

$$a(x) = \frac{1}{6\pi^2} \sqrt{\det g_{\alpha\beta}(x)},$$
 (1.4.5)

$$b(x) = \frac{1}{8\pi^2} \left(\left[3 \,\mathbf{c} * T^{\mathrm{ax}} - 2 \,\mathrm{tr} \,A_{\mathrm{sub}} \right] \sqrt{\det g_{\alpha\beta}} \,\right)(x) \,, \tag{1.4.6}$$

where

$$T_{\alpha\beta\gamma}^{\rm ax} := \frac{1}{3} (T_{\alpha\beta\gamma} + T_{\gamma\alpha\beta} + T_{\beta\gamma\alpha}) \tag{1.4.7}$$

is axial torsion (totally antisymmetric piece of the torsion tensor) and * is the Hodge star, see formula (4.3.4) in Chapter 4.

Turning our attention to the massless Dirac operator on half-densities, using Theorem 1.4.1 we can state a second theorem addressing the following question: is a given operator A, under Assumptions 1–4, a massless Dirac operator on half-densities? The answer is given by the following:

Theorem 1.4.2. The operator A is a massless Dirac operator on half-densities if and only if the following conditions are satisfied at every point of the manifold M:

- The subprincipal symbol of the operator A is proportional to the identity matrix.
- The second asymptotic coefficient of the spectral function, b(x), is zero.

This theorem completely characterizes the massless Dirac operator in terms of spectral theoretic quantities.

1.5 The massless Dirac action as the second coefficient in the asymptotics of the counting function

In Chapter 5 we study the massless Dirac action rather than the massless Dirac operator, i.e. the variational functional corresponding to the operator (1.1.1). This is motivated by the following observations.

- Observation 1. The formula for the second asymptotic coefficient of the spectral function for an operator A under Assumptions 1–4, (1.4.6), contains a term proportional to the subprincipal symbol, A_{sub} . If we consider operators under Assumptions 1–4 with the additional assumption that the subprincipal principal is identically zero, then:
 - The operator A is completely determined by its principal symbol.
 - The second asymptotic coefficient of the spectral function of the operator A, (1.4.6), is proportional to the Hodge star of the axial torsion of the Teleparallel connection (1.4.7).
- Observation 2. It is known that the massless Dirac action is directly related to axial torsion, see for example [23].

These two facts motivate our interest in the massless Dirac action. Our aim in Chapter 5 is to investigate the geometric meaning of the second asymptotic coefficient of the spectral and counting functions and, as we will show, link this to the massless Dirac action. Therefore, in this chapter, we work with a first order differential 2×2 matrix operator acting on two-columns of complex-valued halfdensities over a compact oriented 3-manifold under the following assumptions.

Assumption 5: The principal symbol of our operator is trace-free.

Assumption 6: The subprincipal symbol of our operator is identically zero.

As in Chapter 4, there are a number of objects encoded in our operator A.

- The metric. As we again work with a 2×2 matrix operator we are free to use the metric defined by (1.4.1).
- The nonvanishing spinor field. The determinant of the principal symbol (metric) does not uniquely determine the principal symbol itself. Supposing that the metric (1.4.1) is fixed, we are able to define a new geometric object by considering operators corresponding to the same metric. To make this more precise, fixing a reference principal symbol B_1 corresponding to the prescribed metric and considering all principal symbols A_1 corresponding to the same metric that are sufficiently close (a more detailed discussion is presented in Chapter 5) we are able to relate the two through

$$A_1(x,\xi) = R(x)B_1(x,\xi)R^*(x)$$
(1.5.1)

where R is an infinitely smooth SU(2) matrix-function close to the identity. In other words, particular SU(2) matrix-functions parametrise principal symbols corresponding to a given metric.

It is known that SU(2) matrices can be conveniently described by a (Weyl) spinor, i.e. a pair of complex numbers ζ^a , a = 1, 2. The relationship between the spinor field and SU(2) matrices is given by the formula

$$R = \frac{1}{\|\zeta\|} \begin{pmatrix} \overline{\zeta^1} & \overline{\zeta^2} \\ -\zeta^2 & \zeta^1 \end{pmatrix}, \qquad (1.5.2)$$

where complex conjugation is denoted by the overline and $\|\cdot\|$ is the appropriate Euclidean norm. Equation (1.5.2) establishes a one-to-one correspondence between SU(2) matrices and non-zero spinors, modulo a rescaling of the spinor by an arbitrary positive real factor.

Note that in this introduction we assume, for simplicity, that the spinor field is normalised, i.e. $\|\zeta\|^2 = |\zeta^1|^2 + |\zeta^2|^2 = 1$. In Chapter 5 we actually deal with a more general weighted spectral problem which leads to a natural scaling of the spinor field, but we want to minimise the technical details at this point.

• The topological charge. As indicated in Chapter 4, the existence of a principal symbol implies our manifold is parallelizable and, hence, orientable. In accordance with the argument given in Chapter 4, we use the topological charge as defined by (1.4.4).

Having identified these three geometric objects encoded within our operator A, it is reasonable to assume that the corresponding coefficients of the spectral/counting functions would be expressed through these same objects. The key result of Chapter 5 is the following theorem which confirms this assertion.

Theorem 1.5.1. The coefficients in the two-term asymptotics of the counting function (1.1.10) for the operator A as described in this section and under Assumptions 5 and 6 above are given by the formulae

$$a = \frac{1}{6\pi^2} \int_M \|\zeta\|^3 \sqrt{\det g_{\alpha\beta}} \, dx \,, \tag{1.5.3}$$

$$b = \frac{S(\zeta)}{2\pi},\tag{1.5.4}$$

where $S(\zeta)$ is the massless Dirac action (5.A.1) with Pauli matrices

$$\sigma^{\alpha} := (B_1)_{\xi_{\alpha}} \tag{1.5.5}$$

Note that the topological charge \mathbf{c} does not appear in (1.5.1) explicitly. However, it is implicitly present in our choice of Pauli matrices.

In addition formula (1.5.3) contains the term $\|\zeta\|^3$. As indicated, in this introduction we assume the spinor field to be normalised. We wrote formula (1.5.3) in a way that would make it applicable in the more general case dealt with in Chapter 5. The theorem applies equally well to a normalised spinor field nonetheless.

Theorem 1.5.1 provides a new perspective on two-term spectral asymptotics for first order systems. The operator we have described in this section (first order differential 2×2 matrix operator in dimension three with trace-free principal symbol and vanishing subprincipal symbol) is, compared to a general system, a very special mathematical object. However, even for this relatively simple case, the two-term spectral asymptotics have a highly nontrivial geometric meaning.

1.6 Spectral asymmetry and the massless Dirac operator

In the final chapter of this thesis we consider the issue of spectral asymmetry of the massless Dirac operator on a 3-torus. It is known that, taking the standard Euclidean metric on the 3-torus, the spectrum of the massless Dirac operator can be calculated explicitly. Indeed, the spectrum is symmetric about zero and zero is itself an eigenvalue. We will show that, by perturbing the metric, one can shift the zero eigenvalue and thus achieve spectral asymmetry.

The main reason for studying this problem is that producing explicit examples of spectral asymmetry for the massless Dirac operator is difficult. While it is known that there is no particular reason that the spectrum of the massless Dirac operator should be symmetric on a general oriented Riemannian 3-manifold, to our knowledge the only explicit example is given in [65]. The example is based on selecting a 3-manifold with Euclidean metric but highly nontrivial topology. In Chapter 6 we take a different approach. We maintain a simple topology and perturb the metric instead. We present a theorem concerning a general perturbation of this kind and then explicitly construct several families of Riemannian metrics which produce various shifts of the zero eigenvalue.

From this point we work on the unit torus \mathbb{T}^3 parametrised by periodic coordinates x^{α} , $\alpha = 1, 2, 3$, of period 2π . Taking the (unperturbed) metric to be Euclidean $(g_{\alpha\beta} = \delta_{\alpha\beta})$ the massless Dirac operator (corresponding to the standard spin structure, see Section 5.4) reads

$$W = -i \begin{pmatrix} \partial_3 & \partial_1 - i\partial_2 \\ \partial_1 + i\partial_2 & -\partial_3 \end{pmatrix}, \qquad (1.6.1)$$

cf. (1.1.7) and (1.1.8). The operator (1.6.1) admits separation of variables, i.e. one can seek eigenfunctions of the form $v(x) = ue^{im_{\alpha}x^{\alpha}}$ with $u \in \mathbb{C}^2 \setminus \{0\}$ and $m \in \mathbb{Z}^3$: eigenvalues and eigenfunctions can be calculated explicitly in this case.

The spectrum of (1.6.1) is as follows.

- Zero is an eigenvalue of multiplicity two.
- For each $m \in \mathbb{Z}^3 \setminus \{0\}$, ||m|| is an eigenvalue with corresponding eigenfunction of the form $ue^{im_{\alpha}x^{\alpha}}$, where u is unique up to multiplication by a constant factor.
- For each $m \in \mathbb{Z}^3 \setminus \{0\}$, -||m|| is an eigenvalue with corresponding eigenfunction of the form $ue^{im_{\alpha}x^{\alpha}}$, where u is unique up to multiplication by a constant factor.

The spectrum of (1.6.1) is clearly symmetric about zero.

We now perturb the metric, i.e. consider a new metric $g_{\alpha\beta}(x;\epsilon)$, the components of which are smooth functions of the cyclic coordinates x^{α} , $\alpha = 1, 2, 3$, and a small real parameter ϵ such that

$$g_{\alpha\beta}(x;0) = \delta_{\alpha\beta}.\tag{1.6.2}$$

We aim to demonstrate that, by choosing an appropriate perturbation, we can break the symmetry of the spectrum of the massless Dirac operator corresponding to the perturbed metric.

Demonstrating spectral asymmetry for the massless Dirac operator is complicated by the following issues.

- Issue 1. One way of establishing spectral asymmetry is to compare the asymptotic distribution of large positive eigenvalues and large negative eigenvalues, see Section 10 of [22] for details of the application of this technique to generic first order differential operators. However, the massless Dirac operator is special in the sense that the second asymptotic coefficient of its counting function is zero, see Theorem 1.4.2. Therefore, in the first two approximations, its large positive eigenvalues are distributed in the same way as its large negative eigenvalues. To circumvent this issue, we will deal instead with small eigenvalues to demonstrate spectral asymmetry.
- *Issue 2.* The massless Dirac operator commutes with the antilinear operator of *charge conjugation*, C, defined by

$$v = \begin{pmatrix} v^1 \\ v^2 \end{pmatrix} \rightarrow \begin{pmatrix} -\overline{v^2} \\ \overline{v^1} \end{pmatrix} =: \mathbf{C}(v).$$
 (1.6.3)

Therefore, if v is an eigenfunction of the massless Dirac operator, so is C(v). This means that all eigenvalues have even multiplicity. As a result of this property we are unable to apply standard techniques to calculate the shift of the zero eigenvalue under a perturbation of the metric as described above. To counter this difficulty, we develop a perturbation theory that takes into account the charge conjugation symmetry of the massless Dirac operator. The double eigenvalue at zero does not split under perturbation, and we show that it can, effectively, be treated as a simple eigenvalue.

The perturbation theory we develop culminates in the theorem presented below. Note that we raise and lower tensor indices using the Euclidean metric throughout the rest of this section. In addition, given a function $f: \mathbb{T}^3 \to \mathbb{C}$, we denote by

$$\hat{f}(m) := \frac{1}{(2\pi)^3} \int_{\mathbb{T}^3} e^{-im_{\alpha}x^{\alpha}} f(x) dx, \quad m \in \mathbb{Z}^3,$$
 (1.6.4)

its Fourier coefficients. We can now state the following theorem, the main result of Chapter 6.

Theorem 1.6.1. Let $W(\epsilon)$ be the massless Dirac operator corresponding to the perturbed metric $g_{\alpha\beta}(x;\epsilon)$ so that

$$W(0) = -i \begin{pmatrix} \partial_3 & \partial_1 - i\partial_2 \\ \partial_1 + i\partial_2 & -\partial_3 \end{pmatrix}.$$
 (1.6.5)

Let $\lambda_0(\epsilon)$ be the eigenvalue of $W(\epsilon)$ with smallest modulus and

$$h_{\alpha\beta}(x) := \left. \frac{\partial g_{\alpha\beta}}{\partial \epsilon} \right|_{\epsilon=0}.$$
 (1.6.6)

Then, we have

$$\lambda_0(\epsilon) = c\epsilon^2 + O(\epsilon^3) \quad as \quad \varepsilon \to 0, \tag{1.6.7}$$

where the constant c is given by the formula

$$c = \frac{i}{16} \varepsilon_{\alpha\beta\gamma} \sum_{m \in \mathbb{Z}^3 \setminus \{0\}} \left(\delta_{\mu\nu} - \frac{m_{\mu}m_{\nu}}{\|m\|^2} \right) m_{\alpha} \hat{h}_{\beta\mu}(m) \overline{\hat{h}_{\gamma\nu}(m)}, \qquad (1.6.8)$$

with the overline standing for complex conjugation and where ε is the totally antisymmetric quantity, $\varepsilon_{123} := +1$.

At the end of Chapter 6 we present two families of Riemannian metrics on \mathbb{T}^3 for which the eigenvalue with smallest modulus can be calculated explicitly. One family gives a quadratic shift of the zero eigenvalue, and the other produces a quartic shift. We show that, for both families, the explicit formulae for the eigenvalue with smallest modulus are in agreement with Theorem 1.6.1. Finally, we consider the relationship between Theorem 1.6.1 and the eta invariant, the traditional measure of spectral asymmetry.

Chapter 2

Rotational elasticity

2.1 Introduction

We work in 3-dimensional Euclidean space and view it as an elastic continuum whose material points can experience no displacements, only rotations, with rotations of different material points being kinematically independent. Rotations of material points of the 3-dimensional elastic continuum are described mathematically by attaching to each geometric point an orthonormal basis. This gives a field of orthonormal bases called the *coframe*.

The purpose of Chapter 2 is to develop a theory of elasticity on rotations, i.e. a theory of elasticity in which the coframe plays the role of the dynamical variable (unknown quantity). Recall that in classical elasticity the vector field of displacements is the dynamical variable.

Our motivation for studying such a seemingly exotic problem comes from four main sources.

The first source is *Cosserat elasticity*. In 1909 the Cosserat brothers proposed a theory of elasticity [24] which generalised classical elasticity by giving each material point rotational degrees of freedom. Cosserat elasticity has since become an accepted part of solid mechanics, though for most real life materials effects resulting from rotations of material points are small compared to effects resulting from displacements. From a purely mathematical point of view classical elasticity and rotational elasticity are two limit cases of Cosserat elasticity. One of these limit cases, classical elasticity, has been extensively studied so it seems natural to now examine the other limit case.

The second source is *Teleparallelism* (= absolute parallelism = fernparallelismus), a subject promoted by A. Einstein and É. Cartan [18, 73, 79] in the late 1920s. The idea of rotating material points lies at the heart of Teleparallelism and can easily be traced back to Cosserat elasticity: when in 1922 Cartan started developing what eventually became modern differential geometry he acknowledged [17] that he drew inspiration from the 'beautiful' work of the Cosserat brothers. The relationship between Cosserat elasticity and Teleparallelism is examined in detail in the review paper [39]. The basic difference between Teleparallelism and our model is that in Teleparallelism the metric is viewed as a dynamical variable rather than as a prescribed tensor field: formula (2.2.1) ceases being a kinematic constraint and becomes the definition of the metric. Consequently, in Teleparallelism the coframe is varied without any constraints whatsoever which, on the one hand, makes calculations easier and, on the other, allows one to model gravitational phenomena.

The third source is the theory of liquid crystals and, in particular, the concept of an *Ericksen fluid*. According to [29], in a liquid crystal one can observe 'orientation waves which propagate, inducing little or no motion of the fluid' and Ericksen's mathematical model is the natural way of describing this phenomenon. The only difference between Ericksen's model and ours is that in Ericksen's model one attaches to each geometric point a single unit vector rather than an orthonormal basis as we do. More precisely, in Ericksen's model there is also no distinction between the head and tail of the unit vector, which leads to mathematical analysis based on projective geometry.

Finally, the fourth source is the concept of *gyrocontinuum* proposed by M. Brocato and G. Capriz [13]. Quoting from [13], Brocato and Capriz "consider a body endowed with the fine structure of very many, very small gyroscopes and describe it as a sort of composite (perhaps constrained) Cosserat continuum (gyrocontinuum)". Note that even though the basic idea of a rotating material point in [13] is similar to ours, there are substantial differences between our model and that of [13]. In particular, the equations in [13] are nonlinear both in terms of geometry and material behaviour (while the material behaves linearly in our model) and the displacements of material points in [13] are nonzero.

From a physical point of view we are studying a classical field theory which gives a novel description of fermions, i.e. elementary particles such as the neutrino and the electron. Our theory produces a wave description for particles, however, a formal quantisation scheme is yet to be developed. This means that we are attempting to model "world aether" rather than some realistic macroscopic material. A more modern way of rephrasing this is that we are modelling possible excitations of the vacuum and giving them a particle interpretation.

Chapter 2 has the following structure. In Section 2.2 we define our dynamical variables (unknowns of our theory), in Section 2.3 we write down the kinetic energy and in Section 2.4 we write down the potential energy. The Lagrangian of rotational elasticity is written down in Section 2.5. In Section 2.6 we reformulate our model in the language of spinors and in Section 2.7 we discuss the corresponding Euler-Lagrange equation. In Section 2.8 we construct an explicit class of solutions which we call plane wave solutions; this construction is summarised in Theorem 2.8.1 which is the main result of Chapter 2. Finally, in Section 2.9 we compare our model with the massless Dirac equation and in Section 2.10 we briefly outline a linearised version of rotational elasticity.

2.2 Setting the playing field

We work in Euclidean space \mathbb{R}^3 equipped with Cartesian coordinates x^{α} , $\alpha = 1, 2, 3$, and standard Euclidean metric. We denote time by t. Partial differentiation in t and x^{α} , $\alpha = 1, 2, 3$, is denoted by ∂_t and ∂_{α} respectively.

The coframe e is a triple of orthonormal covector fields e^j , j = 1, 2, 3, in \mathbb{R}^3 . Each covector field e^j can be written more explicitly as $e^j{}_{\alpha}$ where the tensor index $\alpha = 1, 2, 3$ enumerates the components. The orthonormality condition for the coframe can be represented as a single tensor identity

$$g = \delta_{jk} e^j \otimes e^k \tag{2.2.1}$$

where δ is the Kronecker delta and $g_{\alpha\beta} = \delta_{\alpha\beta}$ is the Euclidean metric. For the sake of clarity we repeat formula (2.2.1) giving tensor indices explicitly and performing summation over frame indices explicitly:

$$\delta_{\alpha\beta} = e^1{}_{\alpha}e^1{}_{\beta} + e^2{}_{\alpha}e^2{}_{\beta} + e^3{}_{\alpha}e^3{}_{\beta} \tag{2.2.2}$$

where α and β run through the values 1, 2, 3. We view the identity (2.2.1) as a kinematic constraint: the covector fields e^j are chosen so that they satisfy (2.2.1), which leaves us with three real degrees of freedom at every point of \mathbb{R}^3 .

We work only with coframes which have positive orientation, i.e. which satisfy the condition

$$\det e^{j}{}_{\alpha} = +1 > 0. \tag{2.2.3}$$

If one views $e^{j}{}_{\alpha}$ as a 3×3 real matrix function, then conditions (2.2.1) and (2.2.3) mean that this matrix function is special orthogonal. Thus, the coframe can be thought of as a field of special orthogonal matrices. This matrix approach is explained in greater detail in Appendix 2.E.

As dynamical variables in our model we choose the coframe e and a positive density ρ . Our coframe and density are functions of Cartesian coordinates x^{α} , $\alpha = 1, 2, 3$, as well as of time t. At a physical level, making the density ρ a dynamical variable means that we view our continuum more like a fluid rather than a solid. In other words, we allow the material to redistribute itself so that it finds its equilibrium density distribution. Observe that the total number of real dynamical degrees of freedom contained in the coframe e and positive density ρ is four, exactly as in a two-component complex-valued spinor field.

The underlying motivation for taking the density to be a dynamical variable is our aim of modelling quantum mechanical phenomena. This will become clear when we eventually switch to spinors: we will arrive at formula (2.6.3) which is the standard formula for the density of a massless neutrino field (zeroth component of the neutrino 4-current).

Note that there is nothing fundamentally wrong in taking a prescribed density (as opposed to a density which is a dynamical variable): the theory one gets is very similar to the one described in the current chapter and most formulae carry through with minimal changes. We will briefly examine a version of our theory with prescribed ρ in the very end of Chapter 2, in Section 2.10.

Below is the list of the main assumptions on which our model will be based.

- Assumption 1: our model is geometrically nonlinear. This means that we do not linearise rotations and we do not linearise the density. In other words, we allow our material points to experience full turns and we allow our density to experience changes comparable to the density itself.
- Assumption 2: our material is physically linear. This means that our potential energy is chosen to be quadratic in torsion (the latter serves as the measure of rotational deformations, see Section 2.4.1). Note that physical linearity does not contradict geometric nonlinearity: locally (in space and time) material points "do not know" that they may eventually experience full rotations and the density "does not know" that it may eventually experience a change comparable to its current value. This assumption is similar to the hyperelasticity assumption sometimes made in classical elasticity. It describes a highly idealised situation which is not likely to occur in realistic materials.

- Assumption 3: our material is homogeneous and isotropic. Homogeneity means that physical properties of the material are the same at all points of our continuum and isotropy means that there are no preferred directions.
- Assumption 4: our model is invariant under rigid rotations of the coframe. By a rigid rotation of the coframe we understand the transformation

$$e^j \mapsto O^j{}_k e^k \tag{2.2.4}$$

where O_k^j is a *constant* special orthogonal matrix. The thinking here is that when we attach an orthonormal basis to each geometric point of our continuum there is no reason to associate one particular direction with e^1 , another with e^2 and a third with e^3 . What matters is how these directions change when we move from one point to another, i.e. how orthonormal bases at different points differ relative to each other. A rigid rotation of the coframe means that we simultaneously rotate all our orthonormal bases by the same angle around the same axis. We view rigid rotations of the coframe as gauge transformations and assume that our model does not feel them. See also [51] for a detailed exposition of gauge theory for problems similar to the ones considered in Chapter 2.

Assumption 4 effectively means that the actual dynamical variable in our model is not the coframe itself, but the coframe modulo a rigid rotation. A loose way of rephrasing this would be to say that the actual dynamical variable is not the coframe but rather its rotation.

Assumption 4 is similar to the assumption of objectivity or frame-indifference in classical (without displacements) elasticity theory. However, it is not the same. When in classical elasticity one talks of objectivity or frame-indifference this normally refers to rotations of the coordinate system. Assumption 4, on the other hand, refers to "internal" degrees of freedom present in the rotating material point and the coframe here is unrelated to the choice of coordinate system.

2.3 Kinetic energy

Kinetic energy is given by the formula

$$K(t) = c^{\rm kin} \int \|\omega\|^2 \rho \, dx^1 dx^2 dx^3$$
 (2.3.1)

where $c^{\rm kin}$ is some positive constant and ω is the (pseudo)vector of angular velocity

$$\omega = \frac{1}{2} * (\delta_{jk} e^j \wedge \partial_t e^k). \tag{2.3.2}$$

Here \wedge is the exterior product and * is the Hodge star (2.A.1).

In writing the formula for kinetic energy (2.3.1) we think of each material point as a uniform ball possessing a moment of inertia and without a preferred axis of rotation.

If we were to consider a more complete model, with displacements, the formula for angular velocity (2.3.2) would change as displacements generate macrorotations.

We give for reference a more explicit version of the formula for angular velocity (2.3.2):

$$\omega_{\alpha} = \frac{1}{2} \sum_{j=1}^{3} \begin{pmatrix} e^{j}_{2} \partial_{t} e^{j}_{3} - e^{j}_{3} \partial_{t} e^{j}_{2} \\ e^{j}_{3} \partial_{t} e^{j}_{1} - e^{j}_{1} \partial_{t} e^{j}_{3} \\ e^{j}_{1} \partial_{t} e^{j}_{2} - e^{j}_{2} \partial_{t} e^{j}_{1} \end{pmatrix}.$$
 (2.3.3)

2.4 Potential energy

2.4.1 Measuring rotational deformations

In order to write down the formula for the potential energy we need to measure deformations caused by rotations of the coframe. More specifically, we need to measure deformations caused by the fact that at different points the coframe is oriented differently. Obvious candidates for a measure of deformations are the
rank two tensors

$$K^j := \partial e^j, \qquad j = 1, 2, 3,$$
 (2.4.1)

or, in more explicit form, $K^{j}{}_{\alpha\beta} := \partial_{\alpha}e^{j}{}_{\beta}$. The problem is that taken separately the three rank two tensors (2.4.1) are not invariant under rigid rotations of the coframe (2.2.4). The natural way of forming a truly invariant object is to make one rank three tensor out of the three rank two tensors K^{j} according to the formula

$$K := \delta_{jk} e^j \otimes K^k = \delta_{jk} e^j \otimes \partial e^k.$$
(2.4.2)

The rank three tensor K is invariant under rigid rotations of the coframe (2.2.4) and, moreover, the individual rank two tensors K^j can be recovered from K as $K^j{}_{\gamma\delta} = e^{j\alpha}K_{\alpha\gamma\delta}$, where $e^{j\alpha} = e^j{}_{\beta}g^{\alpha\beta}$, so there is no loss of information.

Let us examine the symmetries of the tensor K. Observe that formula (2.2.1) implies

$$0 = \partial_{\alpha}g_{\beta\gamma} = \partial_{\alpha}(\delta_{jk}e^{j}{}_{\beta}e^{k}{}_{\gamma}) = \delta_{jk}(\partial_{\alpha}e^{j}{}_{\beta})e^{k}{}_{\gamma} + \delta_{jk}e^{j}{}_{\beta}(\partial_{\alpha}e^{k}{}_{\gamma}) = K_{\gamma\alpha\beta} + K_{\beta\alpha\gamma}$$

which means that the rank three tensor K is antisymmetric in the first and third indices,

$$K_{\gamma\alpha\beta} = -K_{\beta\alpha\gamma}.\tag{2.4.3}$$

Now, let us introduce another rank three tensor

$$T := \delta_{ik} e^j \otimes de^k \tag{2.4.4}$$

where d stands for the exterior derivative, see Definition 5.5, Section 5.4.2, in [53]. The tensor (2.4.4) is obviously antisymmetric in the second and third indices

$$T_{\alpha\beta\gamma} = -T_{\alpha\gamma\beta} \tag{2.4.5}$$

and is expressed via our original deformation tensor (2.4.2) as

$$T_{\alpha\beta\gamma} = K_{\alpha\beta\gamma} - K_{\alpha\gamma\beta}. \tag{2.4.6}$$

Formulae (2.4.6) and (2.4.3) imply

$$T_{\alpha\beta\gamma} = K_{\alpha\beta\gamma} + K_{\beta\gamma\alpha},$$

$$T_{\gamma\alpha\beta} = K_{\gamma\alpha\beta} + K_{\alpha\beta\gamma},$$

$$T_{\beta\gamma\alpha} = K_{\beta\gamma\alpha} + K_{\gamma\alpha\beta}$$

where the last two identities were obtained from the first one by a cyclic relabelling of tensor indices. Adding up the first and second identities and subtracting the third one we get

$$K_{\alpha\beta\gamma} = \frac{1}{2}(T_{\alpha\beta\gamma} + T_{\gamma\alpha\beta} - T_{\beta\gamma\alpha}) = \frac{1}{2}(T_{\alpha\beta\gamma} + T_{\beta\alpha\gamma} + T_{\gamma\alpha\beta})$$
(2.4.7)

(here we also used (2.4.5)). Note that the argument carried out above is a rephrasing of the standard argument that for a metric compatible affine connection contortion can be expressed via torsion, see Section 7.2.6 in [53].

Formulae (2.4.6) and (2.4.7) show that the tensors K and T are expressed via each other so either of them can be used as a measure of rotational deformations. We choose to use the tensor T because it has a clear geometric meaning: it is the torsion of the Teleparallel connection generated by the coframe e, see Appendix A of [15] for a concise exposition. An additional advantage of using the tensor T is that the definition (2.4.4) of this tensor does not require the use of covariant derivatives so it works when the metric g appearing in formula (2.2.1) is not assumed to be Euclidean. The latter was important for Einstein and Cartan who arrived at the mathematical model similar to the one described in in Chapter 2 coming from general relativity. Recall that in general relativity the metric plays the role of dynamical variable so for someone with a relativistic background assuming the metric to be Euclidean (i.e. space to be flat) is unnatural. Starting from Einstein's works [79] torsion is traditionally used as a measure of deformations when modelling elastic continua with rotations. We shall follow this tradition and construct our potential energy as a function(al) of T. However, before writing down the formula for potential energy we will simplify matters by using the fact that we are working in three dimensions (our previous arguments were dimension-independent).

Applying the Hodge star (2.A.1) in the second and third indices we switch from the original torsion tensor T to the tensor

$${}^{*}_{\alpha\beta} := \frac{1}{2} T_{\alpha}{}^{\gamma\delta} \varepsilon_{\gamma\delta\beta}.$$
(2.4.8)

Of course, the tensor T can be recovered from T as

$$T_{\alpha\beta\gamma} = \overset{*}{T}_{\alpha}{}^{\delta}\varepsilon_{\delta\beta\gamma}.$$
 (2.4.9)

Formulae (2.4.8) and (2.4.9) show that the tensors T and $\overset{*}{T}$ are expressed via each other so either of them can be used as a measure of rotational deformations. We choose to use the tensor $\overset{*}{T}$ because it has lower rank, two instead of three.

Formulae (2.4.4) and (2.4.8) imply

$$\overset{*}{T} = \delta_{jk} e^{j} \otimes * de^{k} = \delta_{jk} e^{j} \otimes \operatorname{curl} e^{k}.$$
(2.4.10)

We see that $\stackrel{*}{T}$ is a rank two tensor without any symmetries and with arbitrary trace. This is the tensor we will be using as a measure of rotational deformations when writing down the formula for potential energy. The tensor $\stackrel{*}{T}$ is sometimes called the *dislocation density tensor* [49].

We give for reference a more explicit version of formula (2.4.10):

$${}^{*}_{T_{\alpha\beta}} = \sum_{j=1}^{3} \begin{pmatrix} e^{j}_{1}\partial_{2}e^{j}_{3} - e^{j}_{1}\partial_{3}e^{j}_{2} & e^{j}_{1}\partial_{3}e^{j}_{1} - e^{j}_{1}\partial_{1}e^{j}_{3} & e^{j}_{1}\partial_{1}e^{j}_{2} - e^{j}_{1}\partial_{2}e^{j}_{1} \\ e^{j}_{2}\partial_{2}e^{j}_{3} - e^{j}_{2}\partial_{3}e^{j}_{2} & e^{j}_{2}\partial_{3}e^{j}_{1} - e^{j}_{2}\partial_{1}e^{j}_{3} & e^{j}_{2}\partial_{1}e^{j}_{2} - e^{j}_{2}\partial_{2}e^{j}_{1} \\ e^{j}_{3}\partial_{2}e^{j}_{3} - e^{j}_{3}\partial_{3}e^{j}_{2} & e^{j}_{3}\partial_{3}e^{j}_{1} - e^{j}_{3}\partial_{1}e^{j}_{3} & e^{j}_{3}\partial_{1}e^{j}_{2} - e^{j}_{3}\partial_{2}e^{j}_{1} \end{pmatrix}.$$

$$(2.4.11)$$

2.4.2 Irreducible decomposition of rotational deformations

Recall the logic of classical linear elasticity [50]: after identifying the measure of deformation one decomposes it into irreducible pieces. We follow this logic by decomposing the tensor $\stackrel{*}{T}$ into irreducible pieces. The construction presented below is similar to [50], the only difference being that instead of a symmetric rank two tensor, strain, we deal with a rank two tensor, $\stackrel{*}{T}$, without any symmetries.

Decomposing the rank two tensor $\stackrel{*}{T}$ into irreducible pieces means the following. We fix a point in \mathbb{R}^3 and at this point look at all real rank two tensors P. Such tensors can be viewed as elements of a real 9-dimensional vector space V equipped with inner product

$$(P,Q)_V := P_{\alpha\beta}Q^{\alpha\beta} \tag{2.4.12}$$

and corresponding norm

$$||P||_V = \sqrt{(P,P)_V} = \sqrt{P_{\alpha\beta}P^{\alpha\beta}}.$$
 (2.4.13)

Let us now examine what happens when we rotate our Cartesian coordinate system x^{α} , i.e. when we perform a linear change of coordinates preserving the metric $g_{\alpha\beta}$ and orientation. The components of our tensors $P_{\alpha\beta}$ change in a particular way under rotations of the coordinate system, so we get an action of the group SO(3) on the vector space V. Looking for irreducible pieces of torsion means identifying subspaces of V which are invariant under the action of the group SO(3), i.e. which map into themselves, and which do not contain smaller nontrivial invariant subspaces. In our case the invariant subspaces can be shown to be

- the 1-dimensional subspace of real rank two tensors proportional to the metric,
- the 3-dimensional subspace of real antisymmetric rank two tensors and
- the 5-dimensional subspace of real symmetric trace-free rank two tensors.

These three subspaces are irreducible and mutually orthogonal in the inner product (2.4.12); explicit calculations are omitted for brevity.

Our rank two tensor $\stackrel{*}{T}$ can now be written as a sum of three irreducible pieces

$${\stackrel{*}{T}} = {\stackrel{*}{T}}{}^{\text{ax}} + {\stackrel{*}{T}}{}^{\text{vec}} + {\stackrel{*}{T}}{}^{\text{ten}}$$
 (2.4.14)

where

$$\stackrel{*}{T}_{\alpha\beta}^{\mathrm{ax}} := \frac{\stackrel{*}{T}\gamma_{\gamma}}{3} g_{\alpha\beta}, \qquad (2.4.15)$$

$$\stackrel{*}{T}_{\alpha\beta}^{\text{vec}} := \frac{\stackrel{*}{T}_{\alpha\beta} - \stackrel{*}{T}_{\beta\alpha}}{2}, \qquad (2.4.16)$$

$$\overset{*}{T}_{\alpha\beta}^{\text{ten}} := \overset{*}{T}_{\alpha\beta} - \overset{*}{T}_{\alpha\beta}^{\text{ax}} - \overset{*}{T}_{\alpha\beta}^{\text{vec}} = \frac{\overset{*}{T}_{\alpha\beta} + \overset{*}{T}_{\beta\alpha}}{2} - \frac{\overset{*}{T}_{\gamma}}{3} g_{\alpha\beta} .$$
 (2.4.17)

We label the irreducible pieces (2.4.15), (2.4.16) and (2.4.17) by the adjectives *axial, vector* and *tensor* respectively, which is terminology traditional in alternative theories of gravity [39]. Note that authors working in "mainstream" Cosserat elasticity use different terminology. Say, Neff's [46, 57, 58, 59, 60, 61, 62] use of the adjective "axial" corresponds to our vector piece (2.4.16).

2.4.3 Formula for potential energy

Following the logic of classical linear elasticity [50] we now write down the explicit formula for potential energy:

$$P(t) = \int \left(c^{\mathrm{ax}} \| T^{\mathrm{ax}} \|_{V}^{2} + c^{\mathrm{vec}} \| T^{\mathrm{vec}} \|_{V}^{2} + c^{\mathrm{ten}} \| T^{\mathrm{ten}} \|_{V}^{2} \right) \rho \, dx^{1} dx^{2} dx^{3} \tag{2.4.18}$$

where c^{ax} , c^{vec} and c^{ten} are some nonnegative constants (elastic moduli), not all zero, and $\|\cdot\|_{V}$ is the norm (2.4.13). Comparing our formula (2.4.18) with formula (4.3) from [50] we see a difference between classical and rotational elasticity: classical elasticity involves two elastic moduli whereas rotational elasticity involves three. The extra elastic modulus c^{vec} is needed because the tensor $\stackrel{*}{T}$ which we use as measure of rotational deformations is not necessarily symmetric.

Formula (2.4.18) is traditionally used in Teleparallelism. This formula already appears in the original papers of Einstein [79], though for some reason¹ Einstein did not include the axial term $c^{ax} \|_{T}^{*ax} \|_{2}^{2}$. Subsequent authors always used three terms, see, for example, formula (26) in [39].

2.4.4 Simplifying the formula for potential energy

Let us introduce the (pseudo)scalar

$$f := \overset{*}{T}{}^{\alpha}{}_{\alpha} \tag{2.4.19}$$

and the vector

$$v_{\alpha} := T^{\beta\gamma} \varepsilon_{\beta\gamma\alpha} \,. \tag{2.4.20}$$

Formulae (2.4.13)–(2.4.17), (2.4.19) and (2.4.20) imply

$$\|\overset{*}{T}^{\mathrm{ax}}\|_{V}^{2} = \frac{1}{3}f^{2}, \qquad (2.4.21)$$

¹The reason could be that Einstein was primarily interested in providing a geometric interpretation of electromagnetism and might have felt that the axial term would not contribute to the electromagnetic field.

$$\|\overset{*}{T}^{\text{vec}}\|_{V}^{2} = \frac{1}{2} \|v\|^{2}, \qquad (2.4.22)$$

$$\|\overset{*}{T}^{\text{ten}}\|_{V}^{2} = \|\overset{*}{T}\|_{V}^{2} - \frac{1}{3}f^{2} - \frac{1}{2}\|v\|^{2}.$$
 (2.4.23)

Substituting (2.4.11) into (2.4.19) and (2.4.20) we get more explicit formulae for f and v:

$$f = \sum_{j=1}^{3} (e^{j}{}_{1}\partial_{2}e^{j}{}_{3} - e^{j}{}_{1}\partial_{3}e^{j}{}_{2} + e^{j}{}_{2}\partial_{3}e^{j}{}_{1} - e^{j}{}_{2}\partial_{1}e^{j}{}_{3} + e^{j}{}_{3}\partial_{1}e^{j}{}_{2} - e^{j}{}_{3}\partial_{2}e^{j}{}_{1}), \quad (2.4.24)$$

$$v_{\alpha} = \sum_{j=1}^{3} \begin{pmatrix} e^{j}{}_{2}\partial_{1}e^{j}{}_{2} - e^{j}{}_{2}\partial_{2}e^{j}{}_{1} - e^{j}{}_{3}\partial_{3}e^{j}{}_{1} + e^{j}{}_{3}\partial_{1}e^{j}{}_{3} \\ e^{j}{}_{3}\partial_{2}e^{j}{}_{3} - e^{j}{}_{3}\partial_{3}e^{j}{}_{2} - e^{j}{}_{1}\partial_{1}e^{j}{}_{2} + e^{j}{}_{1}\partial_{2}e^{j}{}_{1} \\ e^{j}{}_{1}\partial_{3}e^{j}{}_{1} - e^{j}{}_{1}\partial_{1}e^{j}{}_{3} - e^{j}{}_{2}\partial_{2}e^{j}{}_{3} + e^{j}{}_{2}\partial_{3}e^{j}{}_{2} \end{pmatrix}.$$
 (2.4.25)

Substituting formulae (2.4.19)–(2.4.21) into formula (2.4.18) we get

$$P(t) = \int \left(\frac{c^{\text{ax}} - c^{\text{ten}}}{3}f^2 + \frac{c^{\text{vec}} - c^{\text{ten}}}{2}\|v\|^2 + c^{\text{ten}}\tilde{T}_{\alpha\beta}^*\tilde{T}^{\alpha\beta}\right)\rho\,dx^1dx^2dx^3\,.$$
 (2.4.26)

The advantage of writing potential energy in the form (2.4.26) is that the geometric quantities f, v and $\stackrel{*}{T}$ appearing in this formula have relatively compact explicit representations (2.4.24), (2.4.25) and (2.4.11).

2.5 Lagrangian of Rotational elasticity

We combine our potential energy (2.4.26) and kinetic energy (2.3.1) to form the action (variational function) of dynamic rotational elasticity

$$S_{\rm Re}(e,\rho) = \int (P(t) - K(t))dt = \int L_{\rm Re}(e,\rho) \, dt \, dx^1 dx^2 dx^3 \tag{2.5.1}$$

where

$$L_{\rm Re}(e,\rho) = \left(\frac{c^{\rm ax} - c^{\rm ten}}{3}f^2 + \frac{c^{\rm vec} - c^{\rm ten}}{2}\|v\|^2 + c^{\rm ten}T^*_{\alpha\beta}T^{\alpha\beta} - c^{\rm kin}\|\omega\|^2\right)\rho \quad (2.5.2)$$

is the Lagrangian density. Recall that the geometric quantities f, v, T and ω appearing in formula (2.5.2) are defined by formulae (2.4.24), (2.4.25), (2.4.11) and (2.3.3) respectively.

An equivalent matrix representation of the Lagrangian density (2.5.2) is provided in Appendix 2.E.

It is worth noting that the choice of parameters in the Lagrangian of Cosserat elasticity is highly controversial. The Cosserat model as such is not accepted in the mainstream elasticity theory community and it has never been shown to be superior to classical models. On the positive side, the Cosserat model allows for the rigorous derivation [59, 60] of certain plate theories.

A comprehensive analysis of the equations of Cosserat elasticity in the complete (with both displacements and microrotations) setting was performed in [46, 57, 58, 59, 60, 61, 62]. In particular, the authors of [62] performed a scale analysis which led them to the conclusion that the constants c^{ax} and c^{vec} appearing in formula (2.5.2) should be zero. We do not believe that the arguments presented in [62] are directly relevant to our case because the authors of [62] assumed the strain part of the Lagrangian to be nonzero. In other words, the authors of [62] had in mind a realistic macroscopic material whereas we do not aim at modelling such a material.

2.6 Reformulating the problem in the language of spinors

Our field equations (Euler–Lagrange equations) are obtained by varying the action (2.5.1) with respect to the coframe e and density ρ . Varying with respect to the density ρ is easy: this gives the field equation

$$\frac{c^{\text{ax}} - c^{\text{ten}}}{3}f^2 + \frac{c^{\text{vec}} - c^{\text{ten}}}{2}\|v\|^2 + c^{\text{ten}} \mathring{T}^{\alpha\beta}_{\alpha\beta} \mathring{T}^{\alpha\beta} - c^{\text{kin}}\|\omega\|^2 = 0$$
(2.6.1)

which is equivalent to $L_{\text{Re}}(e, \rho) = 0$. Varying with respect to the coframe e is more difficult because we have to maintain the kinematic constraint (2.2.1).

This technical difficulty can be overcome by switching to a different dynamical variable. Namely, it is known [23] that in dimension 3 a coframe e and a (positive) density ρ are equivalent to a 2-component complex-valued spinor field

$$\xi = \xi^a = \begin{pmatrix} \xi^1 \\ \xi^2 \end{pmatrix} \tag{2.6.2}$$

modulo the sign of ξ (i.e. modulo a factor ± 1). The explicit formulae establishing this equivalence are

$$\rho = \bar{\xi}^{\dot{a}} \sigma_{0\dot{a}b} \xi^{b}, \qquad (2.6.3)$$

$$(e^1 + ie^2)_{\alpha} = \rho^{-1} \epsilon^{\dot{c}\dot{b}} \sigma_{0\dot{b}a} \xi^a \sigma_{\alpha\dot{c}d} \xi^d, \qquad (2.6.4)$$

$$e^{3}{}_{\alpha} = \rho^{-1} \bar{\xi}^{\dot{a}} \sigma_{\alpha \dot{a} b} \xi^{b}. \tag{2.6.5}$$

Here σ are Pauli matrices and ϵ is "metric spinor" (see (2.A.2)–(2.A.4)), the free tensor index α runs through the values 1, 2, 3, and the spinor summation indices run through the values 1, 2 or 1, 2. The advantage of switching to a spinor field ξ is that there are no kinematic constraints on its components, so the derivation of field equations becomes straightforward.

We give for reference more explicit versions of formulae (2.6.3)-(2.6.5):

$$\rho = \bar{\xi}^{\dot{1}} \xi^1 + \bar{\xi}^{\dot{2}} \xi^2, \qquad (2.6.6)$$

$$(e^{1} + ie^{2}) = (\bar{\xi}^{\dot{i}}\xi^{1} + \bar{\xi}^{\dot{2}}\xi^{2})^{-1} \begin{pmatrix} (\xi^{1})^{2} - (\xi^{2})^{2} \\ i(\xi^{1})^{2} + i(\xi^{2})^{2} \\ -2\xi^{1}\xi^{2} \end{pmatrix}, \qquad (2.6.7)$$
$$e^{3} = (\bar{\xi}^{\dot{i}}\xi^{1} + \bar{\xi}^{\dot{2}}\xi^{2})^{-1} \begin{pmatrix} \bar{\xi}^{\dot{2}}\xi^{1} + \bar{\xi}^{\dot{1}}\xi^{2} \\ i\bar{\xi}^{\dot{2}}\xi^{1} - i\bar{\xi}^{\dot{1}}\xi^{2} \\ \bar{\xi}^{\dot{1}}\xi^{1} - \bar{\xi}^{\dot{2}}\xi^{2} \end{pmatrix}. \qquad (2.6.8)$$

Let us rewrite the geometric quantities $f, v, \overset{*}{T}$ and ω appearing in formula (2.5.2) in terms of the spinor field ξ . The spinor representation of angular velocity ω was derived in [23]:

$$\omega_{\alpha} = i \frac{\bar{\xi}^{\dot{a}} \sigma_{\alpha \dot{a} b} \partial_t \xi^b - \xi^b \sigma_{\alpha \dot{a} b} \partial_t \bar{\xi}^{\dot{a}}}{\bar{\xi}^{\dot{c}} \sigma_{0 \dot{c} d} \xi^d}$$
(2.6.9)

or, more explicitly,

$$\omega = \frac{1}{\bar{\xi}^{\dot{i}}\xi^{1} + \bar{\xi}^{\dot{2}}\xi^{2}} \begin{pmatrix} i\bar{\xi}^{\dot{2}}\partial_{t}\xi^{1} + i\bar{\xi}^{\dot{1}}\partial_{t}\xi^{2} \\ -\bar{\xi}^{\dot{2}}\partial_{t}\xi^{1} + \bar{\xi}^{\dot{1}}\partial_{t}\xi^{2} \\ i\bar{\xi}^{\dot{1}}\partial_{t}\xi^{1} - i\bar{\xi}^{\dot{2}}\partial_{t}\xi^{2} \end{pmatrix} + \text{c.c.}$$
(2.6.10)

where the "c.c." stands for "complex conjugate term". The spinor representation of the tensor $\stackrel{*}{T}$ is derived in Appendix 2.B, see formula (2.B.1) or its more explicit version (2.B.7). Substituting (2.B.1) or (2.B.7) into (2.4.19) and (2.4.20) we arrive at spinor representations for the (pseudo)scalar f and vector v:

$$f = -2i \frac{\bar{\xi}^{\dot{a}} \sigma^{\alpha}{}_{\dot{a}b} \partial_{\alpha} \xi^{b} - \xi^{b} \sigma^{\alpha}{}_{\dot{a}b} \partial_{\alpha} \bar{\xi}^{\dot{a}}}{\bar{\xi}^{\dot{c}} \sigma_{0\dot{c}d} \xi^{d}}, \qquad (2.6.11)$$

$$v_{\alpha} = -i\varepsilon_{\beta\gamma\alpha} \frac{\bar{\xi}^{\dot{a}}\sigma^{\beta}{}_{\dot{a}b}\partial^{\gamma}\xi^{b} - \xi^{b}\sigma^{\beta}{}_{\dot{a}b}\partial^{\gamma}\bar{\xi}^{\dot{a}}}{\bar{\xi}^{\dot{c}}\sigma_{0\dot{c}d}\xi^{d}}$$
(2.6.12)

where $\partial^{\alpha} := g^{\alpha\beta} \partial_{\beta} = \partial_{\alpha}$, or, more explicitly,

$$f = \frac{2}{\bar{\xi}^{\dot{1}}\xi^{1} + \bar{\xi}^{\dot{2}}\xi^{2}} (-i\bar{\xi}^{\dot{1}}\partial_{1}\xi^{2} - i\bar{\xi}^{\dot{2}}\partial_{1}\xi^{1} - \bar{\xi}^{\dot{1}}\partial_{2}\xi^{2} + \bar{\xi}^{\dot{2}}\partial_{2}\xi^{1} - i\bar{\xi}^{\dot{1}}\partial_{3}\xi^{1} + i\bar{\xi}^{\dot{2}}\partial_{3}\xi^{2}) + \text{c.c.},$$
(2.6.13)

$$v = \frac{1}{\bar{\xi}^{\bar{i}}\xi^{1} + \bar{\xi}^{\bar{2}}\xi^{2}} \begin{pmatrix} i\bar{\xi}^{\bar{i}}\partial^{2}\xi^{1} - i\bar{\xi}^{\bar{2}}\partial^{2}\xi^{2} - \bar{\xi}^{\bar{i}}\partial^{3}\xi^{2} + \bar{\xi}^{\bar{2}}\partial^{3}\xi^{1} \\ i\bar{\xi}^{\bar{i}}\partial^{3}\xi^{2} + i\bar{\xi}^{\bar{2}}\partial^{3}\xi^{1} - i\bar{\xi}^{\bar{1}}\partial^{1}\xi^{1} + i\bar{\xi}^{\bar{2}}\partial^{1}\xi^{2} \\ \bar{\xi}^{\bar{i}}\partial^{1}\xi^{2} - \bar{\xi}^{\bar{2}}\partial^{1}\xi^{1} - i\bar{\xi}^{\bar{i}}\partial^{2}\xi^{2} - i\bar{\xi}^{\bar{2}}\partial^{2}\xi^{1} \end{pmatrix} + \text{c.c.} \quad (2.6.14)$$

Note that formula (2.6.11) is a rephrasing of formula (B.5) from [23].

From now on we write our action (2.5.1) and Lagrangian density (2.5.2) as $S_{\text{Re}}(\xi)$ and $L_{\text{Re}}(\xi)$ rather than $S_{\text{Re}}(e,\rho)$ and $L_{\text{Re}}(e,\rho)$, thus indicating that we have switched to spinors. The explicit formula for $L_{\text{Re}}(\xi)$ is obtained by substituting formulae (2.6.3), (2.6.11), (2.6.12), (2.B.1) and (2.6.9) into (2.5.2). The nonvanishing spinor field ξ is the new dynamical variable and it will be varied without any constraints.

2.7 Euler–Lagrange equation

Let us perform a formal variation of our spinor field $\xi \mapsto \xi + \delta \xi$, where $\delta \xi$: $\mathbb{R} \times \mathbb{R}^3 \to \mathbb{C}^2$ is an arbitrary (infinitely) smooth function with compact support. Then, after integration by parts, the variation of our action can be written as

$$\delta S_{\rm Re} = \int (F_{\dot{a}}\delta\bar{\xi}^{\dot{a}} + \bar{F}_{a}\delta\xi^{a}) dt dx^{1}dx^{2}dx^{3}$$
(2.7.1)

where F is a dotted spinor field uniquely determined by the undotted spinor field ξ . The Euler–Lagrange for our unknown spinor field ξ is, therefore,

$$F = 0.$$
 (2.7.2)

The map

$$\xi \mapsto F \tag{2.7.3}$$

defines a nonlinear second order partial differential operator in the variables t (time) and x^{α} , $\alpha = 1, 2, 3$ (Cartesian coordinates).

We shall refrain from writing down the Euler–Lagrange equation (2.7.2) explicitly. The reason for this is that in the current chapter we are interested in finding a particular class of solutions for which the procedure is much simpler.

Note that for the special case of a purely axial material, i.e. material with

$$c^{\text{vec}} = 0, \qquad c^{\text{ten}} = 0, \qquad (2.7.4)$$

the Euler-Lagrange equation (2.7.2) was written down explicitly in [23]. The calculations in [23] were carried out under the additional assumption

$$c^{\rm kin} = \frac{4}{3}c^{\rm ax} \tag{2.7.5}$$

which can always be achieved by rescaling time t.

2.8 Plane wave solutions

In this section, as well as in Section 2.9 and Appendix 2.C, we denote time by x^0 and the time derivative by ∂_0 . This will allow us to simplify notation.

We seek solutions of the form

$$\xi(\mathbf{x}) = e^{-i\mathbf{p}\cdot\mathbf{x}}\zeta \tag{2.8.1}$$

where $\zeta \neq 0$ is a constant (complex) spinor and **p** is a constant real covector. Here we use relativistic notation, incorporating time x^0 into our coordinates. This means that $\mathbf{x} = (x^0, x^1, x^2, x^3)$ and $\mathbf{p} = (p_0, p_1, p_2, p_3)$; bold type indicates that we are working in (1+3)-dimensional spacetime. The number $|p_0|$ is the wave frequency and the covector (p_1, p_2, p_3) is the wave vector in original 3-dimensional Euclidean space. The 4-component covector $\mathbf{p} = (p_0, p_1, p_2, p_3)$ has the meaning of relativistic 4-momentum.

Throughout this section, as well as the next one, we assume that

$$p_0 \neq 0 \tag{2.8.2}$$

which means that we are not interested in static (time-independent) solutions. The sign of p_0 can be arbitrary.

Our Euler-Lagrange equation (2.7.2) is highly nonlinear so it is by no means obvious that one can seek solutions in the form of plane waves (2.8.1). Fortunately (and miraculously) this is the case. In order to see this, we rewrite our Euler– Lagrange equation (2.7.2) in equivalent form

$$e^{i\mathbf{p}\cdot\mathbf{x}}F = 0. \tag{2.8.3}$$

Note that the sign in the exponent in (2.8.3) is opposite to that in (2.8.1). We have

Lemma 2.8.1. If the spinor field ξ is a plane wave (2.8.1) then the left-hand side of equation (2.8.3) is constant, i.e. it does not depend on \mathbf{x} .

Of course, Lemma 2.8.1 can be equivalently reformulated as follows: the nonlinear partial differential operator (2.7.3) maps a plane wave (2.8.1) into a plane wave with the same relativistic 4-momentum **p**.

The proof of Lemma 2.8.1 is quite technical and is given in Appendix 2.C.

Lemma 2.8.1 justifies separation of variables, i.e. it reduces the study of the nonlinear partial differential equation (2.7.2) for the unknown spinor field ξ to the study of the rational algebraic equation (2.8.3) for the unknown constant spinor ζ . We suspect that the underlying group-theoretic reason for our nonlinear partial differential equation (2.7.2) admitting separation of variables is the fact that our model is U(1)-invariant, i.e. it is invariant under the multiplication of the spinor field ξ by a complex constant of modulus 1. Hence, it is feasible that one could prove Lemma 2.8.1, as well as Lemma 2.8.2 stated further down in this section, without performing the laborious calculations presented in Appendix 2.C.

We are now faced with the task of writing down the LHS of equation (2.8.3) explicitly and with minimal calculations. To this end we address a seemingly different issue: we examine what happens when we substitute our plane wave (2.8.1) into our Lagrangian density $L_{\text{Re}}(\xi)$ (i.e. (2.5.2) rephrased in terms of spinors, see Section 2.6 for details), rather than the Euler-Lagrange equation (2.7.2).

Substituting formula (2.8.1) into formulae (2.6.3), (2.6.11), (2.6.12), (2.B.1) and (2.6.9) we get

$$\rho = \bar{\zeta}^{\dot{a}} \sigma_{0\dot{a}b} \zeta^{b}, \qquad (2.8.4)$$

$$f = -\frac{4\bar{\zeta}^{\dot{a}}\sigma^{\alpha}{}_{\dot{a}b}p_{\alpha}\zeta^{b}}{\bar{\zeta}^{\dot{c}}\sigma_{0\dot{c}d}\zeta^{d}},\qquad(2.8.5)$$

$$v_{\alpha} = -\frac{2\varepsilon_{\beta\gamma\alpha}\bar{\zeta}^{\dot{a}}\sigma^{\beta}{}_{\dot{a}b}p^{\gamma}\zeta^{b}}{\bar{\zeta}^{\dot{c}}\sigma_{0\dot{c}d}\zeta^{d}},\qquad(2.8.6)$$

$${}^{*}_{\alpha\beta} = 2 \frac{\bar{\zeta}^{\dot{a}} \sigma_{\beta\dot{a}b} p_{\alpha} \zeta^{b} - \bar{\zeta}^{\dot{a}} \sigma^{\gamma}{}_{\dot{a}b} p_{\gamma} \zeta^{b} g_{\alpha\beta}}{\bar{\zeta}^{\dot{c}} \sigma_{0\dot{c}d} \zeta^{d}} , \qquad (2.8.7)$$

$$\omega_{\alpha} = \frac{2\bar{\zeta}^{\dot{a}}\sigma_{\alpha\dot{a}b}p_{0}\zeta^{b}}{\bar{\zeta}^{\dot{c}}\sigma_{0\dot{c}d}\zeta^{d}},\qquad(2.8.8)$$

or, more explicitly,

$$\rho = \bar{\zeta}^{\dot{1}} \zeta^1 + \bar{\zeta}^{\dot{2}} \zeta^2, \qquad (2.8.9)$$

$$f = \frac{4}{\bar{\zeta}^{\dot{i}}\zeta^{1} + \bar{\zeta}^{\dot{2}}\zeta^{2}} \left(p_{1}(-\bar{\zeta}^{\dot{i}}\zeta^{2} - \bar{\zeta}^{\dot{2}}\zeta^{1}) + ip_{2}(\bar{\zeta}^{\dot{i}}\zeta^{2} - \bar{\zeta}^{\dot{2}}\zeta^{1}) + p_{3}(-\bar{\zeta}^{\dot{i}}\zeta^{1} + \bar{\zeta}^{\dot{2}}\zeta^{2}) \right),$$
(2.8.10)

$$v = \frac{2}{\bar{\zeta}^{\dot{i}}\zeta^{1} + \bar{\zeta}^{\dot{2}}\zeta^{2}} \begin{pmatrix} p_{2}\left(\bar{\zeta}^{\dot{i}}\zeta^{1} - \bar{\zeta}^{\dot{2}}\zeta^{2}\right) + ip_{3}\left(\bar{\zeta}^{\dot{i}}\zeta^{2} - \bar{\zeta}^{\dot{2}}\zeta^{1}\right) \\ p_{3}\left(\bar{\zeta}^{\dot{i}}\zeta^{2} + \bar{\zeta}^{\dot{2}}\zeta^{1}\right) + p_{1}\left(-\bar{\zeta}^{\dot{i}}\zeta^{1} + \bar{\zeta}^{\dot{2}}\zeta^{2}\right) \\ ip_{1}\left(-\bar{\zeta}^{\dot{i}}\zeta^{2} + \bar{\zeta}^{\dot{2}}\zeta^{1}\right) + p_{2}\left(-\bar{\zeta}^{\dot{i}}\zeta^{2} - \bar{\zeta}^{\dot{2}}\zeta^{1}\right) \end{pmatrix}, \quad (2.8.11)$$

$$\begin{pmatrix} {}^{*}_{11} \\ {}^{*}_{12} \\ {}^{*}_{13} \\ {}^{*}_{21} \\ {}^{*}_{22} \\ {}^{*}_{23} \\ {}^{*}_{31} \\ {}^{*}_{32} \\ {}^{*}_{33} \end{pmatrix} = \frac{2}{\bar{\zeta}^{\bar{i}}\zeta^{1} + \bar{\zeta}^{\bar{2}}\zeta^{2}} \begin{pmatrix} ip_{2}\left(\bar{\zeta}^{\bar{i}}\zeta^{2} - \bar{\zeta}^{\bar{2}}\zeta^{1}\right) + p_{3}\left(-\bar{\zeta}^{\bar{i}}\zeta^{1} + \bar{\zeta}^{\bar{2}}\zeta^{2}\right) \\ p_{1}\left(\bar{\zeta}^{\bar{i}}\zeta^{1} - \bar{\zeta}^{\bar{2}}\zeta^{2}\right) \\ p_{2}\left(\bar{\zeta}^{\bar{i}}\zeta^{2} + \bar{\zeta}^{\bar{2}}\zeta^{1}\right) \\ p_{2}\left(\bar{\zeta}^{\bar{i}}\zeta^{2} + \bar{\zeta}^{\bar{2}}\zeta^{1}\right) \\ p_{3}\left(-\bar{\zeta}^{\bar{i}}\zeta^{1} + \bar{\zeta}^{\bar{2}}\zeta^{2}\right) + p_{1}\left(-\bar{\zeta}^{\bar{i}}\zeta^{2} - \bar{\zeta}^{\bar{2}}\zeta^{1}\right) \\ p_{2}\left(\bar{\zeta}^{\bar{i}}\zeta^{1} - \bar{\zeta}^{\bar{2}}\zeta^{2}\right) \\ p_{2}\left(\bar{\zeta}^{\bar{i}}\zeta^{1} - \bar{\zeta}^{\bar{2}}\zeta^{2}\right) \\ p_{3}\left(\bar{\zeta}^{\bar{i}}\zeta^{2} + \bar{\zeta}^{\bar{2}}\zeta^{1}\right) \\ p_{3}\left(-\bar{\zeta}^{\bar{i}}\zeta^{2} + \bar{\zeta}^{\bar{2}}\zeta^{1}\right) \\ p_{3}\left(-\bar{\zeta}^{\bar{i}}\zeta^{2} - \bar{\zeta}^{\bar{2}}\zeta^{1}\right) \\ p_{1}\left(-\bar{\zeta}^{\bar{i}}\zeta^{2} - \bar{\zeta}^{\bar{2}}\zeta^{1}\right) + ip_{2}\left(\bar{\zeta}^{\bar{i}}\zeta^{2} - \bar{\zeta}^{\bar{2}}\zeta^{1}\right) \end{pmatrix}, \quad (2.8.12)$$

$$\omega = \frac{2p_0}{\bar{\zeta}^{\dot{i}}\zeta^1 + \bar{\zeta}^{\dot{2}}\zeta^2} \begin{pmatrix} -i\bar{\zeta}^{\dot{i}}\zeta^2 - i\bar{\zeta}^{\dot{2}}\zeta^1 \\ -\bar{\zeta}^{\dot{i}}\zeta^2 + \bar{\zeta}^{\dot{2}}\zeta^1 \\ -i\bar{\zeta}^{\dot{i}}\zeta^1 + i\bar{\zeta}^{\dot{2}}\zeta^2 \end{pmatrix}.$$
 (2.8.13)

Substituting formulae (2.8.4)–(2.8.8) or their more explicit versions (2.8.9)–(2.8.13) into formula (2.5.2) we arrive at a Lagrangian density $L_{\text{Re}}(\zeta; \mathbf{p})$ which does not depend on \mathbf{x} . The self-contained formula for this Lagrangian density, written in terms of 4-momentum \mathbf{p} and 4-current

$$\mathbf{j}_{\boldsymbol{\alpha}} := \bar{\zeta}^{\dot{a}} \sigma_{\boldsymbol{\alpha} \dot{a} b} \zeta^{b}, \qquad (2.8.14)$$

is

$$L_{\rm Re}(\zeta; \mathbf{p}) = \frac{2}{j_0} \left(c^{\rm vec} + c^{\rm ten} \right) \|p\|^2 \|j\|^2 + \frac{4}{j_0} \left(\frac{4}{3} c^{\rm ax} - \frac{1}{2} c^{\rm vec} + \frac{1}{6} c^{\rm ten} \right) (j \cdot p)^2 - \frac{4}{j_0} c^{\rm kin} p_0^2 \|j\|^2. \quad (2.8.15)$$

Here we write our 4-covectors as $\mathbf{p} = (p_0, p)$ and $\mathbf{j} = (j_0, j)$, where p and j are 3-covectors.

We view the 4-momentum \mathbf{p} as a parameter and the constant spinor $\zeta \neq 0$ as the dynamical variable. The Lagrangian density $L_{\text{Re}}(\zeta; \mathbf{p})$ is a smooth function of $\text{Re}\zeta$ and $\text{Im}\zeta$, so varying ζ we get

$$\delta L_{\rm Re} = G_{\dot{a}} \delta \bar{\zeta}^{\dot{a}} + \bar{G}_a \delta \zeta^a \tag{2.8.16}$$

where G is a dotted constant spinor expressed via the partial derivatives of $L_{\text{Re}}(\zeta; \mathbf{p})$ with respect to $\text{Re} \zeta$ and $\text{Im} \zeta$. It is natural to ask the question: what is the relationship between the spinor field F appearing in formula (2.7.1) and the constant spinor G appearing in formula (2.8.16)? The answer is given by

Lemma 2.8.2. If the spinor field ξ is a plane wave (2.8.1) then $G = e^{i\mathbf{p}\cdot\mathbf{x}}F$.

The proof of Lemma 2.8.2 is presented in Appendix 2.C.

Lemma 2.8.2 reduces the construction of plane wave solutions of rotational elasticity to finding the critical, with respect to ζ , points of the function $L_{\text{Re}}(\zeta; \mathbf{p})$. Varying (2.8.15), we arrive at the following equation for critical points:

$$\frac{4}{j_{0}} \left(c^{\text{vec}} + c^{\text{ten}} \right) \|p\|^{2} j^{\alpha} \sigma_{\alpha \dot{a} b} \zeta^{b} - \frac{2}{j_{0}^{2}} \left(c^{\text{vec}} + c^{\text{ten}} \right) \|p\|^{2} \|j\|^{2} \sigma_{0 \dot{a} b} \zeta^{b}
+ \frac{8}{j_{0}} \left(\frac{4}{3} c^{\text{ax}} - \frac{1}{2} c^{\text{vec}} + \frac{1}{6} c^{\text{ten}} \right) (j \cdot p) p^{\alpha} \sigma_{\alpha \dot{a} b} \zeta^{b} - \frac{4}{j_{0}^{2}} \left(\frac{4}{3} c^{\text{ax}} - \frac{1}{2} c^{\text{vec}} + \frac{1}{6} c^{\text{ten}} \right) (j \cdot p)^{2} \sigma_{0 \dot{a} b} \zeta^{b}
- \frac{8}{j_{0}} c^{\text{kin}} p_{0}^{2} j^{\alpha} \sigma_{\alpha \dot{a} b} \zeta^{b} + \frac{4}{j_{0}^{2}} c^{\text{kin}} p_{0}^{2} \|j\|^{2} \sigma_{0 \dot{a} b} \zeta^{b} = 0. \quad (2.8.17)$$

Recall that the 4-current $\mathbf{j} = (j_0, j)$ appearing in the above equation is defined in accordance with formula (2.8.14).

It now remains to find the 4-momenta **p** and spinors $\zeta \neq 0$ which satisfy equation (2.8.17). We carry out the analysis of equation (2.8.17) assuming that

$$\rho = j_0 = \bar{\zeta}^{\dot{1}} \zeta^1 + \bar{\zeta}^{\dot{2}} \zeta^2 = 1.$$
(2.8.18)

Condition (2.8.18) is a normalisation of the density: general plane wave solutions differ from those satisfying condition (2.8.18) by a real scaling factor. Furthermore, we assume that

$$\zeta^b = \begin{pmatrix} 1\\ 0 \end{pmatrix}. \tag{2.8.19}$$

Indeed, any spinor ζ satisfying condition (2.8.18) can be transformed into the spinor (2.8.19) by the linear transformation

$$\zeta \mapsto U\zeta \tag{2.8.20}$$

where U is a special (det U = 1) unitary matrix. The transformation (2.8.20) leads to a rotation of the spatial part of the 4-current (2.8.14), so plane wave solutions with general ζ differ from those with ζ of the form (2.8.19) by a rotation. Substituting (2.8.17), (2.8.19), (2.A.3), (2.A.4) and (2.8.13) into (2.8.16) we get

$$4 (c^{\text{vec}} + c^{\text{ten}}) \|p\|^2 \begin{pmatrix} 1 \\ 0 \end{pmatrix} - 2 (c^{\text{vec}} + c^{\text{ten}}) \|p\|^2 \begin{pmatrix} 1 \\ 0 \end{pmatrix}$$
$$+ 8 \left(\frac{4}{3}c^{\text{ax}} - \frac{1}{2}c^{\text{vec}} + \frac{1}{6}c^{\text{ten}}\right) p_3 \begin{pmatrix} p_3 \\ p_1 + ip_2 \end{pmatrix} - 4 \left(\frac{4}{3}c^{\text{ax}} - \frac{1}{2}c^{\text{vec}} + \frac{1}{6}c^{\text{ten}}\right) p_3^2 \begin{pmatrix} 1 \\ 0 \end{pmatrix}$$
$$- 4c^{\text{kin}} \begin{pmatrix} p_0^2 \\ 0 \end{pmatrix} = 0,$$

or, equivalently,

$$2\left(c^{\text{vec}} + c^{\text{ten}}\right) \begin{pmatrix} \|p\|^2\\ 0 \end{pmatrix} + 4\left(\frac{4}{3}c^{\text{ax}} - \frac{1}{2}c^{\text{vec}} + \frac{1}{6}c^{\text{ten}}\right) \begin{pmatrix} p_3^2\\ 2p_3(p_1 + ip_2) \end{pmatrix} - 4c^{\text{kin}} \begin{pmatrix} p_0^2\\ 0 \end{pmatrix} = 0$$
(2.8.21)

Put

$$v_1 := \sqrt{\frac{4c^{\text{ax}} + 2c^{\text{ten}}}{3c^{\text{kin}}}}, \qquad v_2 := \sqrt{\frac{c^{\text{vec}} + c^{\text{ten}}}{2c^{\text{kin}}}}.$$
 (2.8.22)

Note that because we assumed our three elastic moduli to be nonnegative and not all zero, our v_1 and v_2 are nonnegative and not both zero. Using (2.8.22) we can now rewrite equation (2.8.21) in more compact form

$$v_2^2 \begin{pmatrix} \|p\|^2\\ 0 \end{pmatrix} + (v_1^2 - v_2^2) \begin{pmatrix} p_3^2\\ 2p_3(p_1 + ip_2) \end{pmatrix} - \begin{pmatrix} p_0^2\\ 0 \end{pmatrix} = 0.$$
 (2.8.23)

The analysis of equation (2.8.23) is elementary and the outcome is summarised in the following theorem, which is the main result of Chapter 2.

Theorem 2.8.1. Plane wave solutions of rotational elasticity can, up to rescaling and rotation, be explicitly written down in the form (2.8.1), (2.8.19) with arbitrary nonzero p_0 and $p = (p_1, p_2, p_3)$ determined as follows.

• If $v_1 > 0$ and $v_2 > 0$ and $v_1 \neq v_2$ then we have two possibilities:

$$- p = \left(0, 0, \pm \frac{p_0}{v_1}\right)$$
 (type 1 wave), or

$$-p = \left(\frac{|p_0|}{v_2}\cos\varphi, \frac{|p_0|}{v_2}\sin\varphi, 0\right) \text{ where } \varphi \in \mathbb{R} \text{ is arbitrary (type 2 wave).}$$

• If $v_1 > 0$ and $v_2 > 0$ and $v_1 = v_2$ then p is an arbitrary 3-vector satisfying $||p|| = \frac{|p_0|}{v_1}$.

• If
$$v_1 > 0$$
 and $v_2 = 0$ then $p = \left(0, 0, \pm \frac{p_0}{v_1}\right)$.

• If $v_1 = 0$ and $v_2 > 0$ then $p = \left(\frac{|p_0|}{v_2}\cos\varphi, \frac{|p_0|}{v_2}\sin\varphi, 0\right)$ where $\varphi \in \mathbb{R}$ is arbitrary.

Theorem 2.8.1 shows that rotational elasticity, like classical linear elasticity, produces two distinct types of plane wave solutions. We call these solutions type 1 and type 2 and they propagate with velocities v_1 and v_2 respectively, with v_1 and v_2 given by formulae (2.8.22).

However, unlike with classical linear elasticity, in rotational elasticity the two wave velocities, v_1 and v_2 , are not ordered, i.e. we do not know a priori which one, v_1 or v_2 , is bigger. The reason the two wave velocities are not ordered is because rotational elasticity has three elastic moduli compared to the two elastic moduli of classical linear elasticity. The "extra" elastic modulus is c^{vec} , the one associated with the antisymmetric part of the rank two tensor T which we use as a measure of rotational deformations. If we set $c^{\text{vec}} = 0$, we end up with the inequality

$$v_1 \ge \sqrt{\frac{4}{3}} \, v_2 \tag{2.8.24}$$

similar to the well known inequality from classical linear elasticity, see formula (22.5) in [50].

Throughout this chapter we considered our equations of rotational elasticity in the whole of \mathbb{R}^3 . It would be natural to examine, in due course, boundary value problems in domains of \mathbb{R}^3 , starting with the static case. We do not yet have existence or uniqueness theorems for static boundary value problems generated by the Lagrangian of rotational elasticity. Note that for the complete (with both displacements and microrotations) Cosserat model theorems of this type have been proved in [46, 57, 58, 59, 60, 61, 62]. Note also that the result of [63] may be particularly useful in the proof of possible existence theorems for rotational elasticity: as shown in [63], partial derivatives of the coframe are bounded from above by the norm of torsion.

2.9 The massless Dirac equation

In this section we consider a purely axial material (2.7.4), (2.7.5). Substituting (2.7.4) and (2.7.5) into (2.8.22) we get $v_1 = 1$ and $v_2 = 0$, so a purely axial material supports only type 1 waves. Throughout this section we also retain the assumption (2.8.2).

Note that a purely axial material has the remarkable property that its potential energy is invariant under conformal rescalings of the spatial metric by an arbitrary positive scalar function. We do not elaborate on this issue in the current chapter because we chose to work with a specific (standard Euclidean) metric. The appropriate arguments are presented in Section 2 of [23].

Our aim is to compare our model with the linear partial differential equation (or, more precisely, system of two linear partial differential equations)

$$i(\pm\sigma^{0}{}_{\dot{a}b}\partial_{0} + \sigma^{\alpha}{}_{\dot{a}b}\partial_{\alpha})\xi^{b} = 0.$$
(2.9.1)

Here ∂_0 is the time derivative, ∂_{α} are spatial derivatives, σ are Pauli matrices (2.A.3), (2.A.4), the free spinor index \dot{a} runs through the values $\dot{1}, \dot{2}$, summation is carried out over the tensor index $\alpha = 1, 2, 3$ as well as the spinor index b = 1, 2, and ξ is the unknown spinor field. We give for reference a more explicit version of equation (2.9.1):

$$i \begin{pmatrix} \mp \partial_0 + \partial_3 & \partial_1 - i\partial_2 \\ \partial_1 + i\partial_2 & \mp \partial_0 - \partial_3 \end{pmatrix} \begin{pmatrix} \xi^1 \\ \xi^2 \end{pmatrix} = 0.$$
 (2.9.2)

Equation (2.9.1) is called *the massless Dirac equation* or *the Weyl equation*. This equation is the accepted mathematical model for a massless neutrino field. The two choices of sign in (2.9.1) give two versions of the massless Dirac equation which differ by time reversal. Thus, we have a pair of massless Dirac equations.

We want to compare plane wave solutions (see (2.8.1)) of our model with those of the massless Dirac equation. As both models are invariant under the rescaling of the spinor field by a positive real constant as well as the rotations of Euclidean 3-space, it is sufficient to compare plane wave solutions for ζ of the form (2.8.19). Substituting (2.8.1) and (2.8.19) into (2.9.1) or its more explicit version (2.9.2) we get $p = (0, 0, \pm p_0)$ which is exactly what Theorem 2.8.1 gives us. Thus, we have established

Theorem 2.9.1. In the case of a purely axial material a plane wave spinor field is a solution of rotational elasticity if and only if it is a solution of one of the two massless Dirac equations (2.9.1).

It turns out that, in fact, a much stronger result holds. Consider a spinor field of the form

$$\xi(x^0, x^1, x^2, x^3) = e^{-ip_0 x^0} \eta(x^1, x^2, x^3).$$
(2.9.3)

We will call spinor fields of the form (2.9.3) *stationary*. Here, stationary refers to "time-harmonic" as opposed to "time-independent" as is shown by (2.9.3): while this is slightly non-standard language, it preserves accord between this thesis and [23]. In considering stationary spinor fields what we are doing is separating out time only as opposed to separating out all the variables.

The following result generalises Theorem 2.9.1.

Theorem 2.9.2. In the case of a purely axial material a nonvanishing stationary spinor field is a solution of rotational elasticity if and only if it is a solution of one of the two massless Dirac equations (2.9.1).

Theorem 2.9.2 was proved in [23] and the proof is quite delicate. It involves an argument which reduces a nonlinear second order partial differential equation of

a particular type to a pair of linear first order partial differential equations, which is, effectively, a form of integrability. An abstract self-contained version of this argument is given in Appendix B of [16].

It is interesting that Brocato and Capriz wrote in Section 6 of [13] "we are not aiming at a model for ether". Theorem 2.9.2 shows that one can, in fact, model world aether (ether) based on their idea of gyrocontinuum.

2.10 The linearised problem

Let R be the matrix function (2.E.1). Suppose that $R \approx I$, i.e. the coframe is almost aligned with the coordinate axes. Then

$$R = e^{*w} = I + *w + O(w^2)$$
(2.10.1)

where w is a vector function. In Cosserat elasticity literature the vector function w is called the *vector of microrotations*. It should not be confused with the *vector of macrorotations* which is one half of the curl of the vector function of displacements.

Formula (2.10.1) can be equivalently rewritten as

$$e^{j} \mapsto e^{j} + w \times e^{j} + O(w^{2}), \qquad j = 1, 2, 3,$$
(2.10.2)

where $e^j = e^j{}_{\alpha}$ is the *j*th element of the coframe and × denotes the cross (vector) product.

Substituting formula (2.10.2) into formulae (2.4.24), (2.4.25), (2.4.11) and (2.3.3) and linearising in w, we get

$$f = -2\operatorname{div} w, \tag{2.10.3}$$

$$v = \operatorname{curl} w, \tag{2.10.4}$$

$$\overset{*}{T}_{\alpha\beta} = \partial_{\alpha}w_{\beta} - \delta_{\alpha\beta}\operatorname{div} w, \qquad (2.10.5)$$

$$\omega = \dot{w} \tag{2.10.6}$$

where the dot indicates the time derivative. Finally, substituting formulae (2.10.3)–(2.10.6) into the Lagrangian density (2.5.2) of our rotational theory of elasticity, we get

$$L_{\rm Re}(w,\rho) = \left(\frac{4c^{\rm ax}}{3}(\operatorname{div} w)^2 + \frac{c^{\rm vec}}{2}\|\operatorname{curl} w\|^2 + c^{\rm ten} \left\|\frac{\partial_\alpha w_\beta + \partial_\beta w_\alpha}{2} - \frac{\delta_{\alpha\beta}}{3}\operatorname{div} w\right\|_V^2 - c^{\rm kin}\|\dot{w}\|^2\right)\rho$$
(2.10.7)

where the norm $\|\cdot\|_V$ of a rank two tensor is defined in accordance with (2.4.13). Formula (2.10.7) is similar to the formula for the Lagrangian density of classical linear elasticity, the only difference being the appearance of the term with curl w.

Up till now in this chapter we viewed the density ρ as a *dynamical variable*. The aim of this section is to formulate a linear analogue of the nonlinear problem considered in Sections 2.2–2.9 and in order to achieve this we have to make our density ρ prescribed. Moreover, as we assume our material to be homogeneous, the prescribed value of ρ will be constant. This leaves us with only one dynamical variable, the vector function w.

The linear field equation corresponding to the quadratic Lagrangian density (2.10.7) is

$$\ddot{w} = v_1^2 \operatorname{grad} \operatorname{div} w - v_2^2 \operatorname{curl} \operatorname{curl} w \qquad (2.10.8)$$

where v_1 and v_2 are defined by (2.8.22). Equation (2.10.8) is the equation of classical linear elasticity, the only difference being that we do not have the a priori inequality (2.8.24).

The equations of linearised Cosserat elasticity were previously examined by a number of authors. Publications [56, 55, 54] give a good overview of the subject. The difference between [56, 55, 54] and what is done in the current section is that in [56, 55, 54] the authors consider a model with both displacements and

microrotations whereas in our model the vector of microrotations is the only dynamical variable. Our Lagrangian density (2.10.7) is a special case of the Lagrangian density from [56, 55, 54] and our equation (2.10.8) is a special case of the equations from [56, 55, 54].

Equation (2.10.8) gives two types of waves, longitudinal waves propagating with velocity v_1 and transversal waves propagating with velocity v_2 . Of course, care is required in interpreting the terms "longitudinal" and "transversal" as w has the geometric meaning of the vector function of microrotations rather than the vector function of displacements (as it would be in classical linear elasticity).

Comparing the results of this section with Theorem 2.8.1 we see that the wave velocities are the same. This indicates that the mathematical model presented in the main part of Chapter 2 (Sections 2.2–2.5) is a natural geometrically nonlinear extension of linear rotational elasticity. Type 1 waves from Theorem 2.8.1 are a nonlinear version of longitudinal waves from equation (2.10.8) whereas type 2 waves are a nonlinear version of transversal ones.

2.A Notation

In Chapter 2 our notation follows [15, 23, 16, 81]. The only difference with [15, 81] is that in the latter the spacetime metric has opposite signature. In [23, 16] the signature is the same as in the current chapter, i.e. the 3-dimensional spatial metric has signature +++.

We use Greek letters for tensor (holonomic) indices and Latin letters for frame (anholonomic) indices. We identify differential forms with covariant antisymmetric tensors.

We define the action of the Hodge star on a rank r antisymmetric tensor X as

$$(*X)_{\alpha_{r+1}\dots\alpha_3} := (r!)^{-1} X^{\alpha_1\dots\alpha_r} \varepsilon_{\alpha_1\dots\alpha_3}$$
(2.A.1)

where ε is the totally antisymmetric quantity, $\varepsilon_{123} := +1$.

We use two-component complex-valued spinors (Weyl spinors) whose indices run through the values 1, 2 or $\dot{1}, \dot{2}$. Complex conjugation makes the undotted indices dotted and vice versa.

We define the "metric spinor"

$$\epsilon_{ab} = \epsilon_{\dot{a}\dot{b}} = \epsilon^{ab} = \epsilon^{\dot{a}\dot{b}} = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$$
(2.A.2)

and choose Pauli matrices

$$\sigma_{0\dot{a}b} = \begin{pmatrix} 1 & 0\\ 0 & 1 \end{pmatrix} = -\sigma^{0}{}_{\dot{a}b}, \qquad (2.A.3)$$

$$\sigma_{1\dot{a}b} = \begin{pmatrix} 0 & 1\\ 1 & 0 \end{pmatrix} = \sigma^{1}{}_{\dot{a}b}, \qquad \sigma_{2\dot{a}b} = \begin{pmatrix} 0 & -i\\ i & 0 \end{pmatrix} = \sigma^{2}{}_{\dot{a}b}, \qquad \sigma_{3\dot{a}b} = \begin{pmatrix} 1 & 0\\ 0 & -1 \end{pmatrix} = \sigma^{3}{}_{\dot{a}b}$$
(2.A.4)

Here the first index enumerates rows and the second enumerates columns.

2.B Spinor representation of torsion

We show in this appendix that the tensor $\overset{*}{T}_{\alpha\beta}$ defined by formula (2.4.10) (it is the Hodge dual, in the last pair indices, of the torsion tensor) is expressed via the spinor field ξ as

$$\overset{*}{T}_{\alpha\beta} = i \frac{\bar{\xi}^{\dot{a}} \sigma_{\beta \dot{a} b} \partial_{\alpha} \xi^{b} - \xi^{b} \sigma_{\beta \dot{a} b} \partial_{\alpha} \bar{\xi}^{\dot{a}} - (\bar{\xi}^{\dot{a}} \sigma^{\gamma}{}_{\dot{a} b} \partial_{\gamma} \xi^{b} - \xi^{b} \sigma^{\gamma}{}_{\dot{a} b} \partial_{\gamma} \bar{\xi}^{\dot{a}}) g_{\alpha\beta}}{\bar{\xi}^{\dot{c}} \sigma_{0 \dot{c} d} \xi^{d}} \,.$$
(2.B.1)

Note that formula (2.B.1) is invariant under the rescaling of our spinor field by an arbitrary positive scalar function.

Formula (2.B.1) is proved by direct substitution of formulae (2.6.4) and (2.6.5) into (2.4.10). In order to simplify calculations we observe that the expressions in

the left- and right-hand sides of formula (2.B.1) have an invariant nature, hence it is sufficient to prove formula (2.B.1) at a point at which the spinor field takes the value $\xi^a = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$. Then at this point we have

$$e^{j}{}_{\beta} = \delta^{j}{}_{\beta} \,, \tag{2.B.2}$$

$$\partial_{\alpha}(e^{1} + ie^{2}) = \begin{pmatrix} \partial_{\alpha}\xi^{1} - \partial_{\alpha}\bar{\xi}^{1} \\ i\partial_{\alpha}\xi^{1} - i\partial_{\alpha}\bar{\xi}^{1} \\ -2\partial_{\alpha}\xi^{2} \end{pmatrix}, \qquad \partial_{\alpha}e^{3} = \begin{pmatrix} \partial_{\alpha}\xi^{2} + \partial_{\alpha}\bar{\xi}^{2} \\ -i\partial_{\alpha}\xi^{2} + i\partial_{\alpha}\bar{\xi}^{2} \\ 0 \end{pmatrix}, \quad (2.B.3)$$

where $\alpha = 1, 2, 3$. Note that formulae (2.B.3) imply

$$\operatorname{curl}(e^{1} + ie^{2}) = \begin{pmatrix} -2\partial_{2}\xi^{2} - \partial_{3}(i\xi^{1} - i\bar{\xi}^{\dot{1}}) \\ 2\partial_{1}\xi^{2} + \partial_{3}(\xi^{1} - \bar{\xi}^{\dot{1}}) \\ \partial_{1}(i\xi^{1} - i\bar{\xi}^{\dot{1}}) - \partial_{2}(\xi^{1} - \bar{\xi}^{\dot{1}}) \end{pmatrix}, \qquad (2.B.4)$$
$$\begin{pmatrix} -\partial_{3}(-i\xi^{2} + i\bar{\xi}^{\dot{2}}) \\ -\partial_{3}(-i\xi^{2} + i\bar{\xi}^{\dot{2}}) \end{pmatrix}$$

$$\operatorname{curl} e^{3} = \begin{pmatrix} \partial_{3}(\bar{\xi}^{2} + i\bar{\xi}^{2}) \\ \partial_{3}(\bar{\xi}^{2} + \bar{\xi}^{2}) \\ \partial_{1}(-i\xi^{2} + i\bar{\xi}^{2}) - \partial_{2}(\xi^{2} + \bar{\xi}^{2}) \end{pmatrix}.$$
(2.B.5)

We now rewrite formula (2.4.10) in the form

$${}^{*}_{T} = \frac{1}{2}(e^{1} - ie^{2}) \otimes \operatorname{curl}(e^{1} + ie^{2}) + \frac{1}{2}(e^{1} + ie^{2}) \otimes \operatorname{curl}(e^{1} - ie^{2}) + e^{3} \otimes \operatorname{curl} e^{3}.$$
(2.B.6)

Substituting formulae (2.B.2), (2.B.4) and (2.B.5) into formula (2.B.6) we get

$${}^{*}_{T} = \begin{pmatrix} -\partial_{2}(\xi^{2} + \bar{\xi}^{\dot{2}}) - i\partial_{3}(\xi^{1} - \bar{\xi}^{\dot{1}}) & \partial_{1}(\xi^{2} + \bar{\xi}^{\dot{2}}) & i\partial_{1}(\xi^{1} - \bar{\xi}^{\dot{1}}) \\ i\partial_{2}(\xi^{2} - \bar{\xi}^{\dot{2}}) & -i\partial_{1}(\xi^{2} - \bar{\xi}^{\dot{2}}) - i\partial_{3}(\xi^{1} - \bar{\xi}^{\dot{1}}) & i\partial_{2}(\xi^{1} - \bar{\xi}^{\dot{1}}) \\ i\partial_{3}(\xi^{2} - \bar{\xi}^{\dot{2}}) & \partial_{3}(\xi^{2} + \bar{\xi}^{\dot{2}}) & -i\partial_{1}(\xi^{2} - \bar{\xi}^{\dot{2}}) - \partial_{2}(\xi^{2} + \bar{\xi}^{\dot{2}}) \end{pmatrix}$$

which coincides with the RHS of formula (2.B.1). This completes the proof.

We give for reference a more explicit version of formula (2.B.1):

$$\begin{pmatrix} {}^{*}_{11} \\ {}^{*}_{112} \\ {}^{*}_{113} \\ {}^{*}_{121} \\ {}^{*}_{121} \\ {}^{*}_{121} \\ {}^{*}_{122} \\ {}^{*}_{122} \\ {}^{*}_{123} \\ {}^{*}_{123} \\ {}^{*}_{123} \\ {}^{*}_{131} \\ {}^{*}_{132} \\ {}^{*}_{132} \\ {}^{*}_{133} \end{pmatrix} = \frac{1}{\bar{\xi}^{\bar{i}} \xi^{1} + \bar{\xi}^{\bar{2}} \xi^{2}} \begin{pmatrix} \bar{\xi}^{\bar{i}} \partial_{2} \xi^{1} - \bar{\xi}^{\bar{i}} \partial_{2} \xi^{2} + i \bar{\xi}^{\bar{i}} \partial_{3} \xi^{1} \\ i \bar{\xi}^{\bar{i}} \partial_{1} \xi^{2} - i \bar{\xi}^{\bar{2}} \partial_{1} \xi^{1} \\ i \bar{\xi}^{\bar{i}} \partial_{2} \xi^{2} \\ -i \bar{\xi}^{\bar{i}} \partial_{1} \xi^{2} - i \bar{\xi}^{\bar{2}} \partial_{1} \xi^{1} - i \bar{\xi}^{\bar{i}} \partial_{3} \xi^{1} \\ i \bar{\xi}^{\bar{i}} \partial_{2} \xi^{1} - i \bar{\xi}^{\bar{i}} \partial_{3} \xi^{1} + i \bar{\xi}^{\bar{2}} \partial_{3} \xi^{2} \\ i \bar{\xi}^{\bar{i}} \partial_{2} \xi^{1} - i \bar{\xi}^{\bar{2}} \partial_{2} \xi^{2} \\ i \bar{\xi}^{\bar{i}} \partial_{3} \xi^{2} + i \bar{\xi}^{\bar{2}} \partial_{3} \xi^{1} \\ \bar{\xi}^{\bar{i}} \partial_{3} \xi^{2} - \bar{\xi}^{\bar{2}} \partial_{3} \xi^{1} \\ -i \bar{\xi}^{\bar{i}} \partial_{1} \xi^{2} - i \bar{\xi}^{\bar{2}} \partial_{1} \xi^{1} + \bar{\xi}^{\bar{2}} \partial_{2} \xi^{1} - \bar{\xi}^{\bar{i}} \partial_{2} \xi^{2} \end{pmatrix} + \text{c.c.} \quad (2.B.7)$$

2.C Separation of variables

In this appendix we prove Lemmata 2.8.1 and 2.8.2. Note that it would suffice to prove Lemma 2.8.2 only, because Lemma 2.8.1 follows from Lemma 2.8.2. However, we prove Lemma 2.8.1 first for the sake of clarity of exposition.

Throughout this appendix we denote time by x^0 and the time derivative by ∂_0 .

Let us arrange the (pseudo)scalar f, the three components of the vector v_{α} , the nine components of the tensor $\overset{*}{T}_{\alpha\beta}$ and the three components of the (pseudo)vector ω_{α} into one 16-component "vector" V_J , $J = 1, \ldots, 16$. Then our Lagrangian density (2.5.2) can be written as

$$L_{\rm Re}(\xi) = \rho \sum_{J=1}^{16} A_J V_J^2$$
 (2.C.1)

where the A_J are some real constants. Put $W_J := \sqrt{\rho} V_J$. Then formula (2.C.1) takes the form

$$L_{\rm Re}(\xi) = \sum_{J=1}^{10} A_J W_J^2.$$
 (2.C.2)

According to formulae (2.6.11), (2.6.12), (2.B.1), (2.6.9) and (2.6.3) the components of the "vector" W are expressed via the spinor field ξ as

$$W_J = i \frac{\bar{\xi}^{\dot{a}} B^{\alpha}_{J \dot{a}b} \partial_{\alpha} \xi^b - \xi^b B^{\alpha}_{J \dot{a}b} \partial_{\alpha} \bar{\xi}^{\dot{a}}}{(\bar{\xi}^{\dot{c}} \sigma_{0\dot{c}d} \xi^d)^{1/2}}$$
(2.C.3)

where $B_{J\dot{a}b}^{\alpha}$ are some constants and summation is carried out over the spinor indices $\dot{a} = \dot{1}, \dot{2}, b = 1, 2$, and over $\alpha = 0, 1, 2, 3$. Here we use bold type to indicate relativistic notation, when time x^0 is viewed as one of the coordinates in (1+3)-dimensional spacetime.

Note that for given J and $\boldsymbol{\alpha}$ the 2 × 2 matrices $B_{J\,\dot{a}b}^{\boldsymbol{\alpha}}$ are Hermitian. This is because each of these matrices is a linear combination with real coefficients of the Pauli matrices $\sigma^{\beta}{}_{\dot{a}b}, \beta = 1, 2, 3$.

Our action is $S_{\text{Re}}(\xi) = \int L_{\text{Re}}(\xi) dx^0 dx^1 dx^2 dx^3$. Substituting (2.C.3) into (2.C.2) and varying the spinor field ξ we get

$$\delta S_{\rm Re}(\xi) = 2 \int \sum_{J=1}^{16} A_J \left(i \frac{(\delta \bar{\xi}^{\dot{a}}) B_{J \dot{a}b}^{\alpha} \partial_{\alpha} \xi^b - \xi^b B_{J \dot{a}b}^{\alpha} \partial_{\alpha} \delta \bar{\xi}^{\dot{a}}}{(\bar{\xi}^{\dot{c}} \sigma_{0\dot{c}d} \xi^d)^{1/2}} - \frac{(\delta \bar{\xi}^{\dot{a}}) \sigma_{0\dot{a}b} \xi^b}{2 \bar{\xi}^{\dot{c}} \sigma_{0\dot{c}d} \xi^d} W_J \right) W_J + \text{c.c.}$$

where we wrote down explicitly the terms with $\delta \bar{\xi}$ and incorporated the terms with $\delta \xi$ into the "c.c." (complex conjugate term). Here, when writing the integral, we dropped, for the sake of brevity, $dx^0 dx^1 dx^2 dx^3$. Integrating the term with $\partial_{\alpha} \delta \bar{\xi}^{\dot{a}}$ by parts and taking out the common factor $\delta \bar{\xi}^{\dot{a}}$ we rewrite the above formula as

$$\delta S_{\rm Re}(\xi) = 2 \int (\delta \bar{\xi}^{\dot{a}}) \sum_{J=1}^{16} A_J \left[\frac{i W_J B_J^{\alpha}{}_{\dot{a}\dot{b}} \partial_{\alpha} \xi^b}{(\bar{\xi}^{\dot{c}} \sigma_{0\dot{c}\dot{d}} \xi^d)^{1/2}} - \frac{W_J^2 \sigma_{0\dot{a}\dot{b}} \xi^b}{2 \bar{\xi}^{\dot{c}} \sigma_{0\dot{c}\dot{d}} \xi^d} + i \partial_{\alpha} \left(\frac{W_J B_J^{\alpha}{}_{\dot{a}\dot{b}} \xi^b}{(\bar{\xi}^{\dot{c}} \sigma_{0\dot{c}\dot{d}} \xi^d)^{1/2}} \right) \right] + \text{c.c.}$$

Comparing this formula with formula (2.7.1) we conclude that the spinor field F appearing in the latter is given by formula

$$F_{\dot{a}} = 2\sum_{J=1}^{16} A_J \left[\frac{iW_J B_J^{\alpha}{}_{\dot{a}\dot{b}}\partial_{\alpha}\xi^b}{(\bar{\xi}^{\dot{c}}\sigma_{0\dot{c}d}\xi^d)^{1/2}} - \frac{W_J^2\sigma_{0\dot{a}\dot{b}}\xi^b}{2\bar{\xi}^{\dot{c}}\sigma_{0\dot{c}d}\xi^d} + i\partial_{\alpha} \left(\frac{W_J B_J^{\alpha}{}_{\dot{a}\dot{b}}\xi^b}{(\bar{\xi}^{\dot{c}}\sigma_{0\dot{c}d}\xi^d)^{1/2}} \right) \right]$$
(2.C.4)

where the W_J are, in turn, given by formula (2.C.3).

If we now substitute the plane wave (2.8.1) into formula (2.C.3) we get

$$W_J = \frac{2\bar{\zeta}^{\dot{a}} B_{J\,\dot{a}b}^{\alpha} \mathbf{p}_{\alpha} \zeta^b}{(\bar{\zeta}^{\dot{c}} \sigma_{0\dot{c}d} \zeta^d)^{1/2}} \,. \tag{2.C.5}$$

Note that the above W_J are constant (do not depend on \mathbf{x}), which simplifies the next step: substituting (2.8.1) into (2.C.4) and dividing through by the common factor $e^{-i\mathbf{p}\cdot\mathbf{x}}$ we get

$$e^{i\mathbf{p}\cdot\mathbf{x}}F_{\dot{a}} = 2\sum_{J=1}^{16} A_J \left[\frac{2W_J B_J^{\alpha}{}_{\dot{a}\dot{b}}\mathbf{p}_{\alpha}\zeta^b}{(\bar{\zeta}^{\dot{c}}\sigma_{0\dot{c}\dot{d}}\zeta^d)^{1/2}} - \frac{W_J^2\sigma_{0\dot{a}\dot{b}}\zeta^b}{2\bar{\zeta}^{\dot{c}}\sigma_{0\dot{c}\dot{d}}\zeta^d}\right].$$
 (2.C.6)

The remarkable feature of formula (2.C.6) is that its RHS is constant, i.e. it does not depend on **x**. This completes the proof of Lemma 2.8.1.

Let us now substitute the plane wave (2.8.1) directly into our Lagrangian density (2.C.2). Our Lagrangian density takes the form

$$L_{\rm Re}(\zeta; \mathbf{p}) = \sum_{J=1}^{16} A_J W_J^2$$
 (2.C.7)

where the W_J are given by formula (2.C.5). The Lagrangian density (2.C.7) does not depend on **x**. The dynamical variable in this Lagrangian density is the constant 2-component complex spinor ζ , whereas the relativistic 4-momentum **p** plays the role of a parameter. Varying the spinor ζ we get

$$\delta L_{\rm Re}(\zeta;\mathbf{p}) = 2\sum_{J=1}^{16} A_J \left(\frac{2(\delta\bar{\zeta}^{\dot{a}})B_J^{\alpha}{}_{\dot{a}b}\mathbf{p}_{\alpha}\zeta^b}{(\bar{\zeta}^{\dot{c}}\sigma_{0\dot{c}d}\zeta^d)^{1/2}} - \frac{(\delta\bar{\zeta}^{\dot{a}})\sigma_{0\dot{a}b}\zeta^b}{2\bar{\zeta}^{\dot{c}}\sigma_{0\dot{c}d}\zeta^d} W_J \right) W_J + \text{c.c.}$$

Comparing this formula with formula (2.8.16) we conclude that the constant spinor G appearing in the latter is given by formula

$$G_{\dot{a}} = 2\sum_{J=1}^{16} A_J \left[\frac{2W_J B_J^{\alpha}{}_{\dot{a}\dot{b}} \mathbf{p}_{\alpha} \zeta^b}{(\bar{\zeta}^{\dot{c}} \sigma_{0\dot{c}d} \zeta^d)^{1/2}} - \frac{W_J^2 \sigma_{0\dot{a}\dot{b}} \zeta^b}{2\bar{\zeta}^{\dot{c}} \sigma_{0\dot{c}d} \zeta^d} \right].$$
(2.C.8)

It remains to observe that the right-hand sides of formulae (2.C.6) and (2.C.8) are the same. This completes the proof of Lemma 2.8.2.

2.D Conservation of mass

In this appendix we address the issue of conservation of total mass of our continuum. We feel the need to clarify this matter because in the main part of this chapter (Sections 2.2–2.9) we took the density ρ to be a dynamical variable, making it an arbitrary function of time t and spatial coordinates x^{α} , $\alpha = 1, 2, 3$.

A variable density ρ generates a material flow with velocity V. These are related by the continuity equation

$$\partial_t \rho + \operatorname{div}(\rho V) = 0.$$
 (2.D.1)

If we take the flow to be potential, $V = \operatorname{grad} \phi$, then equation (2.D.1) becomes

$$\partial_t \rho + \operatorname{div}(\rho \operatorname{grad} \phi) = 0.$$
 (2.D.2)

Equation (2.D.2) is Poisson's equation for the unknown velocity potential ϕ .

As equation (2.D.2) is written in the whole of \mathbb{R}^3 , it has to be supplemented by an appropriate condition at infinity. Assume, for definiteness, that $\partial_t \rho \to 0$ and $\rho \to \text{const} > 0$ as $||x|| \to \infty$, and that convergence is sufficiently fast. Then the natural condition at infinity is

$$\phi \to 0 \quad \text{as} \quad ||x|| \to \infty.$$
 (2.D.3)

The problem (2.D.2), (2.D.3) has a unique solution ϕ which admits the asymptotic expansion

$$\phi = c \|x\|^{-1} + O(\|x\|^{-2}) \text{ as } \|x\| \to \infty.$$
 (2.D.4)

This means that the total flow at infinity is $-4\pi c$. More specifically, if the constant c in equation (2.D.4) is positive then we have material flowing from infinity, and if it is negative then we have material flowing to infinity. In other words, we implicitly assume in our model that infinity can act as a source or a sink.

It remains to examine what happens when we seek plane wave solutions (2.8.1) or, more generally, stationary solutions (2.9.3). Substituting (2.8.1) or (2.9.3) into (2.6.3) we see that in this case the density ρ is static, hence the velocity potential ϕ is identically zero and there is no flow to or from infinity.

2.E Matrix representation of the Lagrangian of Rotational elasticity

In Chapter 2 we predominantly use tensor notation which comes from alternative theories of gravity [39]. This requires care in distinguishing upper indices from lower indices and tensor (holonomic) indices from frame (anholonomic) indices. In this appendix we establish correspondence between our notation and the more user-friendly matrix notation of [46, 57, 58, 59, 60, 61, 62, 63] the advantage of which is that it does not require such a high degree of sophistication: all indices become lower indices, there is no distinction between tensor and frame indices, and only the order of indices is important.

Define the 3×3 special orthogonal matrix function R which is expressed via our coframe e as

$$R_{j\alpha} := e^{j}{}_{\alpha} \tag{2.E.1}$$

with the index j enumerating the rows of the matrix R and the index α enumerating the columns. This matrix function R is the matrix function \overline{R} from [59], we simply dropped the bar because in our setting we do not have to distinguish macro- and microrotations (we have only microrotations). Formulae (2.3.2) and (2.E.1) imply that the vector of angular velocity ω can be rewritten in terms of the matrix function R as

$$\omega = *(R^T \dot{R}) \tag{2.E.2}$$

where the dot indicates the time derivative and the application of the Hodge star is justified because the matrix $R^T \dot{R}$ is antisymmetric (we can identify it with a rank two antisymmetric tensor).

Define the 3×3 matrix function Curl R in accordance with formula (2.1) from [63]:

$$(\operatorname{Curl} R)_{j\gamma} := (\operatorname{curl} e^j)^{\gamma} = (*de^j)^{\gamma} = \varepsilon^{\alpha\beta\gamma} \partial_{\alpha} e^j{}_{\beta} \qquad (2.\mathrm{E.3})$$

with the index j enumerating the rows of the matrix $\operatorname{Curl} R$ and the index γ enumerating the columns.

Let us now examine our rank two tensor $\stackrel{*}{T}$ defined by formula (2.4.10). We can view the tensor $\stackrel{*}{T}$ as a matrix function with the first tensor index enumerating the rows and the second index enumerating the columns. Formulae (2.4.10) and (2.E.3) imply the matrix identity

$$\overset{*}{T} = R^T \operatorname{Curl} R. \tag{2.E.4}$$

Formulae (2.E.2), (2.E.4), (2.4.19) and (2.4.20) imply that the Lagrangian density of rotational elasticity (2.5.2) can be rewritten as

$$L_{\rm Re}(R,\rho) = \left(\frac{c^{\rm ax}}{3} [\operatorname{tr}(R^T \operatorname{Curl} R)]^2 + c^{\rm vec} \|\operatorname{skew}(R^T \operatorname{Curl} R)\|_V^2 + c^{\rm ten} \|\operatorname{dev}\operatorname{sym}(R^T \operatorname{Curl} R)\|_V^2 - \frac{c^{\rm kin}}{2} \|R^T \dot{R}\|_V^2 \right) \rho \quad (2.E.5)$$

where dev $X = X - \frac{1}{3}(\operatorname{tr} X)I$, sym $X = \frac{1}{2}(X + X^T)$, skew $X = \frac{1}{2}(X - X^T)$ and $\|\cdot\|_V$ is the Frobenius norm (2.4.13).

Chapter 3

Microlocal analysis of first order systems I

The purpose of this chapter is to study an elliptic self-adjoint $m \times m$ first order pseudodifferential operator A acting on m-columns of half-densities over a connected compact n-dimensional manifold M without boundary, where $m, n \geq 2$. We construct the propagator which provides a solution to the Cauchy problem for the dynamic equation, evolving the solution to the Cauchy problem in time. This object is related to the spectral function and counting function of the operator A. We go on to present the two-term asymptotic expansions of the spectral and counting functions, correcting errors in the literature.

The structure of this chapter is as follows. We give the absolute minimum microlocal analysis theory, providing definitions and references. We state the key results of [22] and give an overview of the iterative process developed to construct the propagator. The appearance of a U(1) connection in our construction is discussed, and we consider briefly the issue of spectral asymmetry (which, in part, led us towards the work presented in Chapter 6). Finally, we provide a short review discussing the literature and motivating this study.

3.1 Key ideas in microlocal analysis

In this chapter we work, as stated, on an *n*-dimensional connected compact manifold M without boundary. By T^*M we denote the cotangent bundle of the manifold M, and by

$$T'M = T^*M \setminus \{\xi = 0\}$$
 (3.1.1)

the cotangent bundle with the zero section removed. Local coordinates on our manifold M are denoted by $x = (x^1, \ldots, x^n)$ and their dual coordinates (referred to as momentum in the physics literature) $\xi = (\xi_1, \ldots, \xi_n)$ on the fibre T_x^*M . Then (x, ξ) are local coordinates on the cotangent bundle.

A highly detailed treatment of the differential geometry required in this thesis (including (co)tangent bundles) can be found in [53], Chapter 9.

3.1.1 Hamiltonian trajectories

A Hamiltonian $h^{(j)}(x,\xi)$ is a nonvanishing smooth function on T'M, positively homogeneous of degree 1 in ξ , that is, for $\lambda \in \mathbb{R}$ s.t. $\lambda > 0$:

$$h^{(j)}(x,\lambda\xi) = \lambda h^{(j)}(x,\xi).$$
 (3.1.2)

Although seemingly redundant at this stage, the use of the additional superscript (j) will become clear in due course: the Hamiltonians $h^{(j)}$, j = 1, ..., m, will be the eigenvalues of the principal symbol of our pseudodifferential operator A.

Given a Hamiltonian $h^{(j)}(x,\xi)$ and a point $(y,\eta) \in T'M$ we denote by

$$(x^{(j)}(t;y,\eta),\xi^{(j)}(t;y,\eta)), \quad t \in \mathbb{R},$$
(3.1.3)

the integral curve of the Hamiltonian vector field generated by $h^{(j)}(x,\xi)$, i.e. the parametric curve in T'M providing the solution of the Hamiltonian system

$$\dot{x}^{(j)}(t;y,\eta) = h_{\xi}^{(j)}(x^{(j)}(t;y,\eta),\xi^{(j)}(t;y,\eta)), \qquad (3.1.4)$$

$$\dot{\xi}^{(j)}(t;y,\eta) = -h_x^{(j)}(x^{(j)}(t;y,\eta),\xi^{(j)}(t;y,\eta)), \qquad (3.1.5)$$

where the dot denotes differentiation in the parameter $t \in \mathbb{R}$, with initial condition

$$\left(x^{(j)}(0;y,\eta),\xi^{(j)}(0;y,\eta)\right) = (y,\eta).$$
(3.1.6)

The curve $(x^{(j)}(t; y, \eta), \xi^{(j)}(t; y, \eta))$ is referred to as the Hamiltonian trajectory generated by the Hamiltonian $h^{(j)}$ starting at the point (y, η) . The Hamiltonian is preserved along Hamiltonian trajectories.

The abstract situation described above becomes relevant to our study when the Hamiltonians are taken to be the eigenvalues of the principal symbol (see Section 3.2). Roughly speaking, the Hamiltonian trajectories trace out the path of waves governed by the hyperbolic equation (3.2.3) and, therefore, are a basic building block in the construction of the propagator.

3.1.2 Oscillatory integrals

Basic to our construction is the notion of an oscillatory integral. Using this object we will be able to define pseudodifferential operators. Oscillatory integrals are also important for the construction of the propagator: in this case we must extend the most basic definition of an oscillatory integral given in this section to a more general time-dependent object.

Let Ω be a sufficiently small open set in M. (By "sufficiently small" we mean that by introducing local coordinates x we can identify Ω with an open set in \mathbb{R}^{n} .) We denote by S^l the class of complex-valued infinitely smooth $m \times m$ matrixfunctions $a(x, y, \xi)$ which admit the asymptotic expansion

$$a(x, y, \xi) \sim \sum_{k=0}^{\infty} a_{l-k}(x, y, \xi), \quad |\xi| \to \infty$$
 (3.1.7)

with each $a_{l-k}(x, y, \xi)$ positively homogeneous in ξ of degree l - k.

We assume that for $a \in S^l$

$$\operatorname{supp} a \subset K \times K \times (\mathbb{R}^n \setminus \{\xi = 0\}) \tag{3.1.8}$$

where $K \subset \Omega$ is some compact set. This can be achieved by introducing a suitable cutoff function, see Example 3.10 and Definition 3.11 from [71].

Note that $S^p \subset S^q$ for q > p, and define $S^{-\infty} := \bigcap_l S^l$. Then $S^{-\infty}$ consists of all infinitely smooth matrix-functions decreasing with all derivatives faster than any power of $|\xi|$ as $|\xi| \to \infty$.

The formal integral

$$\mathcal{I}_a(x,y) = \int e^{i(x-y)^{\alpha}\xi_{\alpha}} a(x,y,\xi) d\xi \qquad (3.1.9)$$

with $a \in S^l$ is called an oscillatory integral of order l with (matrix) amplitude a. While this integral may not converge in the usual sense, it can be treated as a distribution in the variable x or the variable y. This means that one can operate with (3.1.9) as though it were a convergent integral, that is, one can formally integrate by parts, differentiate under the integral sign, etc.

Note that it is possible to replace the function $(x - y)^{\alpha} \xi_{\alpha}$ appearing in (3.1.9) with a more general phase function which must satisfy certain nondegeneracy conditions. See [72], Section 2.1, for further details.

3.1.3 Pseudodifferential operators

Consider the oscillatory integral (3.1.9) to be the distribution kernel of an operator A, which is then expressed explicitly as

$$Au(x) = \int e^{i(x-y)^{\alpha}\xi_{\alpha}} a(x,y,\xi) u(y) dy d\xi \qquad (3.1.10)$$

where x, y are coordinates on $\Omega \subset M$ and $u(x) = (u_1(x) \dots u_m(x))^T$ is a column of complex-valued half-densities, cf. Subsection 1.3. An operator A of the form (3.1.10) is called an $m \times m$ matrix pseudodifferential operator of order l on Ω acting in the space of half-densities. Throughout Chapter 3 we will always consider operators acting on m-columns of half-densities.

We can extend this definition to the whole manifold M as follows. We say the operator A is an $m \times m$ matrix pseudodifferential operator of order l on M if its distribution kernel (3.1.9) is infinitely smooth outside the diagonal

$$\{(x,y) \in M \times M : x = y\}, \qquad (3.1.11)$$

and if, for any point on the diagonal, there exists a neighbourhood $\Sigma \subset M \times M$ such that the distribution kernel of A can be written in the form (3.1.9) with $a \in S^l$ and $\Sigma \subset \Omega \times \Omega$.

We can remove the dependence on the variable y (or x) in the amplitude of the pseudodifferential operator (3.1.10). Indeed, if A is an $m \times m$ pseudodifferential operator of order l with amplitude $a \in S^l$, then A differs by an integral operator with a C^{∞} kernel from the pseudodifferential operator with amplitude $A(x,\xi)$, $(x,\xi) \in T'M$, given by

$$A(x,\xi) \sim \sum_{\alpha} \frac{i^{-|\alpha|}}{\alpha!} \left(\partial_{\xi}^{\alpha} \partial_{y}^{\alpha} a(x,y,\xi) \right) \Big|_{y=x}.$$
(3.1.12)

This is achieved by expanding the amplitude a at the point y = x using Taylor's formula and then integrating by parts in the variable ξ . Equivalently, we can
express (3.1.10) as follows:

$$Au(x) = \int e^{i(x-y)^{\alpha}\xi_{\alpha}} A(x,\xi)u(y) \, dyd\xi \pmod{C^{\infty}}$$
(3.1.13)

where $A(x,\xi)$ is given by (3.1.12). The reduced amplitude $A(x,\xi)$ is called the (full) symbol of the operator A. The symbol $A(x,\xi)$ determines the operator A modulo an integral operator with infinitely smooth kernel. Conversely, the operator A determines the symbol $A(x,\xi)$ modulo an amplitude from $S^{-\infty}$.

As $A(x,\xi) \in S^l$ it can be expanded asymptotically as the sum

$$A(x,\xi) \sim \sum_{k=0}^{\infty} A_{l-k}(x,\xi), \quad |\xi| \to +\infty$$
(3.1.14)

where each term $A_{l-k}(x,\xi)$ is positively homogeneous in ξ of degree l-k. The leading term $A_l(x,\xi)$ in the expansion (3.1.14) is called the *principal symbol* of the pseudodifferential operator A.

The principal symbol $A_l(x,\xi)$ is invariant under changes of local coordinates. However, the term $A_{l-1}(x,\xi)$ is not invariant. To combat this issue, we introduce the *subprincipal symbol* of the operator A, defined as

$$A_{\rm sub}(x,\xi) = A_{l-1}(x,\xi) + \frac{i}{2} (A_l)_{x^{\alpha}\xi_{\alpha}}(x,\xi).$$
(3.1.15)

The subprincipal symbol (3.1.15) is positively homogeneous of degree l - 1 in ξ and invariant under changes of local coordinates. The (full) symbol behaves in a more complicated manner under changes of local coordinates.

As stated in the introduction of this chapter, we work with elliptic pseudodifferential operators. An $m \times m$ pseudodifferential operator A of order l is said to be elliptic if det $A_l(x,\xi) \neq 0, \forall (x,\xi) \in T'M$.

Note that, as stated in Subsection 3.1.1, the nonzero Hamiltonians $h^{(j)}$, $j = 1, \ldots, m$, will be the eigenvalues of the principal symbol of our intended object of

study, a first order elliptic pseudodifferential operator. The fact that our Hamiltonians are nonvanishing is encoded in the ellipticity condition, i.e. the nonvanishing of the determinant of the principal symbol, see Section 3.2 for further details.

3.1.4 Time-dependent oscillatory integrals

The final component of established theory we need to proceed to the key results from [22] pertains to so-called time-dependent oscillatory integrals. At a formal level, one can view the extension from the local case, see Subsection 3.1.2, to the time-dependent case as the introduction of time dependence into the definition of the oscillatory integral (specifically into the phase function and amplitude). At a more fundamental level, we relate this time dependence to an underlying Hamiltonian system as described in Subsection 3.1.1.

The purpose of this construction is reasonably clear. One of the key applications of the microlocal techniques presented here is to allow us to study the propagation of the singularities of solutions of certain hyperbolic systems. We construct the propagator for our system as a sum of global oscillatory integrals. The propagator is the object that shows how a solution evolves in time, and the singularities of our initial data under its action will propagate along Hamiltonian trajectories.

The limitations of the definition of oscillatory integrals in this regard are obvious: oscillatory integrals of the form (3.1.9) are defined only on (sufficiently small) open sets $\Omega \subset M$. The highly local nature of this definition means we are, in effect, unable to propagate information between, say, any two points of our compact connected manifold M.

The definition of time-dependent oscillatory integrals is a delicate issue and must be handled with care. Our construction of time-dependent oscillatory integrals follows [72]. The remarkable feature of the technique proposed in [72] is that the time-dependent oscillatory integrals developed are global, in the sense that they are global in time, $t \in \mathbb{R}$, and do not, in general, depend on the choice of local coordinates on M. The superlative reference for scalar operators is [72]: we follow this treatment, using notation adapted for systems. A matrix version of this construction was previously considered in [64].

From this point until the end of this section we associate a seemingly redundant index (j) with a number of different quantities. This will become essential in Section 3.2 when the index j will enumerate the eigenvalues of the principal symbol of our operator of study.

The (formal) integral

$$\mathcal{I}_{\varphi^{(j)},a^{(j)}}(t,x;y) = \int e^{i\varphi^{(j)}(t,x;y,\eta)} a^{(j)}(t,x;y,\eta) \varsigma^{(j)}(t,x;y,\eta) d_{\varphi^{(j)}}(t,x;y,\eta) d\eta,$$
(3.1.16)

is called a *time-dependent oscillatory integral of order l*, where the expressions appearing in (3.1.16) have the following meaning.

• The function $\varphi^{(j)}(t, x; y, \eta)$ is a *phase function*, i.e. a function $\mathbb{R} \times M \times T'M \to \mathbb{C}$ positively homogeneous in η of degree 1 and satisfying the conditions

$$\varphi^{(j)}(t,x;y,\eta) = (x - x^{(j)}(t;y,\eta))^{\alpha} \xi^{(j)}_{\alpha}(t;y,\eta) + O(|x - x^{(j)}(t;y,\eta)|^2),$$
(3.1.17)

Im
$$\varphi^{(j)}(t, x; y, \eta) \ge 0,$$
 (3.1.18)

$$\det \varphi_{x^{\alpha}\eta_{\beta}}^{(j)}(t, x^{(j)}(t; y, \eta); y, \eta) \neq 0.$$
(3.1.19)

Recall that according to Corollary 2.4.5 from [72] we are guaranteed to have (3.1.19) if we choose a phase function

$$\varphi^{(j)}(t,x;y,\eta) = (x - x^{(j)}(t;y,\eta))^{\alpha} \xi^{(j)}_{\alpha}(t;y,\eta) + \frac{1}{2} C^{(j)}_{\alpha\beta}(t;y,\eta) (x - x^{(j)}(t;y,\eta))^{\alpha} (x - x^{(j)}(t;y,\eta))^{\beta} + O(|x - x^{(j)}(t;y,\eta)|^3) \quad (3.1.20)$$

with complex-valued symmetric matrix-function $C_{\alpha\beta}^{(j)}$ satisfying the strict inequality Im $C^{(j)} > 0$ (our original requirement (3.1.18) implies only the non-strict inequality Im $C^{(j)} \ge 0$). Note that even though the matrixfunction $C_{\alpha\beta}^{(j)}$ is not a tensor, the inequalities Im $C^{(j)} \ge 0$ and Im $C^{(j)} > 0$ are invariant under transformations of local coordinates x; see Remark 2.4.9 in [72] for details.

• The quantity $a^{(j)}(t, x; y, \eta)$ is the amplitude of our oscillatory integral, i.e. a complex-valued $m \times m$ matrix-function $\mathbb{R} \times M \times T'M \to \mathbb{C}^{m^2}$ which admits the asymptotic expansion

$$a^{(j)}(t,x;y,\eta) \sim \sum_{k=0}^{\infty} a^{(j)}_{l-k}(t,x;y,\eta)$$
 (3.1.21)

as $|\eta| \to +\infty$, where each $a_{l-k}^{(j)}$ is positively homogeneous of degree l-k in η .

Remark 3.1.1. A great deal of this chapter deals with the construction of the propagator, a formal definition of which is given later in (3.2.1). We construct the propagator as a sum of time-dependent oscillatory integrals, i.e. integrals of the type we are defining in this section. An important property of these particular time-dependent oscillatory integrals is that their amplitudes are independent of the variable x, that is, they are actually time-dependent symbols $u^{(j)}(t; y, \eta)$ rather than amplitudes (3.1.21).

• The quantity $d_{\varphi^{(j)}}(t, x; y, \eta)$ is defined in accordance with formula (2.2.4) from [72] as

$$d_{\varphi^{(j)}}(t,x;y,\eta) := (\det^2 \varphi_{x^{\alpha}\eta_{\beta}}^{(j)})^{1/4} = |\det \varphi_{x^{\alpha}\eta_{\beta}}^{(j)}|^{1/2} e^{i \arg(\det^2 \varphi_{x^{\alpha}\eta_{\beta}}^{(j)})/4}.$$
 (3.1.22)

Note that in view of (3.1.19) our $d_{\varphi^{(j)}}$ is well-defined and smooth for x close to $x^{(j)}(t; y, \eta)$. It is known [72] that under coordinate transformations $d_{\varphi^{(j)}}$ behaves as a half-density in x and as a half-density to the power -1 in y. In formula (3.1.22) we wrote $(\det^2 \varphi_{x^{\alpha} \eta_{\beta}}^{(j)})^{1/4}$ rather than $(\det \varphi_{x^{\alpha} \eta_{\beta}}^{(j)})^{1/2}$ in order to make this expression truly invariant under coordinate transformations. Recall that local coordinates x and y are chosen independently and that η is a covector based at the point y. Consequently, $\det \varphi_{x^{\alpha} \eta_{\beta}}^{(j)}$ changes sign under inversion of one of the local coordinates x^{α} , $\alpha = 1, \ldots, n$, or y^{β} , $\beta = 1, \ldots, n$, whereas $\det^2 \varphi_{x^{\alpha} \eta_{\beta}}^{(j)}$ retains sign under inversion.

The choice of (smooth) branch of $\arg(\det^2 \varphi_{x^{\alpha}\eta_{\beta}}^{(j)})$ is assumed to be fixed. Thus, for a given phase function $\varphi^{(j)}$ formula (3.1.22) defines the quantity $d_{\varphi^{(j)}}$ uniquely up to a factor $e^{ik\pi/2}$, k = 0, 1, 2, 3. Observe now that if we set t = 0 and choose the same local coordinates for x and y, we get $\varphi_{x^{\alpha}\eta_{\beta}}^{(j)}(0, y; y, \eta) = I$. This implies that we can fully specify the choice of branch of $\arg(\det^2 \varphi_{x^{\alpha}\eta_{\beta}}^{(j)})$ by requiring that $d_{\varphi^{(j)}}(0, y; y, \eta) = 1$.

The introduction of the factor $d_{\varphi^{(j)}}$ in (3.1.16) is important in the construction of the propagator, discussed in Remark 3.1.1. The purpose of the factor is twofold.

- (a) It ensures that the symbol of the propagator, $u^{(j)}$, is a function on $\mathbb{R} \times T'M$ in the full differential geometric sense of the word, i.e. that it is invariant under transformations of local coordinates x and y.
- (b) It ensures that the principal symbol $u_0^{(j)}$ does not depend on the choice of phase function $\varphi^{(j)}$. See Remark 2.2.8 in [72] for more details.
- The quantity $\varsigma^{(j)}(t, x; y, \eta)$ is a smooth cut-off function $\mathbb{R} \times M \times T'M \to \mathbb{R}$ satisfying the following conditions.
 - (a) $\varsigma^{(j)}(t,x;y,\eta) = 0$ on the set $\{(t,x;y,\eta): |h^{(j)}(y,\eta)| \le 1/2\}.$
 - (b) $\varsigma^{(j)}(t,x;y,\eta) = 1$ on the intersection of a small conic neighbourhood of the set

$$\{(t, x; y, \eta) : x = x^{(j)}(t; y, \eta)\}$$
(3.1.23)

with the set $\{(t, x; y, \eta) : |h^{(j)}(y, \eta)| \ge 1\}.$

(c) $\varsigma^{(j)}(t, x; y, \lambda \eta) = \varsigma^{(j)}(t, x; y, \eta)$ for $|h^{(j)}(y, \eta)| \ge 1, \ \lambda \ge 1.$

• It is known (see Section 2.3 in [72] for details) that Hamiltonian trajectories generated by a Hamiltonian $h^{(j)}(x,\xi)$ positively homogeneous in ξ of degree 1 satisfy the identity

$$(x_{\eta}^{(j)})^{\alpha\beta}\xi_{\alpha}^{(j)} = 0, \qquad (3.1.24)$$

where $(x_{\eta}^{(j)})^{\alpha\beta} := \partial(x^{(j)})^{\alpha}/\partial\eta_{\beta}$. Formulae (3.1.17) and (3.1.24) imply

$$\varphi_{\eta}^{(j)}(t, x^{(j)}(t; y, \eta); y, \eta) = 0.$$
(3.1.25)

This allows us to apply the stationary phase method in the neighbourhood of the set (3.1.23) and disregard what happens away from it.

The definition of a time-dependent oscillatory integral (3.1.16) is the last piece of established theory needed to proceed with the statement of the key results from [22].

3.2 Main results

Consider a first order classical pseudodifferential operator A acting on columns $v = \begin{pmatrix} v_1 & \dots & v_m \end{pmatrix}^T$ of complex-valued half-densities over a connected compact n-dimensional manifold M without boundary. As stated above we assume that $m, n \geq 2$.

We assume the symbol of the operator A to be infinitely smooth. We also assume that the operator A is formally self-adjoint (symmetric):

$$\int_M w^*(Av) \, dx = \int_M (Aw)^* v \, dx$$

for all infinitely smooth $v, w : M \to \mathbb{C}^m$. Here and further on the superscript * in matrices, rows and columns indicates Hermitian conjugation in \mathbb{C}^m . Let $A_1(x,\xi)$ be the principal symbol of the operator A and $h^{(j)}(x,\xi)$ its (Hamiltonian) eigenvalues. We assume these eigenvalues to be nonzero (this is a version of the ellipticity condition) but do not make any assumptions on their sign. We also assume that the eigenvalues $h^{(j)}(x,\xi)$ are simple for all $(x,\xi) \in T'M$. The techniques developed in [22] do not work in the case when eigenvalues of the principal symbol have variable multiplicity, though they could probably be adapted to the case of constant multiplicity different from multiplicity 1.

We enumerate the eigenvalues of the principal symbol $h^{(j)}(x,\xi)$ in increasing order, using a positive index $j = 1, ..., m^+$ for positive $h^{(j)}(x,\xi)$ and a negative index $j = -1, ..., -m^-$ for negative $h^{(j)}(x,\xi)$. Here m^+ is the number of positive eigenvalues of the principal symbol and m^- is the number of negative ones. Of course, $m^+ + m^- = m$.

Under the above assumptions A is a self-adjoint operator, in the full functional analytic sense, in the Hilbert space $L^2(M; \mathbb{C}^m)$ (Hilbert space of square integrable complex-valued column "functions") with domain $H^1(M; \mathbb{C}^m)$ (Sobolev space of complex-valued column "functions" which are square integrable together with their first partial derivatives) and the spectrum of A is discrete.

Let λ_k and $v_k = (v_{k1}(x) \dots v_{km}(x))^T$ be the eigenvalues and normalised eigenfunctions of the operator A. The eigenvalues λ_k are enumerated in increasing order with account of multiplicity, using a positive index $k = 1, 2, \ldots$ for positive λ_k and a nonpositive index $k = 0, -1, -2, \ldots$ for nonpositive λ_k . If the operator A is bounded from below (i.e. if $m^- = 0$) then the index k runs from some integer value to $+\infty$; if the operator A is bounded from above (i.e. if $m^+ = 0$) then the index k runs from $-\infty$ to some integer value; and if the operator A is unbounded from above and from below (i.e. if $m^+ \neq 0$ and $m^- \neq 0$) then the index k runs from $-\infty$ to $+\infty$.

We will be studying the following three objects.

Object 1. Our first object of study is the *propagator*, which is the one-parameter family of operators defined as

$$U(t) := e^{-itA} = \sum_{k} e^{-it\lambda_k} v_k(x) \int_M [v_k(y)]^*(\cdot) \, dy \,, \qquad (3.2.1)$$

 $t \in \mathbb{R}$. The propagator provides a solution to the Cauchy problem

$$w|_{t=0} = v$$
 (3.2.2)

for the dynamic equation

$$D_t w + A w = 0, (3.2.3)$$

where $D_t := -i\partial/\partial t$. Namely, it is easy to see that if the column of half-densities v = v(x) is infinitely smooth, then, setting w := U(t) v, we get a time-dependent column of half-densities w(t, x) which is also infinitely smooth and which satisfies the equation (3.2.3) and the initial condition (3.2.2). The use of the letter "U" for the propagator is motivated by the fact that for each t the operator U(t) is unitary.

Object 2. Our second object of study is the *spectral function*, which is the real density defined as

$$e(\lambda, x, x) := \sum_{0 < \lambda_k < \lambda} \|v_k(x)\|^2,$$
 (3.2.4)

where $||v_k(x)||^2 := [v_k(x)]^* v_k(x)$ is the square of the Euclidean norm of the eigenfunction v_k evaluated at the point $x \in M$ and λ is a positive parameter (spectral parameter).

Object 3. Our third and final object of study is the counting function

$$N(\lambda) := \sum_{0 < \lambda_k < \lambda} 1 = \int_M e(\lambda, x, x) \, dx \,. \tag{3.2.5}$$

In other words, $N(\lambda)$ is the number of eigenvalues λ_k between zero and λ .

It is natural to ask the question: why, in defining the spectral function (3.2.4)

and the counting function (3.2.5), did we choose to perform summation over all *positive* eigenvalues up to a given positive λ rather than over all *negative* eigenvalues up to a given negative λ ? There is no particular reason. One case reduces to the other by the change of operator $A \mapsto -A$. This issue will be revisited in Section 3.5.

Further on we assume that $m^+ > 0$, i.e. that the operator A is unbounded from above.

Our objectives are as follows.

Objective 1. We aim to construct the propagator (3.2.1) explicitly in terms of oscillatory integrals, modulo an integral operator with an infinitely smooth, in the variables t, x and y, integral kernel.

Objectives 2 and 3. We aim to derive, under appropriate assumptions on Hamiltonian trajectories, two-term asymptotics for the spectral function (3.2.4) and the counting function (3.2.5), i.e. formulae of the type

$$e(\lambda, x, x) = a(x) \lambda^{n} + b(x) \lambda^{n-1} + o(\lambda^{n-1}), \qquad (3.2.6)$$

$$N(\lambda) = a\lambda^n + b\lambda^{n-1} + o(\lambda^{n-1})$$
(3.2.7)

as $\lambda \to +\infty$. Obviously, here we expect the real constants a, b and real densities a(x), b(x) to be related in accordance with

$$a = \int_M a(x) \, dx, \qquad (3.2.8)$$

$$b = \int_M b(x) \, dx. \tag{3.2.9}$$

It is well known that the above three objectives are closely related: if one achieves Objective 1, then Objectives 2 and 3 follow via Fourier Tauberian theorems [27, 72, 45, 70].

We are now in a position to state the main results from [22].

Result 1. We construct the propagator as a sum of m oscillatory integrals

$$U(t) \stackrel{\text{mod} C^{\infty}}{=} \sum_{j} U^{(j)}(t) ,$$
 (3.2.10)

where the phase function of each oscillatory integral $U^{(j)}(t)$ is associated with the corresponding Hamiltonian $h^{(j)}(x,\xi)$. The symbol of the oscillatory integral $U^{(j)}(t)$ is a complex-valued $m \times m$ matrix-function $u^{(j)}(t; y, \eta)$, where $y = (y^1, \ldots, y^n)$ is the position of the source of the wave (i.e. this is the same y that appears in formula (3.2.1)) and $\eta = (\eta_1, \ldots, \eta_n)$ is the corresponding dual variable. When $|\eta| \to +\infty$, the symbol admits an asymptotic expansion

$$u^{(j)}(t;y,\eta) = u_0^{(j)}(t;y,\eta) + u_{-1}^{(j)}(t;y,\eta) + \dots$$
(3.2.11)

into components positively homogeneous in η , with the subscript indicating degree of homogeneity.

The formula for the principal symbol of the oscillatory integral $U^{(j)}(t)$ is known [69, 64] and reads as follows:

$$u_0^{(j)}(t;y,\eta) = [v^{(j)}(x^{(j)}(t;y,\eta),\xi^{(j)}(t;y,\eta))] [v^{(j)}(y,\eta)]^* \\ \times \exp\left(-i\int_0^t q^{(j)}(x^{(j)}(\tau;y,\eta),\xi^{(j)}(\tau;y,\eta)) \,d\tau\right), \quad (3.2.12)$$

where $v^{(j)}(z,\zeta)$ is the normalised eigenvector of the principal symbol $A_1(z,\zeta)$ corresponding to the eigenvalue (Hamiltonian) $h^{(j)}(z,\zeta)$, $(x^{(j)}(t;y,\eta),\xi^{(j)}(t;y,\eta))$ is the Hamiltonian trajectory originating from the point (y,η) , and $q^{(j)}: T'M \to \mathbb{R}$ is the function

$$q^{(j)} := [v^{(j)}]^* A_{\text{sub}} v^{(j)} - \frac{i}{2} \{ [v^{(j)}]^*, A_1 - h^{(j)}, v^{(j)} \} - i [v^{(j)}]^* \{ v^{(j)}, h^{(j)} \}.$$
(3.2.13)

Curly brackets in formula (3.2.13) denote the Poisson bracket on matrix-functions

$$\{P,R\} := P_{z^{\alpha}}R_{\zeta_{\alpha}} - P_{\zeta_{\alpha}}R_{z^{\alpha}}$$
(3.2.14)

and its further generalisation

$$\{P, Q, R\} := P_{z^{\alpha}} Q R_{\zeta_{\alpha}} - P_{\zeta_{\alpha}} Q R_{z^{\alpha}}. \qquad (3.2.15)$$

Formula (3.2.12) is invariant under changes of local coordinates on the manifold M, i.e. elements of the $m \times m$ matrix-function $u_0^{(j)}(t; y, \eta)$ are scalars on $\mathbb{R} \times T'M$. Moreover, formula (3.2.12) is invariant under the transformation of the eigenvector of the principal symbol

$$v^{(j)} \mapsto e^{i\phi^{(j)}}v^{(j)},$$
 (3.2.16)

where

$$\phi^{(j)}: T'M \to \mathbb{R} \tag{3.2.17}$$

is an arbitrary smooth function. When some quantity is defined up to the action of a certain transformation, theoretical physicists refer to such a transformation as a gauge transformation. We follow this tradition. Note that our particular gauge transformation (3.2.16), (3.2.17) is quite common in quantum mechanics: when $\phi^{(j)}$ is a function of the position variable x only (i.e. when $\phi^{(j)}: M \to \mathbb{R}$) this gauge transformation is associated with electromagnetism.

Both Yu. Safarov [69] and W.J. Nicoll [64] assumed that the operator A is semi-bounded from below but this assumption is not essential and their formula (3.2.12) remains true in the more general case that we are dealing with.

However, knowing the principal symbol (3.2.12) of the oscillatory integral $U^{(j)}(t)$ is not enough if one wants to derive two-term asymptotics (3.2.6) and (3.2.7). One needs information about $u_{-1}^{(j)}(t; y, \eta)$, the component of the symbol of the oscillatory integral $U^{(j)}(t)$ which is positively homogeneous in η of degree -1, see formula (3.2.11), but here the problem is that $u_{-1}^{(j)}(t; y, \eta)$ is not a true invariant in the sense that it depends on the choice of phase function in the oscillatory integral. We overcome this difficulty by observing that $U^{(j)}(0)$ is a pseudodifferential operator, hence, it has a well-defined subprincipal symbol $[U^{(j)}(0)]_{sub}$. We prove that

$$\operatorname{tr}[U^{(j)}(0)]_{\mathrm{sub}} = -i\{[v^{(j)}]^*, v^{(j)}\}$$
(3.2.18)

and subsequently show that information contained in formulae (3.2.12) and (3.2.18) is sufficient for the derivation of two-term asymptotics (3.2.6) and (3.2.7).

Note that the RHS of formula (3.2.18) is invariant under the gauge transformation (3.2.16), (3.2.17).

Formula (3.2.18) plays a central role in [22].

Let us elaborate briefly on the geometric meaning of the RHS of (3.2.18) (a more detailed exposition is presented in Section 3.4). The eigenvector of the principal symbol is defined up to a gauge transformation (3.2.16), (3.2.17) so it is natural to introduce a U(1) connection on T'M as follows: when parallel transporting an eigenvector of the principal symbol along a curve in T'M we require that the derivative of the eigenvector along the curve be orthogonal to the eigenvector itself. This is equivalent to the introduction of an (intrinsic) electromagnetic field on T'M, with the 2*n*-component real quantity

$$i\left(\left[v^{(j)}\right]^* v^{(j)}_{x^{\alpha}}, \left[v^{(j)}\right]^* v^{(j)}_{\mathcal{E}_{\alpha}}\right)$$
(3.2.19)

playing the role of the electromagnetic covector potential. Our quantity (3.2.19) is a 1-form on T'M, rather than on M itself as is the case in "traditional" electromagnetism. The above U(1) connection generates curvature which is a 2-form on T'M, an analogue of the electromagnetic tensor. Out of this curvature 2-form one can construct, by contraction of indices, a real scalar. This scalar curvature is the expression appearing in the RHS of formula (3.2.18).

Observe now that $\sum_{j} U^{(j)}(0)$ is the identity operator on half-densities. The subprincipal symbol of the identity operator is zero, so formula (3.2.18) implies

$$\sum_{j} \{ [v^{(j)}]^*, v^{(j)} \} = 0.$$
(3.2.20)

One can check the identity (3.2.20) directly, without constructing the oscillatory integrals $U^{(j)}(t)$: it follows from the fact that the $v^{(j)}(x,\xi)$ form an orthonormal basis, see end of Section 3.4 for details. We mentioned the identity (3.2.20) in order to highlight, once again, the fact that the curvature effects we have identified are specific to systems and do not have an analogue in the scalar case.

Results 2 and 3. We prove, under appropriate assumptions on Hamiltonian trajectories (see Theorems 8.3 and 8.4 from [22], asymptotic formulae (3.2.6) and (3.2.7) with

$$a(x) = \sum_{j=1}^{m^+} \int_{h^{(j)}(x,\xi) < 1} d\xi, \qquad (3.2.21)$$

$$b(x) = -n \sum_{j=1}^{m^+} \int_{h^{(j)}(x,\xi) < 1} \left([v^{(j)}]^* A_{\text{sub}} v^{(j)} - \frac{i}{2} \{ [v^{(j)}]^*, A_1 - h^{(j)}, v^{(j)} \} + \frac{i}{n-1} h^{(j)} \{ [v^{(j)}]^*, v^{(j)} \} \right) (x,\xi) \, d\xi \,, \quad (3.2.22)$$

and a and b expressed via the above densities (3.2.21) and (3.2.22) as (3.2.8)and (3.2.9). In (3.2.21) and (3.2.22) the Poisson bracket on matrix-functions $\{\cdot, \cdot\}$ and its further generalisation $\{\cdot, \cdot, \cdot\}$ are defined by formulae (3.2.14)and (3.2.15) respectively.

To our knowledge, formula (3.2.22) is a new result. Note that in [69] this formula (more precisely, its integrated over M version (3.2.9)) was written incorrectly, without the curvature terms

$$-\frac{ni}{n-1}\int h^{(j)}\{[v^{(j)}]^*, v^{(j)}\}d\xi.$$
(3.2.23)

See also Section 3.6 where we give a more detailed bibliographic review.

It is easy to see that the right-hand sides of (3.2.21) and (3.2.22) behave as densities under changes of local coordinates on the manifold M and that these expressions are invariant under gauge transformations (3.2.16), (3.2.17) of the eigenvectors of the principal symbol. Moreover, the right-hand sides of (3.2.21) and (3.2.22) are unitarily invariant, i.e. invariant under the transformation of the operator

$$A \mapsto RAR^*, \tag{3.2.24}$$

where

$$R: M \to \mathrm{U}(m) \tag{3.2.25}$$

is an arbitrary smooth unitary matrix-function. The fact that the RHS of (3.2.22) is unitarily invariant is non-trivial: the appropriate calculations are presented in Section 9 of [22]. The observation that without the curvature terms (3.2.23) (as in [69]) the RHS of (3.2.22) is not unitarily invariant was a major motivating factor in the writing of [22].

Formula (3.2.22) is the main result of this chapter. Note that even though the two-term asymptotic expansion (3.2.6) holds only under certain assumptions on Hamiltonian trajectories (loops), the second asymptotic coefficient (3.2.22) is, in itself, well-defined irrespective of how many loops we have. If one wishes to reformulate the asymptotic expansion (3.2.6) in such a way that it remains valid without assumptions on the number of loops, this can easily be achieved, say, by taking a convolution with a function from Schwartz space $\mathcal{S}(\mathbb{R})$: see Theorem 7.1 from [22].

3.3 Overview of the algorithm for the construction of the propagator

We construct the propagator as a sum of m oscillatory integrals (3.2.10) where each integral is of the form

$$U^{(j)}(t) = \int e^{i\varphi^{(j)}(t,x;y,\eta)} u^{(j)}(t;y,\eta) \varsigma^{(j)}(t,x;y,\eta) d_{\varphi^{(j)}}(t,x;y,\eta) (\cdot) dy \,d\eta \,, \ (3.3.1)$$

defined in accordance with Subsection 3.1.4. Our task now is to construct the symbols $u_0^{(j)}(t; y, \eta)$, j = 1, ..., m, so that our oscillatory integrals $U^{(j)}(t)$, j =

 $1, \ldots, m$, satisfy the dynamic equations

$$(D_t + A(x, D_x)) U^{(j)}(t) \stackrel{\text{mod } C^{\infty}}{=} 0$$
(3.3.2)

and initial condition

$$\sum_{j} U^{(j)}(0) \stackrel{\text{mod} C^{\infty}}{=} I, \qquad (3.3.3)$$

where I is the identity operator on half-densities; compare with formulae (3.2.3), (3.2.2) and (3.2.10). Note that the pseudodifferential operator A in formula (3.3.2) acts on the oscillatory integral U(t) in the variable x; say, if A is a differential operator this means that in order to evaluate $A U^{(j)}(t)$ one has to perform the appropriate differentiations of the oscillatory integral (3.3.1) in the variable x. Following the conventions of Section 3.3 of [72], we emphasise the fact that the pseudodifferential operator A in formula (3.3.2) acts on the oscillatory integral U(t) in the variable x by writing this pseudodifferential operator as $A(x, D_x)$, where $D_{x^{\alpha}} := -i\partial/\partial x^{\alpha}$.

We examine first the dynamic equation (3.3.2). We have

$$(D_t + A(x, D_x)) U^{(j)}(t) = F^{(j)}(t),$$

where $F^{(j)}(t)$ is the oscillatory integral

$$F^{(j)}(t) = \int e^{i\varphi^{(j)}(t,x;y,\eta)} f^{(j)}(t,x;y,\eta) \varsigma^{(j)}(t,x;y,\eta) d_{\varphi^{(j)}}(t,x;y,\eta) (\cdot) dy \,d\eta$$

whose matrix-valued amplitude $f^{(j)}$ is given by the formula

$$f^{(j)} = D_t u^{(j)} + \left(\varphi_t^{(j)} + (d_{\varphi^{(j)}})^{-1} (D_t d_{\varphi^{(j)}}) + s^{(j)}\right) u^{(j)}, \qquad (3.3.4)$$

where the matrix-function $s^{(j)}(t, x; y, \eta)$ is defined as

$$s^{(j)} = e^{-i\varphi^{(j)}} (d_{\varphi^{(j)}})^{-1} A(x, D_x) \left(e^{i\varphi^{(j)}} d_{\varphi^{(j)}} \right).$$
(3.3.5)

When $|\eta| \to +\infty$ the matrix-valued amplitude $f^{(j)}(t, x; y, \eta)$ defined by formula (3.3.4) admits an asymptotic expansion

$$f^{(j)}(t,x;y,\eta) = f_1^{(j)}(t,x;y,\eta) + f_0^{(j)}(t,x;y,\eta) + f_{-1}^{(j)}(t,x;y,\eta) + \dots$$
(3.3.6)

into components positively homogeneous in η , with the subscript indicating degree of homogeneity.

Note that in (3.3.1) there are no terms containing derivatives of $\varsigma^{(j)}(t, x; y, \eta)$: as $\varsigma^{(j)}(t, x; y, \eta)$ is infinitely smooth in all independent variables, its' derivatives give an infinitely smooth contribution which is dealt with by the "mod C^{∞} " in (3.3.2).

Note the following differences between formulae (3.2.11) and (3.3.6).

• The leading term in (3.3.6) has degree of homogeneity 1, rather than 0 as in (3.2.11). In fact, the leading term in (3.3.6) can be easily written out explicitly

$$f_1^{(j)}(t,x;y,\eta) = \left(\varphi_t^{(j)}(t,x;y,\eta) + A_1(x,\varphi_x^{(j)}(t,x;y,\eta))\right) u_0^{(j)}(t;y,\eta), \quad (3.3.7)$$

where $A_1(x,\xi)$ is the (matrix-valued) principal symbol of the pseudodifferential operator A.

• Unlike the symbol $u^{(j)}(t; y, \eta)$, the amplitude $f^{(j)}(t, x; y, \eta)$ depends on x.

We now need to exclude the dependence on x from the amplitude $f^{(j)}(t, x; y, \eta)$. This can be done by means of the algorithm described in Subsection 2.7.3 of [72]. We outline this algorithm below.

Working in local coordinates, define the matrix-function $\varphi_{x\eta}^{(j)}$ in accordance with $(\varphi_{x\eta}^{(j)})_{\alpha}{}^{\beta} := \varphi_{x^{\alpha}\eta_{\beta}}^{(j)}$ and then define its inverse $(\varphi_{x\eta}^{(j)})^{-1}$ from the identity

$$(\varphi_{x\eta}^{(j)})_{\alpha}{}^{\beta}[(\varphi_{x\eta}^{(j)})^{-1}]_{\beta}{}^{\gamma} := \delta_{\alpha}{}^{\gamma}.$$

Define the "scalar" first order linear differential operators

$$L_{\alpha}^{(j)} := [(\varphi_{x\eta}^{(j)})^{-1}]_{\alpha}{}^{\beta} (\partial/\partial x^{\beta}), \qquad \alpha = 1, \dots, n.$$
(3.3.8)

Note that the coefficients of these differential operators are functions of the position variable x and the dual variable ξ (as well as t and y). It is known, see part 2 of Appendix E in [72], that the operators (3.3.8) commute: $L_{\alpha}^{(j)}L_{\beta}^{(j)} = L_{\beta}^{(j)}L_{\alpha}^{(j)}$, $\alpha, \beta = 1, \ldots, n$.

Denote $L_{\boldsymbol{\alpha}}^{(j)} := (L_1^{(j)})^{\alpha_1} \cdots (L_n^{(j)})^{\alpha_n}$, $(-\varphi_{\eta}^{(j)})^{\boldsymbol{\alpha}} := (-\varphi_{\eta_1}^{(j)})^{\alpha_1} \cdots (-\varphi_{\eta_n}^{(j)})^{\alpha_n}$, and, given an $r \in \mathbb{N}$, define the "scalar" linear differential operator

$$\mathfrak{P}_{-1,r}^{(j)} := i(d_{\varphi^{(j)}})^{-1} \frac{\partial}{\partial \eta_{\beta}} d_{\varphi^{(j)}} \left(1 + \sum_{1 \le |\boldsymbol{\alpha}| \le 2r-1} \frac{(-\varphi_{\eta}^{(j)})^{\boldsymbol{\alpha}}}{\boldsymbol{\alpha}! \left(|\boldsymbol{\alpha}|+1\right)} L_{\boldsymbol{\alpha}}^{(j)} \right) L_{\beta}^{(j)}, \quad (3.3.9)$$

where $|\boldsymbol{\alpha}| := \alpha_1 + \ldots + \alpha_n$, $\boldsymbol{\alpha}! = \alpha_1! \cdots \alpha_n!$, and the repeated index β indicates summation over $\beta = 1, \ldots, n$.

Recall Definition 2.7.8 from [72]: the linear operator L is said to be positively homogeneous in η of degree $p \in \mathbb{R}$ if for any $q \in \mathbb{R}$ and any function f positively homogeneous in η of degree q the function Lf is positively homogeneous in η of degree p + q. It is easy to see that the operator (3.3.9) is positively homogeneous in η of degree -1 and the first subscript in $\mathfrak{P}_{-1,r}^{(j)}$ emphasises this fact.

Let $\mathfrak{S}_0^{(j)}$ be the (linear) operator of restriction to $x = x^{(j)}(t; y, \eta)$,

$$\mathfrak{S}_{0}^{(j)} := (\cdot)|_{x=x^{(j)}(t;y,\eta)} , \qquad (3.3.10)$$

and let

$$\mathfrak{S}_{-r}^{(j)} := \mathfrak{S}_0^{(j)} (\mathfrak{P}_{-1,r}^{(j)})^r \tag{3.3.11}$$

for $r = 1, 2, \ldots$ Observe that our linear operators $\mathfrak{S}_{-r}^{(j)}$, $r = 0, 1, 2, \ldots$, are positively homogeneous in η of degree -r. This observation allows us to define

the linear operator

$$\mathfrak{S}^{(j)} := \sum_{r=0}^{+\infty} \mathfrak{S}^{(j)}_{-r} , \qquad (3.3.12)$$

where the series is understood as an asymptotic series in inverse powers of η .

According to Subsection 2.7.3 of [72], the dynamic equation (3.3.2) can now be rewritten in the equivalent form

$$\mathfrak{S}^{(j)}f^{(j)} = 0, \qquad (3.3.13)$$

where the equality is understood in the asymptotic sense, as an asymptotic expansion in inverse powers of η . Recall that the matrix-valued amplitude $f^{(j)}(t, x; y, \eta)$ appearing in (3.3.13) is defined by formulae (3.3.4) and (3.3.5).

Substituting (3.3.12) and (3.3.6) into (3.3.13) we obtain a hierarchy of equations

$$\mathfrak{S}_0^{(j)} f_1^{(j)} = 0, \qquad (3.3.14)$$

$$\mathfrak{S}_{-1}^{(j)} f_1^{(j)} + \mathfrak{S}_0^{(j)} f_0^{(j)} = 0, \qquad (3.3.15)$$

$$\mathfrak{S}_{-2}^{(j)} f_1^{(j)} + \mathfrak{S}_{-1}^{(j)} f_0^{(j)} + \mathfrak{S}_0^{(j)} f_{-1}^{(j)} = 0,$$

positively homogeneous in η of degree 1, 0, -1, These are the *transport* equations for the determination of the unknown homogeneous components $u_0^{(j)}(t; y, \eta)$, $u_{-1}^{(j)}(t; y, \eta)$, $u_{-2}^{(j)}(t; y, \eta)$, ..., of the symbol of the oscillatory integral (3.3.1).

. . .

Let us now examine the initial condition (3.3.3). Each operator $U^{(j)}(0)$ is a pseudodifferential operator, only written in a slightly nonstandard form. The issues here are as follows.

• We use the invariantly defined phase function $\varphi^{(j)}(0, x; y, \eta) = (x - y)^{\alpha} \eta_{\alpha} + O(|x - y|^2)$ rather than the linear phase function $(x - y)^{\alpha} \eta_{\alpha}$ written in local coordinates.

- When defining the (full) symbol of the operator $U^{(j)}(t)$ we excluded the variable x from the amplitude rather than the variable y. Note that when dealing with pseudodifferential operators it is customary to exclude the variable y from the amplitude; exclusion of the variable x gives the dual symbol of a pseudodifferential operator, see Subsection 2.1.3 in [72]. Thus, at t = 0, our symbol $u^{(j)}(0; y, \eta)$ resembles the dual symbol of a pseudodifferential operator is the dual symbol of a pseudodifferential operator.
- We have the extra factor $d_{\varphi^{(j)}}(0, x; y, \eta)$ in our representation of the operator $U^{(j)}(0)$ as an oscillatory integral.

The (full) dual symbol of the pseudodifferential operator $U^{(j)}(0)$ can be calculated in local coordinates in accordance with the following formula which addresses the issues highlighted above:

$$\sum_{\boldsymbol{\alpha}} \frac{(-1)^{|\boldsymbol{\alpha}|}}{\boldsymbol{\alpha}!} \left(D_x^{\boldsymbol{\alpha}} \partial_{\eta}^{\boldsymbol{\alpha}} \left[u^{(j)}(0;y,\eta) e^{i\omega^{(j)}(x;y,\eta)} d_{\varphi^{(j)}}(0,x;y,\eta) \right] \right) \Big|_{x=y} , \quad (3.3.16)$$

where $\omega^{(j)}(x; y, \eta) = \varphi^{(j)}(0, x; y, \eta) - (x - y)^{\beta} \eta_{\beta}$. Formula (3.3.16) is a version of the formula from Subsection 2.1.3 of [72], only with the extra factor $(-1)^{|\alpha|}$. The latter is needed because we are writing down the dual symbol of the pseudodifferential operator $U^{(j)}(0)$ (no dependence on x) rather than its "normal" symbol (no dependence on y).

The initial condition (3.3.3) can now be rewritten in explicit form as

$$\sum_{j} \sum_{\boldsymbol{\alpha}} \frac{(-1)^{|\boldsymbol{\alpha}|}}{\boldsymbol{\alpha}!} \left(D_{x}^{\boldsymbol{\alpha}} \partial_{\eta}^{\boldsymbol{\alpha}} \left[u^{(j)}(0; y, \eta) e^{i\omega^{(j)}(x; y, \eta)} d_{\varphi^{(j)}}(0, x; y, \eta) \right] \right) \Big|_{x=y} = I,$$
(3.3.17)

where I is the $m \times m$ identity matrix. Condition (3.3.17) can be decomposed into components positively homogeneous in η of degree $0, -1, -2, \ldots$, giving us a hierarchy of initial conditions. The leading (of degree of homogeneity 0) initial condition reads

$$\sum_{j} u_0^{(j)}(0; y, \eta) = I, \qquad (3.3.18)$$

whereas lower order initial conditions are more complicated and depend on the choice of our phase functions $\varphi^{(j)}$.

Sections 3 and 4 of [22] go on to apply the algorithm presented above. In particular, (3.2.12) and (3.2.18) are proved. This provides us with sufficient information concerning the propagator $U^{(j)}(t)$ to apply Fourier Tauberian theorems which, in turn, allow us to determine the asymptotics of the spectral and counting functions (3.2.6) and (3.2.7). The appropriate calculations, under suitable assumptions on Hamiltonian trajectories, are given in Sections 6-8 of [22]. In particular, Theorems 8.3 and 8.4 from [22] provide the required conditions on Hamiltonian trajectories.

$3.4 \quad U(1) \text{ connection}$

In the Sections 2-4 of [22] we presented technical details of the construction of the propagator; an overview of this scheme is presented in Section 3.3. We saw that the eigenvectors of the principal symbol, $v^{(j)}(x,\xi)$, play a major role in this construction. As pointed out in Section 3.2, each of these eigenvectors is defined up to a U(1) gauge transformation (3.2.16), (3.2.17). In the end, the full symbols (3.2.11) of our oscillatory integrals $U^{(j)}(t)$ do not depend on the choice of gauge for the eigenvectors $v^{(j)}(x,\xi)$. However, the effect of the gauge transformation (3.2.16), (3.2.17) is not as trivial as it may appear at first sight. We will demonstrate in this section that the gauge transformation (3.2.16), (3.2.17) shows up, in the form of invariantly defined curvature, in the lower order terms $u_{-1}^{(j)}(t; y, \eta)$ of the symbols of our oscillatory integrals $U^{(j)}(t)$. More precisely, we will show that the RHS of formula (3.2.18) is the scalar curvature of a connection associated with the gauge transformation (3.2.16), (3.2.17). Further on in this section, until the very last paragraph, the index j enumerating eigenvalues and eigenvectors of the principal symbol is assumed to be fixed.

Consider a smooth curve $\Gamma \subset T'M$ connecting points (y, η) and (x, ξ) . We write this curve in parametric form as $(z(t), \zeta(t)), t \in [0, 1]$, so that $(z(0), \zeta(0)) = (y, \eta)$ and $(z(1), \zeta(1)) = (x, \xi)$. Put

$$w(t) := e^{i\phi(t)} v^{(j)}(z(t), \zeta(t)), \qquad (3.4.1)$$

where $\phi : [0,1] \to \mathbb{R}$ is an unknown function which is to be determined from the condition

$$iw^*\dot{w} = 0 \tag{3.4.2}$$

with the dot indicating the derivative with respect to the parameter t. Substituting (3.4.1) into (3.4.2) we get an ordinary differential equation for ϕ which is easily solved, giving

$$\phi(1) = \phi(0) + \int_0^1 (\dot{z}^{\alpha}(t) P_{\alpha}(z(t), \zeta(t)) + \dot{\zeta}_{\gamma}(t) Q^{\gamma}(z(t), \zeta(t))) dt$$

= $\phi(0) + \int_{\Gamma} (P_{\alpha} dz^{\alpha} + Q^{\gamma} d\zeta_{\gamma}), \quad (3.4.3)$

where

$$P_{\alpha} := i[v^{(j)}]^* v_{z^{\alpha}}^{(j)}, \qquad Q^{\gamma} := i[v^{(j)}]^* v_{\zeta_{\gamma}}^{(j)}. \tag{3.4.4}$$

Note that the 2*n*-component real quantity (P_{α}, Q^{γ}) is a covector field (1-form) on T'M. This quantity already appeared in Section 3.2 as formula (3.2.19).

Put $f(y,\eta) := e^{i\phi(0)}, f(x,\xi) := e^{i\phi(1)}$ and rewrite formula (3.4.3) as

$$f(x,\xi) = f(y,\eta) e^{i \int_{\Gamma} (P_{\alpha} dz^{\alpha} + Q^{\gamma} d\zeta_{\gamma})}.$$
(3.4.5)

Let us identify the group U(1) with the unit circle in the complex plane, i.e. with $f \in \mathbb{C}$, |f| = 1. We see that formulae (3.4.5) and (3.4.4) give us a rule for the parallel transport of elements of the group U(1) along curves in T'M. This is the natural U(1) connection generated by the normalised field of columns of complex-valued scalars

$$v^{(j)}(z,\zeta) = \left(v_1^{(j)}(z,\zeta) \quad \dots \quad v_m^{(j)}(z,\zeta)\right)^T.$$
 (3.4.6)

Recall that the Γ appearing in formula (3.4.5) is a curve connecting points (y, η)

and (x, ξ) , whereas the $v^{(j)}(z, \zeta)$ appearing in formulae (3.4.4) and (3.4.6) enters our construction as an eigenvector of the principal symbol of our $m \times m$ matrix pseudodifferential operator A.

In practice, dealing with a connection is not as convenient as dealing with the covariant derivative ∇ . The covariant derivative corresponding to the connection (3.4.5) is determined as follows. Let us view the (x,ξ) appearing in formula (3.4.5) as a variable which takes values close to (y,η) , and suppose that the curve Γ is a short straight (in local coordinates) line segment connecting the point (y,η) with the point (x,ξ) . We want the covariant derivative of our function $f(x,\xi)$, evaluated at (y,η) , to be zero. Examination of formula (3.4.5) shows that the unique covariant derivative satisfying this condition is

$$\nabla_{\alpha} := \partial/\partial x^{\alpha} - iP_{\alpha}(x,\xi), \qquad \nabla^{\gamma} := \partial/\partial \xi_{\gamma} - iQ^{\gamma}(x,\xi). \tag{3.4.7}$$

We define the curvature of our U(1) connection as

$$R := -i \begin{pmatrix} \nabla_{\alpha} \nabla_{\beta} - \nabla_{\beta} \nabla_{\alpha} & \nabla_{\alpha} \nabla^{\delta} - \nabla^{\delta} \nabla_{\alpha} \\ \nabla^{\gamma} \nabla_{\beta} - \nabla_{\beta} \nabla^{\gamma} & \nabla^{\gamma} \nabla^{\delta} - \nabla^{\delta} \nabla^{\gamma} \end{pmatrix}.$$
 (3.4.8)

It may seem that the entries of the $(2n) \times (2n)$ matrix (3.4.8) are differential operators. They are, in fact, operators of multiplication by "scalar functions". Namely, the more explicit form of (3.4.8) is

$$R = \begin{pmatrix} \frac{\partial P_{\alpha}}{\partial x^{\beta}} - \frac{\partial P_{\beta}}{\partial x^{\alpha}} & \frac{\partial P_{\alpha}}{\partial \xi_{\delta}} - \frac{\partial Q^{\delta}}{\partial x^{\alpha}} \\ \frac{\partial Q^{\gamma}}{\partial x^{\beta}} - \frac{\partial P_{\beta}}{\partial \xi_{\gamma}} & \frac{\partial Q^{\gamma}}{\partial \xi_{\delta}} - \frac{\partial Q^{\delta}}{\partial \xi_{\gamma}} \end{pmatrix}.$$
 (3.4.9)

The $(2n) \times (2n)$ - component real quantity (3.4.9) is a rank 2 covariant antisymmetric tensor (2-form) on T'M. It is an analogue of the electromagnetic tensor.

Substituting (3.4.4) into (3.4.9) we get an expression for curvature in terms of the eigenvector of the principal symbol

$$R = i \begin{pmatrix} [v_{x^{\beta}}^{(j)}]^* v_{x^{\alpha}}^{(j)} - [v_{x^{\alpha}}^{(j)}]^* v_{x^{\beta}}^{(j)} & [v_{\xi_{\delta}}^{(j)}]^* v_{x^{\alpha}}^{(j)} - [v_{x^{\alpha}}^{(j)}]^* v_{\xi_{\delta}}^{(j)} \\ [v_{x^{\beta}}^{(j)}]^* v_{\xi_{\gamma}}^{(j)} - [v_{\xi_{\gamma}}^{(j)}]^* v_{x^{\beta}}^{(j)} & [v_{\xi_{\delta}}^{(j)}]^* v_{\xi_{\gamma}}^{(j)} - [v_{\xi_{\gamma}}^{(j)}]^* v_{\xi_{\delta}}^{(j)} \end{pmatrix}.$$
(3.4.10)

Examination of formula (3.4.10) shows that, as expected, curvature is invariant under the gauge transformation (3.2.16), (3.2.17).

It is natural to take the trace of the upper right block in (3.4.8) which, in the notation (3.2.14), gives us

$$-i(\nabla_{\alpha}\nabla^{\alpha} - \nabla^{\alpha}\nabla_{\alpha}) = -i\{[v^{(j)}]^*, v^{(j)}\}.$$
(3.4.11)

Thus, we have shown that the RHS of formula (3.2.18) is the scalar curvature of our U(1) connection.

We end this section by proving, as promised in Section 3.2, formula (3.2.20) without referring to microlocal analysis. In the following arguments we use our standard notation for the orthogonal projections onto the eigenspaces of the principal symbol, i.e. we write $P^{(k)} := v^{(k)}[v^{(k)}]^*$. We have $\operatorname{tr}\{P^{(j)}, P^{(j)}\} = 0$ and $\sum_l P^{(l)} = I$ which implies

$$0 = \sum_{l,j} \operatorname{tr}(P^{(l)}\{P^{(j)}, P^{(j)}\})$$
$$= \sum_{j} \operatorname{tr}(P^{(j)}\{P^{(j)}, P^{(j)}\}) + \sum_{l,j: \ l \neq j} \operatorname{tr}(P^{(l)}\{P^{(j)}, P^{(j)}\}). \quad (3.4.12)$$

But, according to formula (4.15) from [22], for $l \neq j$ we have

$$\operatorname{tr}(P^{(l)}\{P^{(j)}, P^{(j)}\}) = -\operatorname{tr}(P^{(j)}\{P^{(l)}, P^{(l)}\}),$$

so the expression in the last sum in the RHS of (3.4.12) is antisymmetric in the indices l, j, which implies that this sum is zero. Hence, formula (3.4.12) can be

rewritten as

$$\sum_{j} \operatorname{tr}(P^{(j)}\{P^{(j)}, P^{(j)}\}) = 0.$$

It remains only to note that, according to formula (4.17) from [22],

$$\operatorname{tr}(P^{(j)}\{P^{(j)}, P^{(j)}\}) = \{[v^{(j)}]^*, v^{(j)}\}\$$

3.5 Spectral asymmetry

In this section we deal with the special case when the operator A is differential (as opposed to pseudodifferential). Our aim is to examine what happens when we change the sign of the operator. In other words, we compare the original operator A with the operator $\tilde{A} := -A$. In theoretical physics the transformation $A \mapsto -A$ would be interpreted as time reversal, see equation (3.2.3).

It is easy to see that for a differential operator the number m (number of equations in our system) has to be even and that the principal symbol has to have the same number of positive and negative eigenvalues. In the notation of Section 3.2 this fact can be expressed as $m = 2m^+ = 2m^-$.

It is also easy to see that the principal symbols of the two operators, A and A, and the eigenvalues and eigenvectors of the principal symbols are related as

$$A_1(x,\xi) = \tilde{A}_1(x,-\xi), \qquad (3.5.1)$$

$$h^{(j)}(x,\xi) = \tilde{h}^{(j)}(x,-\xi), \qquad (3.5.2)$$

$$v^{(j)}(x,\xi) = \tilde{v}^{(j)}(x,-\xi), \qquad (3.5.3)$$

whereas the subprincipal symbols are related as

$$A_{\rm sub}(x) = -\tilde{A}_{\rm sub}(x). \tag{3.5.4}$$

Formulae (3.2.21), (3.2.22), (3.2.15), (3.2.14) and (3.5.1)–(3.5.4) imply

$$a(x) = \tilde{a}(x), \qquad b(x) = -\tilde{b}(x).$$
 (3.5.5)

Substituting (3.5.5) into (3.2.8) and (3.2.9) we get

$$a = \tilde{a}, \qquad b = -\tilde{b}. \tag{3.5.6}$$

Formulae (3.2.7) and (3.5.6) imply that the spectrum of a generic first order differential operator is asymmetric about $\lambda = 0$. This phenomenon is known as *spectral asymmetry* [4, 5, 6, 7].

If we square our operator A and consider the spectral problem $A^2v = \lambda^2 v$, then the terms $\pm b\lambda^{n-1}$ cancel out and the second asymptotic coefficient of the counting function (as well as the spectral function) of the operator A^2 turns to zero. This is in agreement with the known fact that for an even order semi-bounded matrix differential operator acting on a manifold without boundary the second asymptotic coefficient of the counting function is zero, see Section 6 of [80] and [68].

3.6 Bibliographic review

To our knowledge, the first publication on the subject of two-term spectral asymptotics for systems was Ivrii's 1980 paper [42] in Section 2 of which the author stated, without proof, a formula for the second asymptotic coefficient of the counting function. In a subsequent 1982 paper [43] Ivrii acknowledged that the formula from [42] was incorrect and gave a new formula, labelled (0.6), followed by a "proof". In his 1984 Springer Lecture Notes [44] Ivrii acknowledged on page 226 that both his previous formulae for the second asymptotic coefficient were incorrect and stated, without proof, yet another formula. Roughly at the same time Rozenblyum [67] also stated a formula for the second asymptotic coefficient of the counting function of a first order system.

The formulae from [42], [43] and [67] are fundamentally flawed because they are proportional to the subprincipal symbol. As our formulae (3.2.9) and (3.2.22) show, the second asymptotic coefficient of the counting function may be nonzero even when the subprincipal symbol is zero. This illustrates, yet again, the difference between scalar operators and systems.

The formula on page 226 of [44] gives an algorithm for the calculation of the correction term designed to take account of the effect described in the previous paragraph. This algorithm requires the evaluation of a limit of a complicated expression involving the integral, over the cotangent bundle, of the trace of the symbol of the resolvent of the operator A constructed by means of pseudodifferential calculus. This algorithm was revisited in Ivrii's 1998 book, see formulae (4.3.39) and (4.2.25) in [45].

The next contributor to the subject was Safarov who, in his 1989 DSc Thesis [69], wrote down a formula for the second asymptotic coefficient of the counting function which was "almost" correct. This formula appears in [69] as formula (2.4). As explained in Section 3.2, Safarov lost only the curvature terms (3.2.23). Safarov's DSc Thesis [69] provides arguments which are sufficiently detailed and we were able to identify the precise point (page 163) at which the mistake occurred.

In 1998 Nicoll rederived [64] Safarov's formula (3.2.12) for the principal symbols of the propagator, using a method slightly different from [69], but stopped short of calculating the second asymptotic coefficient of the counting function.

In 2007 Kamotski and Ruzhansky [47] performed an analysis of the propagator of a first order elliptic system based on the approach of Rozenblyum [67], but stopped short of calculating the second asymptotic coefficient of the counting function.

Chapter 4

Microlocal analysis of first order systems II

In this chapter we present a detailed consideration of elliptic first order selfadjoint differential operators acting over a connected compact 3-manifold without boundary, with an eventual focus on the massless Dirac operator (1.1.1).

The motivation for this study is simple. The massless Dirac operator in three dimensions considered in Chapter 4 is a special case of the type of operator considered in Chapter 3. We are therefore in a position to apply the theory developed in Chapter 3 to the massless Dirac operator.

4.1 Main results

Consider a first order differential operator A acting on 2-columns $v = \begin{pmatrix} v_1 & v_2 \end{pmatrix}^T$ of complex-valued half-densities over a connected compact 3-dimensional manifold M without boundary. (cf. Section 1.3 for the definition of a half-density.) We assume the coefficients of the operator A to be infinitely smooth. We also assume that the operator A is formally self-adjoint (symmetric):

$$\int_{M} w^{*}(Av) \, dx = \int_{M} (Aw)^{*} v \, dx \tag{4.1.1}$$

for all infinitely smooth $v, w : M \to \mathbb{C}^2$. Here and further on the superscript * in matrices, rows and columns indicates Hermitian conjugation in \mathbb{C}^2 and $dx := dx^1 dx^2 dx^3$, where $x = (x^1, x^2, x^3)$ are local coordinates on M.

Let $A_1(x,\xi)$ be the principal symbol of the operator A, i.e. matrix obtained by leaving in A only the leading (first order) derivatives and replacing each $\partial/\partial x^{\alpha}$ by $i\xi_{\alpha}, \alpha = 1, 2, 3$. Here $\xi = (\xi_1, \xi_2, \xi_3)$ is the variable dual to the position variable x. Our principal symbol $A_1(x,\xi)$ is a 2 × 2 Hermitian matrix-function on the cotangent bundle T^*M , linear in every fibre T^*_xM (i.e. linear in ξ).

Throughout Chapter 4 we assume that the principal symbol $A_1(x,\xi)$ is trace-free for all $(x,\xi) \in T^*M$ and that

$$\det A_1(x,\xi) \neq 0, \qquad \forall (x,\xi) \in T'M, \tag{4.1.2}$$

where $T'M := T^*M \setminus \{\xi = 0\}$ (cotangent bundle with the zero section removed). The assumption (4.1.2) is a version of the ellipticity condition (cf. Subsection 3.1.3).

Under the above assumptions A is a self-adjoint operator in $L^2(M; \mathbb{C}^2)$ (Hilbert space of square integrable complex-valued column "functions") with domain $H^1(M; \mathbb{C}^2)$ (Sobolev space of complex-valued column "functions" which are square integrable together with their first partial derivatives) and the spectrum of A is discrete, with eigenvalues accumulating to $\pm \infty$. Let λ_k and $v_k = \left(v_{k1}(x) \quad v_{k2}(x)\right)^T$ be the eigenvalues and eigenfunctions of the operator A. The eigenvalues λ_k are enumerated in increasing order with account of multiplicity, using a positive index $k = 1, 2, \ldots$ for positive λ_k and a nonpositive index $k = 0, -1, -2, \ldots$ for nonpositive λ_k . We will be studying the spectral function (3.2.4) and the counting function (3.2.5) for the type of operator described above.

We aim to derive, under appropriate assumptions on Hamiltonian trajectories, two-term asymptotics for the spectral function (3.2.4) and the counting function (3.2.5), i.e. formulae of the type (3.2.6) and (3.2.7), respectively, as $\lambda \to +\infty$, where the real constants a, b appearing in (3.2.7) and real densities a(x), b(x) in (3.2.6) are related in accordance with (3.2.8) and (3.2.9) respectively. Note that, in Chapter 4, we always work in three dimensions, which reduces the complexity of the formulae from Chapter 3 somewhat.

In [22] (an overview of which is given in Chapter 3) we performed a comprehensive analysis of two-term spectral asymptotics for general first order elliptic systems. In doing this we showed that all previous publications on systems gave formulae for the second asymptotic coefficient that were either incorrect or incomplete (i.e. an algorithm for the calculation of the second asymptotic coefficient rather than an explicit formula), see Section 3.6 for the appropriate bibliographic review. The correct formula for the coefficient b(x) was the main result presented in Chapter 3.

The problem examined in Chapter 4 is a special case of that from [22]. Namely, in Chapter 4 we make the following additional assumptions as compared to [22] and Chapter 3:

our manifold has dimension 3, (4.1.3)

the number of equations in our system is 2, (4.1.4)

our operator is differential (as opposed to pseudodifferential), (4.1.5)

the principal symbol is trace-free. (4.1.6)

The need for a detailed analysis of the special case (4.1.3)-(4.1.6) is driven by applications to the massless Dirac operator.

The additional assumptions (4.1.3)–(4.1.6) lead to the following simplifications as compared to Chapter 3.

- The subprincipal symbol A_{sub} does not depend on the dual variable ξ (momentum) and is a function of x (position) only. Recall that the subprincipal symbol is the zeroth order term of the full symbol of the first order operator A written in a way which makes it invariant under coordinate transformations, see formula (4.6.2) for formal definition and Subsection 2.1.3 in [72] for background material.
- The principal symbol A_1 admits a geometric description.

The first of these simplifications is trivial whereas the second is not. We list below the geometric objects encoded within the principal symbol.

Geometric object 1: the *metric.* Observe that the determinant of the principal symbol is a negative definite quadratic form

$$\det A_1(x,\xi) = -g^{\alpha\beta}\xi_\alpha\xi_\beta \tag{4.1.7}$$

and the coefficients $g^{\alpha\beta}(x) = g^{\beta\alpha}(x)$, $\alpha, \beta = 1, 2, 3$, appearing in (4.1.7) can be interpreted as components of a (contravariant) Riemannian metric. This implies, in particular, that our Hamiltonian (positive eigenvalue of the principal symbol) takes the form

$$h^{+}(x,\xi) = \sqrt{g^{\alpha\beta}(x)\,\xi_{\alpha}\xi_{\beta}} \tag{4.1.8}$$

and the x-components of our Hamiltonian trajectories become geodesics.

Geometric object 2: the *Teleparallel connection*. This is an affine connection defined as follows. Suppose we have a covector ξ based at the point $x \in M$ and we want to construct a parallel covector $\tilde{\xi}$ based at the point $\tilde{x} \in M$. This is done by solving the linear system of equations

$$A_1(\tilde{x}, \tilde{\xi}) = A_1(x, \xi).$$
 (4.1.9)

Equation (4.1.9) is equivalent to a system of three real linear algebraic equations for the three real unknowns, components of the covector $\tilde{\xi}$, and it is easy to see that this system has a unique solution. It is also easy to see that the affine connection defined by formula (4.1.9) preserves the Riemannian norm of covectors, i.e. $g^{\alpha\beta}(\tilde{x}) \tilde{\xi}_{\alpha} \tilde{\xi}_{\beta} = g^{\alpha\beta}(x) \xi_{\alpha} \xi_{\beta}$, hence, it is metric compatible. The parallel transport defined by formula (4.1.9) does not depend on the curve along which we transport the (co)vector, so our connection has zero curvature. Recall that the word "Teleparallel" (parallel at a distance) is used in theoretical physics [39] to describe metric compatible affine connections with zero curvature. The origins of this terminology go back to the works of A. Einstein and É. Cartan [79, 73, 18], though Cartan preferred to use the term "absolute parallelism" rather than "Teleparallelism", see Section 2.1 for further details.

The Teleparallel connection coefficients $\Gamma^{\alpha}{}_{\beta\gamma}(x)$ can be written down explicitly in terms of the principal symbol, see formula (4.3.7), and this allows us to define yet another geometric object — the torsion tensor

$$T^{\alpha}{}_{\beta\gamma} := \Gamma^{\alpha}{}_{\beta\gamma} - \Gamma^{\alpha}{}_{\gamma\beta} \,. \tag{4.1.10}$$

Further on we raise and lower indices of the torsion tensor using the metric.

Geometric object 3: the topological charge. It turns out, see Section 4.3, that the existence of a principal symbol implies that our manifold M is parallelizable. Parallelizability implies orientability. Having chosen a particular orientation, we allow only changes of local coordinates x^{α} , $\alpha = 1, 2, 3$, which preserve orientation.

We define the topological charge as

$$\mathbf{c} := -\frac{i}{2}\sqrt{\det g_{\alpha\beta}} \operatorname{tr}((A_1)_{\xi_1}(A_1)_{\xi_2}(A_1)_{\xi_3}), \qquad (4.1.11)$$

with the subscripts ξ_{α} indicating partial derivatives. We show in Section 4.3 that the number **c** defined by formula (4.1.11) can take only two values, +1 or -1, and describes the orientation of the principal symbol relative to the chosen orientation of local coordinates. We have identified three geometric objects encoded within the principal symbol — metric, Teleparallel connection and topological charge. Consequently, one would expect the coefficient b(x) from formula (3.2.6) to be expressed via these three geometric objects and the subprincipal symbol. This assertion is confirmed by the following theorem.

Theorem 4.1.1. The coefficients in the two-term asymptotics (3.2.6) are given by the formulae

$$a(x) = \frac{1}{6\pi^2} \sqrt{\det g_{\alpha\beta}(x)},$$
 (4.1.12)

$$b(x) = \frac{1}{8\pi^2} \left(\left[3 \,\mathbf{c} * T^{\mathrm{ax}} - 2 \,\mathrm{tr} \,A_{\mathrm{sub}} \right] \sqrt{\det g_{\alpha\beta}} \right)(x) \,, \tag{4.1.13}$$

where

$$T_{\alpha\beta\gamma}^{\rm ax} := \frac{1}{3} (T_{\alpha\beta\gamma} + T_{\gamma\alpha\beta} + T_{\beta\gamma\alpha}) \tag{4.1.14}$$

is axial torsion (totally antisymmetric piece of the torsion tensor) and * is the Hodge star (4.3.4).

Remark 4.1.1. The spectral and counting functions admit two-term asymptotic expansions (3.2.6) and (3.2.7) only under appropriate assumptions on geodesic loops and closed geodesics respectively, see Theorems 8.3 and 8.4 from [22]. However, one can easily reformulate asymptotic formulae (3.2.6) and (3.2.7) in such a way that they remain valid without assumptions on geodesics: this can easily be achieved, say, by taking a convolution with a function from Schwartz space $S(\mathbb{R})$, see Theorems 7.1 and 7.2 in [22]. Thus, the second asymptotic coefficients of the spectral and counting functions are well-defined irrespective of how many geodesic loops or closed geodesics we have. We introduced the second asymptotic coefficients b(x) and b via the unmollified asymptotic expansions (3.2.6) and (3.2.7) simply for the sake of clarity of presentation.

The proof of Theorem 4.1.1 is given in Sections 4.2–4.5.

We now turn our attention to the massless Dirac operator. This operator is defined in Appendix 4.A, see formula (4.A.3), and it does not fit into our scheme because it is an operator acting on a 2-component complex-valued spinor (Weyl spinor) rather than a pair of complex-valued half-densities. However, on a parallelizable manifold components of a spinor can be identified with half-densities. We call the resulting operator *the massless Dirac operator on half-densities*. The explicit formula for the massless Dirac operator on half-densities is (4.A.19).

The massless Dirac operator on half-densities is an operator of the type we are considering in Chapter 4, i.e. a self-adjoint first order elliptic differential operator acting on 2-columns of complex-valued half-densities and with a trace-free principal symbol. We address the question: is a given operator A a massless Dirac operator? The answer is given by the following theorem which the main result of Chapter 4.

Theorem 4.1.2. The operator A is a massless Dirac operator on half-densities if and only if the following two conditions are satisfied at every point of the manifold M:

- a) the subprincipal symbol of the operator, $A_{sub}(x)$, is proportional to the identity matrix, and
- b) the second asymptotic coefficient of the spectral function, b(x), is zero.

Note that conditions a) and b) in Theorem 4.1.2 are invariant under special unitary transformations, i.e. transformations of the operator

$$A \mapsto RAR^*, \tag{4.1.15}$$

where $R: M \to SU(2)$ is an arbitrary smooth special unitary matrix-function. The invariance of condition b) is obvious. In fact, condition b) is invariant under the action of a broader group: the unitary matrix-function R(x) appearing in formula (4.1.15) does not have to be special. As to condition a), its invariance is established by examination of formula (9.3) from [22] with the use of the special commutation properties of trace-free Hermitian 2×2 matrices. The fact that the conditions of Theorem 4.1.2 are SU(2) invariant is not surprising as the massless Dirac operator is designed around the concept of SU(2) invariance, see Property 4 in Appendix 4.A.

The proof of Theorem 4.1.2 is given in Sections 4.6 and 4.7.

Theorems 4.1.1 and 4.1.2 tell us that for the massless Dirac operator on halfdensities formulae (3.2.6) and (3.2.7) read

$$e(\lambda, x, x) = \frac{\sqrt{\det g_{\alpha\beta}(x)}}{6\pi^2} \lambda^3 + o(\lambda^2), \qquad (4.1.16)$$

$$N(\lambda) = \frac{\operatorname{Vol} M}{6\pi^2} \lambda^3 + o(\lambda^2), \qquad (4.1.17)$$

where $\operatorname{Vol} M$ is the volume of the Riemannian 3-manifold M.

Remark 4.1.2. The factor $\sqrt{\det g_{\alpha\beta}(x)}$ appears in the RHS of (4.1.16) because we are working with the massless Dirac operator on half-densities (4.A.19) rather than with the massless Dirac operator on spinors (4.A.3). For the massless Dirac operator on spinors the spectral function is a scalar field (as opposed to a density) and formula (4.1.16) reads $e(\lambda, x, x) = \frac{1}{6\pi^2}\lambda^3 + o(\lambda^2)$.

4.2 Reduction from the general setting

As explained in Section 4.1, the problem considered in Chapter 4 is a special case of that from [22] and Chapter 3. Formulae (3.2.21) and (3.2.22) from Chapter 3 in this case read

$$a(x) = \int_{h^+(x,\xi)<1} d\xi , \qquad (4.2.1)$$

$$b(x) = b_1(x) + b_2(x),$$
 (4.2.2)

where

$$b_1(x) = -3 \int_{h^+(x,\xi)<1} ([v^+]^* A_{\rm sub} v^+)(x,\xi) \,d\xi \,, \qquad (4.2.3)$$

$$b_2(x) = \frac{3i}{2} \int_{h^+(x,\xi)<1} \{ [v^+]^*, A_1 - 2h^+I, v^+ \} (x,\xi) \, d\xi \,. \tag{4.2.4}$$

Here $h^+(x,\xi)$ is the positive eigenvalue of the principal symbol (see also formula (4.1.8)), $v^+(x,\xi)$ is the corresponding normalized eigenvector (2-column), $d\xi$ is shorthand for $d\xi := (2\pi)^{-3} d\xi = (2\pi)^{-3} d\xi_1 d\xi_2 d\xi_3$ and I is the 2 × 2 identity matrix. Curly brackets in formula (4.1.13) denote the Poisson bracket on matrix-functions (3.2.14) and its further generalization (3.2.15) in three dimensions (rather than n dimensions as was seen in Chapter 3).

Put $P^+(x,\xi) := [v^+(x,\xi)][v^+(x,\xi)]^*$, which is the orthogonal projection onto the eigenspace span v^+ of the principal symbol. We have $A_1 - 2h^+I = 2h^+P^+ - 3h^+I$ and $\{[v^+]^*, P^+, v^+\} = 0$, so formula (4.2.4) can be rewritten as

$$b_2(x) = -\frac{9i}{2} \int_{h^+(x,\xi)<1} (h^+\{[v^+]^*, v^+\})(x,\xi) \,d\xi \,. \tag{4.2.5}$$

Our aim now is to evaluate the integrals (4.2.1), (4.2.3) and (4.2.5) explicitly. Formulae (4.2.1) and (4.1.8) immediately imply (4.1.12).

In order to evaluate the integral (4.2.3) we rewrite this formula as

$$b_1(x) = -3 \int_{h^+(x,\xi) < 1} \operatorname{tr}(A_{\mathrm{sub}}P^+)(x,\xi) \,d\xi$$

and use the fact that $P^+(x,\xi) = \frac{1}{2h^+(x,\xi)}(A_1(x,\xi) + h^+(x,\xi)I)$. We get

$$b_1(x) = -3 \int_{h^+(x,\xi) < 1} \frac{1}{2h^+(x,\xi)} \operatorname{tr}(A_{\operatorname{sub}}(A_1 + h^+ I))(x,\xi) \, d\xi$$

But A_{sub} does not depend on ξ whereas A_1 is an odd function of ξ , so the term $\frac{1}{2h^+} \operatorname{tr}(A_{\text{sub}}A_1)$ integrates to zero, leaving us with

$$b_1(x) = -\frac{3}{2} (\operatorname{tr} A_{\operatorname{sub}})(x) \int_{h^+(x,\xi) < 1} d\xi = -\frac{1}{4\pi^2} (\operatorname{tr} A_{\operatorname{sub}} \sqrt{\det g_{\alpha\beta}})(x) . \quad (4.2.6)$$

In order to complete the proof of Theorem 4.1.1 we need to evaluate explicitly the integral (4.2.5). The next three sections deal with this nontrivial issue.

4.3 Teleparallel connection

We show in this section that the principal symbol generates a Teleparallel connection which allows us to reformulate the results of our spectral analysis in a much clearer geometric language.

Let us show first that the existence of a principal symbol implies that our manifold M is parallelizable. The principal symbol $A_1(x,\xi)$ is linear in ξ so it can be written as

$$A_1(x,\xi) = \sigma^{\alpha}(x)\,\xi_{\alpha}\,,\tag{4.3.1}$$

where $\sigma^{\alpha}(x)$, $\alpha = 1, 2, 3$, are some trace-free Hermitian 2 × 2 matrix-functions. Let us denote the elements of the matrices σ^{α} as $\sigma^{\alpha}{}_{\dot{a}b}$, where the dotted index, running through the values $\dot{1}, \dot{2}$, enumerates the rows and the undotted index, running through the values 1, 2, enumerates the columns; this notation is taken from [23]. Put

$$e_1^{\ \alpha}(x) := \operatorname{Re} \sigma^{\alpha}{}_{12}(x), \quad e_2^{\ \alpha}(x) := -\operatorname{Im} \sigma^{\alpha}{}_{12}(x), \quad e_3^{\ \alpha}(x) := \sigma^{\alpha}{}_{11}(x). \quad (4.3.2)$$

Formula (4.3.2) defines a triple of smooth real vector fields $e_j(x)$, j = 1, 2, 3, on the manifold M. These vector fields are linearly independent at every point xof the manifold: this follows from formula (4.1.2) via routine calculations. Thus, the triple of vector fields e_j is a *frame*. The existence of a frame means that the manifold M is parallelizable.

Conversely, given a frame e_j we uniquely recover the principal symbol $A_1(x, \xi)$ via formulae (4.3.1), (4.A.1) and (4.A.2). Thus, a principal symbol is equivalent to a frame. Of course, this equivalence statement relies on our *a priori* assumptions (4.1.1), (4.1.2) and (4.1.3)–(4.1.6).
It is easy to see that the frame elements e_j are orthonormal with respect to the metric (4.1.7). Moreover, the metric can be determined directly from the frame as

$$g^{\alpha\beta} = \delta^{jk} e_j^{\ \alpha} e_k^{\ \beta} \,, \tag{4.3.3}$$

where the repeated frame indices j and k indicate summation over j, k = 1, 2, 3. The two definitions of the metric, (4.1.7) and (4.3.3), are equivalent.

Parallelizability implies orientability, see Proposition 13.5 in [52]. Having chosen a particular orientation, we allow only changes of local coordinates x^{α} , $\alpha = 1, 2, 3$, which preserve orientation and define the Hodge star in the standard way: the action of * on a rank q antisymmetric tensor Q is

$$(*Q)_{\gamma_{q+1}\dots\gamma_3} := (q!)^{-1} \sqrt{\det g_{\alpha\beta}} Q^{\gamma_1\dots\gamma_q} \varepsilon_{\gamma_1\dots\gamma_3}, \qquad (4.3.4)$$

where ε is the totally antisymmetric quantity, $\varepsilon_{123} := +1$, and g is the Riemannian metric (4.1.7). Here and further on we identify differential forms with covariant antisymmetric tensors. We raise and lower tensor indices using our metric.

Substituting formulae (4.3.1) and (4.3.2) into (4.1.11) we get

$$\mathbf{c} = \operatorname{sgn} \det e_j{}^{\alpha}. \tag{4.3.5}$$

Formula (4.3.5) provides an equivalent (and more natural) definition of topological charge. It also explains why the topological charge, initially defined in Section 4.1 in accordance with formula (4.1.11), can only take values +1 or -1.

The concept of a Teleparallel connection was already defined in Section 4.1 in accordance with formula (4.1.9). This connection can be equivalently defined via the frame as follows. Suppose we have a vector v based at the point $x \in M$ and we want to construct a parallel vector \tilde{v} based at the point $\tilde{x} \in M$. We decompose the vector v with respect to the frame at the point $x, v = c^j e_j(x)$, and reassemble it with the same coefficients c^j at the point \tilde{x} , defining $\tilde{v} := c^j e_j(\tilde{x})$. We now define the covariant derivative corresponding to the Teleparallel connection. Our Teleparallel connection is a special case of an affine connection, so we are looking at a covariant derivative acting on vector/covector fields in the usual manner

$$\nabla_{\mu}v^{\alpha} = \partial v^{\alpha}/\partial x^{\mu} + \Gamma^{\alpha}{}_{\mu\beta}v^{\beta}, \qquad \nabla_{\mu}w_{\beta} = \partial w_{\beta}/\partial x^{\mu} - \Gamma^{\alpha}{}_{\mu\beta}w_{\alpha}.$$

Note that in this thesis we use the symbol ∇ to denote two different covariant derivatives: one with respect to the Levi-Civita connection, and the other with respect to the Teleparallel connection. The connection with respect to which the covariant derivative is defined is always stated. As such, the meaning is clear from the context in which the symbol is used.

The Teleparallel connection coefficients are defined from the conditions

$$\nabla_{\mu} e_j{}^{\alpha} = 0, \qquad (4.3.6)$$

where the e_j are elements of our frame. Formula (4.3.6) gives a system of 27 linear algebraic equations for the determination of 27 unknown connection coefficients. It is known (see, for example, formula (A2) in [15]), that the unique solution of this system is

$$\Gamma^{\alpha}{}_{\mu\beta} = e_k{}^{\alpha} (\partial e^k{}_{\beta} / \partial x^{\mu}), \qquad (4.3.7)$$

where

$$e^k{}_\beta := \delta^{kj} g_{\beta\gamma} e_j{}^\gamma. \tag{4.3.8}$$

The triple of covector fields e^k , k = 1, 2, 3, is called the *coframe*. The frame and coframe uniquely determine each other via the relation

$$e_j{}^{\alpha}e^k{}_{\alpha} = \delta_j{}^k. \tag{4.3.9}$$

One can check by performing explicit calculations that the Teleparallel connection has the following two important properties. Firstly

$$\nabla_{\alpha}g^{\beta\gamma} = 0. \tag{4.3.10}$$

This follows from (4.3.3) which gives

$$\nabla_{\gamma} g^{\alpha\beta} = \nabla_{\gamma} \left(\delta^{jk} e_{j}^{\alpha} e_{k}^{\beta} \right)$$
$$= \delta^{jk} \left(\nabla_{\gamma} e_{j}^{\alpha} \right) e_{k}^{\beta} + \delta^{jk} e_{j}^{\alpha} \left(\nabla_{\gamma} e_{k}^{\beta} \right)$$
$$= 0$$

from condition (4.3.6). This means that the connection is metric compatible. Secondly

$$(\nabla_{\alpha}\nabla_{\beta} - \nabla_{\beta}\nabla_{\alpha})v^{\gamma} = 0 \text{ for any vector field } v, \qquad (4.3.11)$$

which means that the Riemann curvature tensor is zero; this follows from a lengthy calculation which is omitted here for brevity. Properties (4.3.10) and (4.3.11) are the defining properties of a Teleparallel connection: a Teleparallel connection is, by definition [39], an affine connection satisfying (4.3.10) and (4.3.11).

The tensor characterizing the "strength" of the Teleparallel connection is not the Riemann curvature tensor but the torsion tensor (4.1.10). The Teleparallel connection is, in a sense, the opposite of the more common Levi-Civita connection: the Levi-Civita connection has zero torsion but nonzero curvature, whereas the Teleparallel connection has nonzero torsion but zero curvature. In Chapter 4 we distinguish these two affine connections by using different notation for connection coefficients: we write the Teleparallel connection coefficients as $\Gamma^{\alpha}{}_{\beta\gamma}$ and the Levi-Civita connection coefficients (Christoffel symbols) as $\left\{ {\alpha \atop \beta\gamma} \right\}$, see formula (4.A.4). It is known, see formula (7.34) in [53], that the two sets of connection coefficients are related as

$$\Gamma^{\alpha}{}_{\beta\gamma} = \left\{ \begin{array}{c} \alpha\\ \beta\gamma \end{array} \right\} + \frac{1}{2} (T^{\alpha}{}_{\beta\gamma} + T_{\beta}{}^{\alpha}{}_{\gamma} + T_{\gamma}{}^{\alpha}{}_{\beta}).$$
(4.3.12)

Note that in (4.3.12) the position of indices of the torsion tensor is extremely important. The indices are raised and lowered using the metric. For absolute clarity we relate the torsion tensor appearing in (4.3.12) to (4.1.10) in the following manner:

$$T_{\beta}{}^{\alpha}{}_{\gamma} = T^{\mu}{}_{\nu\gamma} g_{\mu\beta} g^{\nu\alpha},$$

see Chapter 7 of [53] for further details.

Substituting (4.3.7) into (4.1.10) we arrive at the following explicit formula for the torsion tensor of the Teleparallel connection

$$T = e_j \otimes de^j \,, \tag{4.3.13}$$

where the d stands for the exterior derivative. For the sake of clarity we rewrite formula (4.3.13) in more detailed form, retaining all tensor indices,

$$T^{\alpha}{}_{\beta\gamma} = e_j{}^{\alpha} (\partial e^j{}_{\gamma} / \partial x^{\beta} - \partial e^j{}_{\beta} / \partial x^{\gamma}) \,. \tag{4.3.14}$$

As always, the repeated index j appearing in formulae (4.3.13) and (4.3.14) indicates summation over j = 1, 2, 3.

Torsion is a rank three tensor antisymmetric in the last two indices. Because we are working in dimension three, it is convenient, as in Chapter 2, to apply the Hodge star in the last two indices and deal with the rank two tensor

$${}^{*}_{\beta} := \frac{1}{2} T^{\alpha\gamma\delta} \varepsilon_{\gamma\delta\beta} \sqrt{\det g_{\mu\nu}}$$
(4.3.15)

instead. Note that in (4.3.15) the position of the indices of the torsion tensor is extremely important. Recalling again that indices are raised and lowered using the metric, we see that the torsion tensor appearing in (4.3.15) is related to (4.1.10) in accordance with:

$$T^{\alpha\gamma\delta} = T^{\alpha}{}_{\mu\nu} \, g^{\mu\gamma} \, g^{\nu\delta}$$

Substituting (4.3.13) into (4.3.15) we get

$$\overset{*}{T} = e_j \otimes \operatorname{curl} e^j \,, \tag{4.3.16}$$

where

$$(\operatorname{curl} e^j)_{\beta} := (*de^j)_{\beta} = \frac{1}{2} \, (de^j)^{\gamma \delta} \, \varepsilon_{\gamma \delta \beta} \, \sqrt{\det g_{\mu \nu}} \,. \tag{4.3.17}$$

4.4 Relation between curvature of the U(1) connection and torsion of the Teleparallel connection

This section is devoted to the examination of the integrand in formula (4.2.5). Recall that the curly brackets in this integrand denote the Poisson bracket on matrix-functions (3.2.14) in three dimensions.

As explained in Section 3.4, the expression $-i\{[v^+]^*, v^+\}$ is the scalar curvature of the U(1) connection generated by the eigenspace span v^+ of the principal symbol. This curvature term appears in the general setting of a first order elliptic system. A feature of the particular case (4.1.3)–(4.1.6) considered in Chapter 4 is that the scalar curvature of the U(1) connection can be expressed via torsion of the Teleparallel connection. This is a substantial simplification. The Teleparallel connection is a simpler geometric object than the U(1) connection because the coefficients of the Teleparallel connection do not depend on the dual variable (momentum), i.e. they are "functions" on the base manifold M. The relationship between the two connections is established by the following lemma.

Lemma 4.4.1. The scalar curvature of the U(1) connection is expressed via the torsion of the Teleparallel connection, metric and topological charge as

$$-i\{[v^+]^*, v^+\}(x,\xi) = \frac{\mathbf{c}}{2} \frac{\overset{*}{T}{}^{\alpha\beta}(x)\,\xi_{\alpha}\xi_{\beta}}{(g^{\mu\nu}(x)\,\xi_{\mu}\xi_{\nu})^{3/2}}\,.$$
(4.4.1)

Recall that the topological charge $\mathbf{c} = \pm 1$ is defined in accordance with formula (4.1.11) or, equivalently, in accordance with formula (4.3.5).

Proof of Lemma 4.4.1 We give the proof for the case

$$\mathbf{c} = +1. \tag{4.4.2}$$

There is no need to give a separate proof for the case $\mathbf{c} = -1$ as the two cases reduce to one another by means of a) the observation that torsion (4.3.13) is invariant under inversion of the frame and b) the identity

$$\{[v^+]^*, v^+\} + \{[v^-]^*, v^-\} = 0, \qquad (4.4.3)$$

where $v^{-}(x,\xi)$ is the normalized eigenvector of the principal symbol corresponding to the negative eigenvalue. Formula (4.4.3) is a special case of formula (3.2.20) from Chapter 3.

We fix an arbitrary point $Q \in T'M$ and prove formula (4.4.1) at this point. As the LHS and RHS of (4.4.1) are invariant under changes of local coordinates x, it is sufficient to prove formula (4.4.1) in Riemann normal coordinates, i.e. local coordinates such that x = 0 corresponds to the projection of the point Q onto the base manifold, $g_{\mu\nu}(0) = \delta_{\mu\nu}$ and $\frac{\partial g_{\mu\nu}}{\partial x^{\lambda}}(0) = 0$. Moreover, as the formula we are proving involves only first partial derivatives in x, we may assume, without loss of generality, that

$$g_{\mu\nu}(x) = \delta_{\mu\nu} \tag{4.4.4}$$

for all x in some neighbourhood of the origin. In other words, it is sufficient to prove formula (4.4.1) for the case of Euclidean metric.

As both the LHS and RHS of (4.4.1) have the same degree of homogeneity in ξ , namely, -1, it is sufficient to prove formula (4.4.1) for ξ of norm 1. Moreover, by rotating our Cartesian coordinate system we can reduce the case of general ξ of norm 1 to the case

$$\xi = \begin{pmatrix} 0 & 0 & 1 \end{pmatrix}^T. \tag{4.4.5}$$

There is one further simplification that can be made: we claim that it is sufficient to prove formula (4.4.1) for the case when

$$e_j^{\ \alpha}(0) = \delta_j^{\ \alpha},\tag{4.4.6}$$

i.e. for the case when at the point x = 0 the elements of the frame are aligned with the coordinate axes. This claim follows from the observation that the LHS of formula (4.4.1) is invariant under rigid special unitary transformations of the column-function v^+ , $v^+ \mapsto Rv^+$, where "rigid" refers to the fact that the matrix $R \in SU(2)$ is constant. Of course, the column-function Rv^+ is no longer an eigenvector of the original principal symbol, but a new principal symbol obtained from the old one by the rigid special orthogonal transformation of the frame (4.A.14) with the 3×3 special orthogonal matrix O expressed in terms of the 2×2 special unitary matrix R in accordance with (4.A.15). One can always choose the special unitary matrix R so that at the point x = 0 the elements of the new frame are aligned with the coordinate axes (in fact, there are two possible choices of R which differ by sign). It remains only to note that direct inspection of formula (4.3.13) shows that torsion is also invariant under rigid special orthogonal transformations of the frame, and, hence, the tensor $\stackrel{*}{T}$ defined by formula (4.3.15) and appearing in the RHS of formula (4.4.1) is invariant under rigid special orthogonal transformations of the frame as well.

Having made the simplifying assumptions (4.4.4)-(4.4.6), we are now in a position to prove formula (4.4.1).

Let us calculate the RHS of (4.4.1) first. In view of (4.4.6) we have, in the linear approximation in x,

$$\begin{pmatrix} e_1^{\ 1}(x) & e_1^{\ 2}(x) & e_1^{\ 3}(x) \\ e_2^{\ 1}(x) & e_2^{\ 2}(x) & e_2^{\ 3}(x) \\ e_3^{\ 1}(x) & e_3^{\ 2}(x) & e_3^{\ 3}(x) \end{pmatrix} = \begin{pmatrix} 1 & w^3(x) & -w^2(x) \\ -w^3(x) & 1 & w^1(x) \\ w^2(x) & -w^1(x) & 1 \end{pmatrix}, \quad (4.4.7)$$

where w is some smooth vector-function which vanishes at x = 0. Formula (4.4.7)

is the standard formula for the linearization of an orthogonal matrix about the identity; see also formula (2.10.1) in Chapter 2. Note that in Cosserat elasticity literature the vector-function w is called the *vector of microrotations*. Substituting (4.4.7) into (4.3.16) and (4.3.17) we get, at x = 0,

$${}^{*}_{\alpha\beta} = \partial w_{\beta} / \partial x^{\alpha} - \delta_{\alpha\beta} \operatorname{div} w, \qquad (4.4.8)$$

which is formula (2.10.5) from Chapter 2. Here we freely lower and raise tensor indices using the fact that the metric is Euclidean (in the Euclidean case (4.4.4) it does not matter whether a tensor index comes as a subscript or a superscript). Substituting (4.4.8) and (4.4.5) into the RHS of (4.4.1) we get, at our point $Q \in T'M$,

$$\frac{1}{2} \frac{\tilde{T}^{\alpha\beta} \xi_{\alpha} \xi_{\beta}}{(g^{\mu\nu} \xi_{\mu} \xi_{\nu})^{3/2}} = -\frac{1}{2} (\partial w^1 / \partial x^1 + \partial w^2 / \partial x^2).$$
(4.4.9)

Let us now calculate the LHS of (4.4.1). The equation for the eigenvector $v^+(x,\xi)$ of the principal symbol is

$$\begin{pmatrix} e_3{}^{\alpha}\xi_{\alpha} - \|\xi\| & (e_1 - ie_2){}^{\alpha}\xi_{\alpha} \\ (e_1 + ie_2){}^{\alpha}\xi_{\alpha} & -e_3{}^{\alpha}\xi_{\alpha} - \|\xi\| \end{pmatrix} \begin{pmatrix} v_1^+ \\ v_2^+ \end{pmatrix} = 0.$$
(4.4.10)

In view of (4.4.5) and (4.4.6) the (normalized) solution of (4.4.10) at our point $Q \in T'M$ is $v^+ = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$. Of course, our $v^+(x,\xi)$ is defined up to the gauge transformation

$$v^+ \mapsto e^{i\phi^+}v^+, \tag{4.4.11}$$

where

$$\phi^+: T'M \to \mathbb{R} \tag{4.4.12}$$

is an arbitrary smooth function, however the LHS of (4.4.1) is invariant under this gauge transformation. We now perturb equation (4.4.10) about the point $Q \in T'M$, that is, about x = 0, $\xi = \begin{pmatrix} 0 & 0 & 1 \end{pmatrix}$, making use of formula (4.4.7), which gives us the following equation for the increment δv^+ of the eigenvector $v^+(x,\xi)$ of the principal symbol:

$$\begin{pmatrix} 0 & 0 \\ 0 & -2 \end{pmatrix} \begin{pmatrix} \delta v_1^+ \\ \delta v_2^+ \end{pmatrix} + \begin{pmatrix} 0 & -w^2(x) - iw^1(x) \\ -w^2(x) + iw^1(x) & 0 \end{pmatrix} \begin{pmatrix} 1 \\ 0 \end{pmatrix} + \begin{pmatrix} 0 & \delta\xi_1 - i\delta\xi_2 \\ \delta\xi_1 + i\delta\xi_2 & -2\delta\xi_3 \end{pmatrix} \begin{pmatrix} 1 \\ 0 \end{pmatrix} = 0,$$

or, equivalently,

$$\delta v_2^+ = \frac{1}{2}(-w^2(x) + iw^1(x) + \delta\xi_1 + i\delta\xi_2). \tag{4.4.13}$$

Formula (4.4.13) has to be supplemented by the normalization condition $||v^+(x,\xi)|| = 1$, which in its linearized form reads

$$\operatorname{Re}\delta v_1^+ = 0. \tag{4.4.14}$$

Formulae (4.4.14) and (4.4.13) define δv^+ modulo an arbitrary Im δv_1^+ , with this degree of freedom being associated with the gauge transformation (4.4.11), (4.4.12). Without loss of generality we may assume that the gauge is chosen so that

$$Im \,\delta v_1^+ = 0. \tag{4.4.15}$$

Combining formulae (4.4.14), (4.4.15) and (4.4.13) we get

$$\delta v^{+} = \frac{1}{2} \begin{pmatrix} 0 \\ -w^{2}(x) + iw^{1}(x) + \delta\xi_{1} + i\delta\xi_{2} \end{pmatrix}.$$
 (4.4.16)

Recall that the w appearing in this formula is some smooth vector-function which vanishes at x = 0.

Differentiation of (4.4.16) gives us

$$\frac{\partial v^+}{\partial x^{\alpha}} = \frac{1}{2} \begin{pmatrix} 0 \\ -\partial w^2 / \partial x^{\alpha} + i \partial w^1 / \partial x^{\alpha} \end{pmatrix}, \qquad (4.4.17)$$

$$\frac{\partial v^+}{\partial \xi_1} = \frac{1}{2} \begin{pmatrix} 0\\1 \end{pmatrix}, \qquad \frac{\partial v^+}{\partial \xi_2} = \frac{1}{2} \begin{pmatrix} 0\\i \end{pmatrix}, \qquad \frac{\partial v^+}{\partial \xi_3} = 0.$$
(4.4.18)

Formulae (4.4.17) and (4.4.18) imply that at our point $Q \in T'M$

$$-i\{[v^{+}]^{*}, v^{+}\} = -\frac{1}{2}(\partial w^{1}/\partial x^{1} + \partial w^{2}/\partial x^{2}).$$
(4.4.19)

Comparing formulae (4.4.9) and (4.4.19) and recalling (4.4.2), we arrive at the required result (4.4.1). \Box

4.5 Integration of the curvature term

Substituting (4.4.1) into (4.2.5) we get

$$b_2(x) = \frac{9\mathbf{c}}{4} \int_{h^+(x,\xi)<1} \frac{\tilde{T}^{\alpha\beta}(x)\,\xi_\alpha\xi_\beta}{g^{\mu\nu}(x)\,\xi_\mu\xi_\nu}\,d\xi\,.$$
(4.5.1)

Recall that $h^+(x,\xi)$ is given by formula (4.1.8).

The tensor $\stackrel{*}{T}$ can be decomposed into pure trace and trace-free pieces, i.e.

$${}^{*}T^{\alpha\beta} = \frac{1}{3}g^{\alpha\beta}T^{\gamma}{}_{\gamma} + \left(T^{\alpha\beta} - \frac{1}{3}g^{\alpha\beta}T^{\gamma}{}_{\gamma}\right).$$
(4.5.2)

It is easy to see that the trace-free piece (second term in the RHS of (4.5.2)) does not contribute to the integral in (4.5.1), hence formula (4.5.1) becomes

$$b_2(x) = \frac{3\mathbf{c}}{4} {}^*{}^{\gamma}{}_{\gamma}(x) \int_{h^+(x,\xi)<1} d\xi = \frac{\mathbf{c}}{8\pi^2} \left({}^*{}^{\gamma}{}_{\gamma} \sqrt{\det g_{\alpha\beta}} \right)(x) \,. \tag{4.5.3}$$

But formulae (4.1.14), (4.3.4) and (4.3.15) imply that

$${}^{*}T^{\gamma}{}_{\gamma} = 3 * T^{\mathrm{ax}}.$$
 (4.5.4)

Combining formulae (4.2.2), (4.2.6), (4.5.3) and (4.5.4) we arrive at formula (4.1.13). This completes the proof of Theorem 4.1.1.

4.6 The subprincipal symbol of the massless Dirac operator

In this section we calculate the subprincipal symbol of the massless Dirac operator, which prepares the ground for the proof of Theorem 4.1.2 in the next section. In view of Remark 2.1.10 from [72], defining the subprincipal symbol for the massless Dirac operator on spinors (4.A.3) is problematic, hence, we work with the massless Dirac operator on half-densities (4.A.19). For the sake of brevity we denote the massless Dirac operator on half-densities by A rather than by $W_{1/2}$.

Lemma 4.6.1. The subprincipal symbol of the massless Dirac operator on halfdensities (4.A.19) is

$$A_{\rm sub}(x) = \frac{3\mathbf{c}}{4} \left(*T^{\rm ax}(x) \right) I , \qquad (4.6.1)$$

where $\mathbf{c} = \pm 1$ is the topological charge (4.3.5), T^{ax} is axial torsion (4.1.14), * is the Hodge star (4.3.4) and I is the 2 × 2 identity matrix.

Proof We give the proof of (4.6.1) for the case (4.4.2). There is no need to give a separate proof for the case $\mathbf{c} = -1$ as the two cases reduce to one another by inversion of the frame: the full symbol of the massless Dirac operator on halfdensities changes sign under inversion of the frame and hence its subprincipal symbol changes sign under inversion of the frame, whereas torsion (4.3.13) is invariant under inversion of the frame.

According to formula (3.1.15) from Chapter 3 the subprincipal symbol is defined as

$$A_{\rm sub} := A_0 + \frac{i}{2} (A_1)_{x^{\alpha} \xi_{\alpha}} , \qquad (4.6.2)$$

where $A_1(x,\xi)$ and $A_0(x)$ are the homogeneous (in ξ) components of the full symbol $A(x,\xi) = A_1(x,\xi) + A_0(x)$ of our first order differential operator, with the subscript indicating degree of homogeneity. For the massless Dirac operator on half-densities (4.A.19) these homogeneous components read (4.3.1) and

$$A_0(x) = -\frac{i}{4}\sigma^{\alpha}\sigma_{\beta}\left(\frac{\partial\sigma^{\beta}}{\partial x^{\alpha}} + \left\{\frac{\beta}{\alpha\gamma}\right\}\sigma^{\gamma}\right) + \frac{i}{2}\sigma^{\alpha}\left\{\frac{\beta}{\alpha\beta}\right\}$$
(4.6.3)

respectively. Note that in writing down (4.6.3) we used the standard formula

$$\frac{1}{2\det g_{\kappa\lambda}}\frac{\partial\det g_{\mu\nu}}{\partial x^{\alpha}} = \left\{ \begin{array}{c} \beta\\ \alpha\beta \end{array} \right\}$$

Our task is to substitute (4.3.1) and (4.6.3) into (4.6.2).

We fix an arbitrary point $P \in M$ and prove formula (4.6.1) at this point. As the LHS and RHS of (4.6.1) are invariant under changes of local coordinates x, it is sufficient to check the identity (4.6.1) in Riemann normal coordinates, i.e. local coordinates such that x = 0 corresponds to the point P, $g_{\mu\nu}(0) = \delta_{\mu\nu}$ and $\frac{\partial g_{\mu\nu}}{\partial x^{\lambda}}(0) = 0$. Moreover, as the principal symbol is linear in ξ and the formula we are proving involves only first partial derivatives in x, we may assume, without loss of generality, that we have (4.4.4) for all x in some neighbourhood of the origin. In other words, it is sufficient to prove formula (4.6.1) for the case of Euclidean metric. Furthermore, by rotating our Cartesian coordinate system we can achieve (4.4.6), which opens the way to the use, in the linear approximation in x, of formula (4.4.7).

Substituting (4.4.7) into (4.A.1), we get, in the linear approximation in x,

$$\sigma^{1} = \begin{pmatrix} w^{2} & 1 + iw^{3} \\ 1 - iw^{3} & -w^{2} \end{pmatrix} = \sigma_{1},$$

$$\sigma^{2} = \begin{pmatrix} -w^{1} & -i + w^{3} \\ i + w^{3} & w^{1} \end{pmatrix} = \sigma_{2},$$

$$\sigma^{3} = \begin{pmatrix} 1 & -iw^{1} - w^{2} \\ iw^{1} - w^{2} & -1 \end{pmatrix} = \sigma_{3}. \quad (4.6.4)$$

Substitution of (4.6.4) into (4.3.1) and (4.6.3) gives us

$$A_{1}(x,\xi) = \begin{pmatrix} \xi_{3} & \xi_{1} - i\xi_{2} \\ \xi_{1} + i\xi_{2} & -\xi_{3} \end{pmatrix} + \begin{pmatrix} w^{2}\xi_{1} - w^{1}\xi_{2} & iw^{3}\xi_{1} + w^{3}\xi_{2} + (-iw^{1} - w^{2})\xi_{3} \\ -iw^{3}\xi_{1} + w^{3}\xi_{2} + (iw^{1} - w^{2})\xi_{3} & -w^{2}\xi_{1} + w^{1}\xi_{2} \end{pmatrix}, \quad (4.6.5)$$

$$A_0(0) = -\frac{i}{4} \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} \partial w^2 / \partial x^1 & i \partial w^3 / \partial x^1 \\ -i \partial w^3 / \partial x^1 & -\partial w^2 / \partial x^1 \end{pmatrix} + \dots$$
(4.6.6)

Here formula (4.6.5) is written in the linear approximation in x, whereas formula (4.6.6) displays, for the sake of brevity, only one term out of nine (the one corresponding to $\alpha = \beta = 1$ in (4.6.3)), with the remaining eight terms concealed within the dots Note also that the Christoffel symbols disappeared because of our assumption that the metric is Euclidean.

Substituting (4.6.6) and (4.6.5) into (4.6.2), we get

$$A_{\rm sub}(0) = -\frac{1}{2} \,(\operatorname{div} w) \,I. \tag{4.6.7}$$

But, according to (4.4.8),

$$\tilde{T}^{\gamma}{}_{\gamma}(0) = -2\operatorname{div} w. \tag{4.6.8}$$

Formulae (4.6.7), (4.6.8), (4.5.4) and (4.4.2) imply formula (4.6.1) at x = 0.

4.7 Proof of Theorem 4.1.2

As Theorem 4.1.2 is an if and only if theorem, our proof comes in two parts.

Part 1 of the proof Let A be a massless Dirac operator on half-densities. We need to prove that a) the subprincipal symbol of this operator, $A_{sub}(x)$, is proportional to the identity matrix and b) the second asymptotic coefficient of the spectral function, b(x), is zero. The required result follows from Lemma 4.6.1 and Theorem 4.1.1.

Part 2 of the proof Let A be a differential operator such that a) the subprincipal symbol of this operator, $A_{sub}(x)$, is proportional to the identity matrix and b) the second asymptotic coefficient of the spectral function, b(x), is zero. We need to prove that A is a massless Dirac operator on half-densities.

Theorem 4.1.1 implies that the subprincipal symbol of our operator A is given by formula (4.6.1). Let e_j be the frame corresponding to the principal symbol of the operator A, see formulae (4.3.1) and (4.3.2). Now, let B be the massless Dirac operator on half-densities corresponding to the same frame. Then the principal symbols of the operators A and B coincide. But Lemma 4.6.1 implies that the subprincipal symbols of the operators A and B coincide as well. A first order differential operator is determined by its principal and subprincipal symbols, hence, A = B. \Box

4.8 Explicit formula for axial torsion

Torsion is a rank three tensor antisymmetric in the last two indices. It is known, see Subsection 2.4.2 from Chapter 2, that torsion has three irreducible pieces. Only one of the three irreducible pieces of torsion, namely, the piece which theoretical physicists label by the adjective "axial", appears in our spectral theoretic results, see Theorem 4.1.1 and Lemma 4.6.1. It is also interesting that axial torsion is the irreducible piece which is used when one models the massless neutrino [23] or the electron [16] by means of Cosserat elasticity.

Axial torsion is defined as the totally antisymmetric piece of the torsion tensor, see formula (4.1.14). This means that axial torsion is a 3-form. In view of the

importance of axial torsion, we give an explicit formula for its Hodge dual in terms of the principal symbol $A_1(x,\xi)$. Formulae (4.3.16), (4.3.17) and (4.5.4) imply

$$*T^{ax} = \frac{\delta_{kl}}{3}\sqrt{\det g^{\alpha\beta}} \left[e^{k}_{1}\partial e^{l}_{3}/\partial x^{2} + e^{k}_{2}\partial e^{l}_{1}/\partial x^{3} + e^{k}_{3}\partial e^{l}_{2}/\partial x^{1} - e^{k}_{1}\partial e^{l}_{2}/\partial x^{3} - e^{k}_{2}\partial e^{l}_{3}/\partial x^{1} - e^{k}_{3}\partial e^{l}_{1}/\partial x^{2}\right]. \quad (4.8.1)$$

Here the coframe e^k is determined from the principal symbol in accordance with formulae (4.3.1), (4.3.2) and (4.3.9), whereas the contravariant metric tensor $g^{\alpha\beta}$ is determined from the principal symbol in accordance with formula (4.1.7).

4.A The massless Dirac operator

Let M be a 3-dimensional connected compact oriented manifold equipped with a Riemannian metric $g_{\alpha\beta}$, $\alpha, \beta = 1, 2, 3$ being the tensor indices. Note that we are more prescriptive in this appendix than in the main text of Chapter 4: in the main text orientability emerged as a consequence of the existence of a principal symbol and the metric was defined via the principal symbol, whereas in this appendix orientability and metric are introduced *a priori*.

We work only in local coordinates with prescribed orientation.

It is known [76, 48] that a 3-dimensional oriented manifold is parallelizable, i.e. there exist smooth real vector fields e_j , j = 1, 2, 3, that are linearly independent at every point x of the manifold. (This fact is often referred to as *Steenrod's theorem.*) Each vector $e_j(x)$ has coordinate components $e_j^{\alpha}(x)$, $\alpha = 1, 2, 3$. Note that we use the Latin letter j for enumerating the vector fields (this is an *anholonomic* or *frame* index) and the Greek letter α for enumerating their components (this is a *holonomic* or *tensor* index). The triple of linearly independent vector fields e_j , j = 1, 2, 3, is called a *frame*. Without loss of generality we assume further on that the vector fields e_j are orthonormal with respect to our metric: this can always be achieved by means of the Gram–Schmidt process.

Define Pauli matrices

$$\sigma^{\alpha}(x) := s^{j} e_{j}^{\alpha}(x), \qquad (4.A.1)$$

where

$$s^{1} := \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} = s_{1}, \quad s^{2} := \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix} = s_{2}, \quad s^{3} := \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} = s_{3}. \quad (4.A.2)$$

In formula (4.A.1) summation is carried out over the repeated frame index j = 1, 2, 3, and $\alpha = 1, 2, 3$ is the free tensor index.

The massless Dirac operator is the matrix operator

$$W := -i\sigma^{\alpha} \left(\frac{\partial}{\partial x^{\alpha}} + \frac{1}{4}\sigma_{\beta} \left(\frac{\partial\sigma^{\beta}}{\partial x^{\alpha}} + \left\{ \begin{array}{c} \beta \\ \alpha\gamma \end{array} \right\} \sigma^{\gamma} \right) \right), \qquad (4.A.3)$$

where summation is carried out over $\alpha, \beta, \gamma = 1, 2, 3$, and

$$\begin{cases} \beta \\ \alpha \gamma \end{cases} := \frac{1}{2} g^{\beta \delta} \left(\frac{\partial g_{\gamma \delta}}{\partial x^{\alpha}} + \frac{\partial g_{\alpha \delta}}{\partial x^{\gamma}} - \frac{\partial g_{\alpha \gamma}}{\partial x^{\delta}} \right)$$
(4.A.4)

are the Christoffel symbols. Here and throughout this appendix we raise and lower tensor indices using the metric. Note that we chose the letter "W" for denoting the massless Dirac operator because in theoretical physics literature it is often referred to as the *Weyl* operator.

Formula (4.A.3) is the formula from [23], only written in matrix notation (i.e. without spinor indices). Note that in the process of transcribing formulae from [23] into matrix notation we used the identity

$$\epsilon \sigma^{\alpha} \epsilon = (\sigma^{\alpha})^T, \qquad (4.A.5)$$

 $\alpha = 1, 2, 3$, where

$$\epsilon := \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} \tag{4.A.6}$$

is the "metric spinor". The identity (4.A.5) gives a simple way of raising/lowering spinor indices in Pauli matrices in the non-relativistic ($\alpha \neq 0$) setting.

Our definition (4.A.3) of the massless Dirac operator is a special case of the definition from [28]. The two definitions coincide when we work with a Spin connection as opposed to a Spin^c connection, see Propositions 2.14 and 2.15 in [28] for details.

Throughout Chapter 4 we work in dimension 3. The definition of the massless Dirac operator acting over a Riemannian manifold of arbitrary dimension can be found, for example, in [33, 30, 32].

Physically, our massless Dirac operator (4.A.3) describes a single massless neutrino living in a 3-dimensional compact universe M. The eigenvalues of the massless Dirac operator are the energy levels.

The massless Dirac operator (4.A.3) acts on 2-columns $v = \begin{pmatrix} v_1 & v_2 \end{pmatrix}^T$ of complexvalued scalar functions. In differential geometry this object is referred to as a (Weyl) spinor so as to emphasize the fact that v transforms in a particular way under transformations of the orthonormal frame e_j . However, as in our exposition the frame e_j is assumed to be chosen *a priori*, we can treat the components of the spinor as scalars. This issue will be revisited below when we state Property 4 of the massless Dirac operator.

We now list the main properties of the massless Dirac operator. We state these without proofs. The proofs can be found in Appendix 3.A of [20] or in [28].

Property 1. The massless Dirac operator is invariant under changes of local coordinates x, i.e. it maps 2-columns of smooth scalar functions $M \to \mathbb{C}^2$ to 2-columns of smooth scalar functions $M \to \mathbb{C}^2$ regardless of the choice of local coordinates.

Property 2. The massless Dirac operator is formally self-adjoint (symmetric) with respect to the inner product

$$\int_{M} w^* v \sqrt{\det g_{\alpha\beta}} \, dx \tag{4.A.7}$$

on 2-columns of smooth scalar functions $v, w : M \to \mathbb{C}^2$.

Property 3. The massless Dirac operator W commutes

$$C(Wv) = WC(v) \tag{4.A.8}$$

with the antilinear map

$$v \mapsto \mathcal{C}(v) := \epsilon \overline{v},\tag{4.A.9}$$

where ϵ is the "metric spinor" (4.A.6). In theoretical physics the transformation (4.A.9) is referred to as *charge conjugation* [14, 28].

Formula (4.A.8) implies that v is an eigenfunction of the massless Dirac operator corresponding to an eigenvalue λ if and only if C(v) is an eigenfunction of the massless Dirac operator corresponding to the same eigenvalue λ . Hence, all eigenvalues of the massless Dirac operator have even multiplicity. Moreover, any eigenfunction v and its "partner" C(v) make the same contribution to the spectral function (3.2.4) at every point x of the manifold M.

If, as in [28], we introduce a magnetic field, then we lose the commutation property (4.A.8) and the double eigenvalues split up. This indicates that the double eigenvalues of the massless Dirac operator correspond to the two different spins.

Property 4. This property has to do with a particular behaviour under SU(2) transformations. Let $R : M \to SU(2)$ be an arbitrary smooth special unitary matrix-function. Let us introduce new Pauli matrices

$$\tilde{\sigma}^{\alpha} := R \sigma^{\alpha} R^* \tag{4.A.10}$$

and a new operator \tilde{W} obtained by replacing the σ in (4.A.3) by $\tilde{\sigma}$. It turns out (and this is Property 4) that the two operators, \tilde{W} and W, are related in exactly the same way as the Pauli matrices, $\tilde{\sigma}$ and σ , that is,

$$\tilde{W} = RWR^*. \tag{4.A.11}$$

We now examine the geometric meaning of the transformation (4.A.10). Let us expand the new Pauli matrices $\tilde{\sigma}$ with respect to the basis (4.A.2):

$$\tilde{\sigma}^{\alpha}(x) = s^{j} \,\tilde{e}_{j}^{\ \alpha}(x). \tag{4.A.12}$$

Formulae (4.A.1), (4.A.12) and (4.A.10) give us the following identity relating the new vector fields \tilde{e}_j and the old vector fields e_j :

$$Rs^k R^* e_k = s^j \tilde{e}_j \,. \tag{4.A.13}$$

Resolving (4.A.13) for \tilde{e}_j we get

$$\tilde{e}_j = O_j^{\ k} e_k \,, \tag{4.A.14}$$

where the real scalars O_{jk} are given by the formula

$$O_j{}^k = \frac{1}{2} \operatorname{tr}(s_j R s^k R^*).$$
 (4.A.15)

Note that in writing formulae (4.A.13) and (4.A.14) we chose to hide the tensor index, i.e. we chose to hide the coordinate components of our vector fields. Say, formula (4.A.14) written in more detailed form reads $\tilde{e}_j^{\ \alpha} = O_j^{\ k} e_k^{\ \alpha}$.

The scalars (4.A.15) can be viewed as elements of a real 3×3 matrix-function O with the first index, j, enumerating rows and the second, k, enumerating columns. It is easy to check that this matrix-function O is special orthogonal. Hence, the new vector fields \tilde{e}_j are orthonormal and have the same orientation as the old vector fields e_j . We have shown that the transformation (4.A.10) has the geometric meaning of switching from our original oriented orthonormal frame e_j to a new oriented orthonormal frame \tilde{e}_j .

Formula (4.A.15) means that the special unitary matrix R is, effectively, a square root of the special orthogonal matrix O. It is easy to see that for a given matrix $O \in SO(3)$ formula (4.A.15) defines the matrix $R \in SU(2)$ uniquely up to sign. This observation allows us to view the issue of the geometric meaning of the transformation (4.A.10) the other way round: given a pair of orthonormal frames, e_j and \tilde{e}_j , with the same orientation (i.e. with sgn det $e_j^{\alpha} = \text{sgn det } \tilde{e}_j^{\alpha}$), we can recover the special orthogonal matrix-function O(x) from formula (4.A.14) and then attempt finding a smooth special unitary matrix-function R(x) satisfying (4.A.15). Unfortunately, this may not always be possible due to topological obstructions. We can only guarantee the absence of topological obstructions when the two frames, e_j and \tilde{e}_j , are sufficiently close to each other, which is equivalent to saying that we can only guarantee the absence of topological obstructions when the special orthogonal matrix-function O(x) is sufficiently close to the identity matrix for all $x \in M$.

We illustrate the possibility of a topological obstruction by means of an explicit example. Consider the unit torus \mathbb{T}^3 parameterized by cyclic coordinates x^{α} , $\alpha = 1, 2, 3$, of period 2π . The metric is assumed to be Euclidean. Define a pair of orthonormal frames

$$e_j{}^{\alpha} := \delta_j{}^{\alpha} \tag{4.A.16}$$

and

$$\tilde{e}_{1}^{\alpha} := \begin{pmatrix} \cos k_{3}x^{3} \\ \sin k_{3}x^{3} \\ 0 \end{pmatrix}, \qquad \tilde{e}_{2}^{\alpha} := \begin{pmatrix} -\sin k_{3}x^{3} \\ \cos k_{3}x^{3} \\ 0 \end{pmatrix}, \qquad \tilde{e}_{3}^{\alpha} := \begin{pmatrix} 0 \\ 0 \\ 1 \end{pmatrix}, \quad (4.A.17)$$

where k_3 is an odd integer. Let W and \tilde{W} be the massless Dirac operators corresponding to the frames (4.A.16) and (4.A.17) respectively. We claim that there does not exist a smooth matrix-function $R: \mathbb{T}^3 \to \mathrm{SU}(2)$ which would give (4.A.15), where O(x) is the special orthogonal matrix-function defined by formula (4.A.14). We justify this claim in two different ways.

Justification 1. Resolving the system (4.A.14)-(4.A.17) locally for R, we get

$$R(x^{3}) = \pm \begin{pmatrix} e^{\frac{i}{2}k_{3}x^{3}} & 0\\ 0 & e^{-\frac{i}{2}k_{3}x^{3}} \end{pmatrix}, \qquad (4.A.18)$$

and this solution is unique modulo choice of sign; here the freedom in the choice of sign is not surprising as SU(2) is the double cover of SO(3). Formula (4.A.18) defines a continuous single-valued matrix-function on the unit torus \mathbb{T}^3 if and only if the integer k_3 is even, which it is not.

Justification 2. It is sufficient to show that the two operators, W and \tilde{W} , have different spectra. Straightforward separation of variables shows that zero is an eigenvalue of the operator W but not an eigenvalue of the operator \tilde{W} .

One can generalize the above example by introducing rotations in three different directions, which leads to eight genuinely distinct parallelizations. See also [74] page 524 or [10] page 21.

Let us emphasize that the topological obstructions we were discussing have nothing to do with Stiefel–Whitney classes. We are working on a parallelizable manifold and the Stiefel–Whitney class of such a manifold is trivial. The topological issue at hand is that our parallelizable manifold may be equipped with different spin structures.

We say that two massless Dirac operators, W and \tilde{W} , are equivalent if there exists a smooth matrix-function $R : M \to SU(2)$ such that the corresponding Pauli matrices, σ^{α} and $\tilde{\sigma}^{\alpha}$, are related in accordance with (4.A.10). In view of Property 4 (see formula (4.A.11)) all massless Dirac operators from the same equivalence class generate the same spectral function (3.2.4) and the same counting function (3.2.5), so for the purposes of Chapter 4 viewing such operators as equivalent is most natural. As explained above, there may be many distinct equivalence classes of massless Dirac operators, the difference between which is topological. Studying the spectral theoretic implications of these topological differences is beyond the scope of Chapter 4. The two-term asymptotics (4.1.16) and (4.1.17) derived in the main text of Chapter 4 do not feel this topology.

In theoretical physics the SU(2) freedom involved in defining the massless Dirac operator is interpreted as a gauge degree of freedom. We do not adopt this point of view (at least explicitly) in order to fit the massless Dirac operator into the standard spectral theoretic framework.

We defined the massless Dirac operator (4.A.3) as an operator acting on 2columns of scalar functions, i.e. on 2-columns of quantities which do not change under changes of local coordinates. This necessitated the introduction of the density $\sqrt{\det g_{\alpha\beta}}$ in the formula (4.A.7) for the inner product. In spectral theory it is more common to work with half-densities. Hence, we introduce the operator

$$W_{1/2} := (\det g_{\kappa\lambda})^{1/4} W (\det g_{\mu\nu})^{-1/4}$$
(4.A.19)

which maps half-densities to half-densities. We call the operator (4.A.19) the massless Dirac operator on half-densities.

4.B The spectrum for the torus and the sphere

In this appendix we examine the massless Dirac operator on the unit torus \mathbb{T}^3 and the unit sphere \mathbb{S}^3 and compare our asymptotic formulae (4.1.16) and (4.1.17) with known explicit formulae. The torus is assumed to be equipped with Euclidean metric (see also Appendix 4.A) whereas the sphere is assumed to be equipped with metric induced by the natural embedding of \mathbb{S}^3 in Euclidean space \mathbb{R}^4 . Note that in view of the obvious symmetries of the torus and the sphere the scalar function $e(\lambda, x, x)/\sqrt{\det g_{\alpha\beta}(x)}$ is constant (see also Remark 4.1.2), so formulae (4.1.16) and (4.1.17) are in this case equivalent, in the sense that they follow from one another. Hence, we will be dealing with formula (4.1.17) only.

We have $\operatorname{Vol} \mathbb{T}^3 = (2\pi)^3$, so for the torus formula (4.1.17) reads

$$N(\lambda) = \frac{4}{3}\pi\lambda^3 + o(\lambda^2).$$
(4.B.1)

The nonperiodicity condition (see Definitions 8.3 and 8.4 from [22]) is fulfilled for the torus, so, according to Theorem 8.4 from [22], the asymptotic formula (4.B.1) holds as it is, without mollification.

In order to test formula (4.B.1) we calculate the spectrum of the massless Dirac operator on \mathbb{T}^3 explicitly. We do this first for the spin structure associated with the frame (4.A.16). Then the spectrum is as follows.

- Zero is an eigenvalue of multiplicity two.
- For each m ∈ Z³ \ {0} we have the eigenvalue ||m|| and unique (up to rescaling) eigenfunction, with eigenfunctions corresponding to different m being linearly independent.
- For each m ∈ Z³ \ {0} we have the eigenvalue -||m|| and unique (up to rescaling) eigenfunction, with eigenfunctions corresponding to different m being linearly independent.

Hence, $N(\lambda) + 1$ is the number of integer lattice points inside a 2-sphere of radius λ in \mathbb{R}^3 centred at the origin. According to [38] the latter admits the asymptotic expansion

$$\frac{4}{3}\pi\lambda^3 + O_{\varepsilon}(\lambda^{21/16+\varepsilon}) \tag{4.B.2}$$

as $\lambda \to +\infty$, with ε being an arbitrary positive number. This agrees with our asymptotic formula (4.B.1).

Let us now consider the spin structure associated with the frame (4.A.17). Then the spectrum is as follows.

- For each m ∈ Z³ we have the eigenvalue ||m − (0, 0, 1/2)|| and unique (up to rescaling) eigenfunction, with eigenfunctions corresponding to different m being linearly independent.
- For each m ∈ Z³ we have the eigenvalue -||m (0, 0, 1/2)|| and unique (up to rescaling) eigenfunction, with eigenfunctions corresponding to different m being linearly independent.

Hence, $N(\lambda)$ is the number of integer lattice points inside a 2-sphere of radius λ in \mathbb{R}^3 centred at (0, 0, 1/2). Here the sphere is shifted from the origin so one cannot apply the result from [38]. However, as the shift is rational, one can reduce the problem to counting integer lattice points in a rational ellipsoid centred at the origin, and an application of the result from [19] gives us for the shifted sphere the same asymptotic expansion (4.B.2) as for the sphere centred at the origin.

As explained in Appendix 4.A, the unit torus \mathbb{T}^3 admits a total of eight different spin structures. For each of these the problem of counting positive eigenvalues of the massless Dirac operator reduces to counting integer lattice points inside a 2-sphere of radius λ in \mathbb{R}^3 (possibly, shifted from the origin by a rational shift), so in all eight cases we do get (4.B.1). In fact, we can replace the remainder $o(\lambda^2)$ in (4.B.1) by $O_{\varepsilon}(\lambda^{21/16+\varepsilon})$ and this holds for all eight different spin structures.

In the remainder of this appendix we examine the massless Dirac operator on the unit sphere \mathbb{S}^3 . We have Vol $\mathbb{S}^3 = 2\pi^2$, so for the sphere formula (4.1.17) reads

$$N(\lambda) = \frac{\lambda^3}{3} + o(\lambda^2). \tag{4.B.3}$$

The nonperiodicity condition fails for the sphere because all geodesics are closed with period 2π , so formula (4.B.3) cannot be used in its original form and has to be mollified, see Remark 4.1.1. We will deal with the mollification issue later and give explicit formulae for the eigenvalues first.

It is known that \mathbb{S}^3 admits a unique spin structure, see Section 5 in [10]. The spectrum of the massless Dirac operator on \mathbb{S}^3 has been computed by different

authors using different methods [77, 78, 8, 10] and reads as follows: the eigenvalues are

$$\pm \left(k + \frac{1}{2}\right), \qquad k = 1, 2, \dots,$$
 (4.B.4)

and their multiplicity is

$$k(k+1).$$
 (4.B.5)

The mollification procedure from Section 7 of [22] goes as follows. Put $N(\lambda) := 0$ for $\lambda \leq 0$ and take an arbitrary real-valued even function $\rho(\lambda)$ from Schwartz space $\mathcal{S}(\mathbb{R})$ whose Fourier transform $\hat{\rho}(t)$ satisfies conditions $\hat{\rho}(0) = 1$ and supp $\hat{\rho} \subset$ $(-2\pi, 2\pi)$. Then, according to Theorem 7.2 from [22], the mollified version of formula (4.B.3) reads

$$\int N(\lambda - \mu) \,\rho(\mu) \,d\mu = \frac{\lambda^3}{3} + O(\lambda)$$

and this result holds notwithstanding the failure of the nonperiodicity condition. However, for the sphere there is a much simpler way of testing our asymptotic formula. Let $\lambda \geq 2$ be integer. Taking an integer λ puts us exactly in the middle of the gap between two consecutive clusters of eigenvalues, see formulae (4.B.4) and (4.B.5), and achieves the same averaging effect as convolution with a function from Schwartz space. For integer $\lambda \geq 2$ we get

$$N(\lambda) = \sum_{k=1}^{\lambda-1} k(k+1) = \frac{\lambda^3}{3} - \frac{\lambda}{3}$$

which agrees with our asymptotic formula (4.B.3).

Chapter 5

Spectral theoretic characterization of the massless Dirac action

5.1 Main result

Consider a first order differential operator A acting on 2-columns $v = \begin{pmatrix} v_1 & v_2 \end{pmatrix}^T$ of complex-valued half-densities over a connected compact 3-dimensional manifold M without boundary. We assume the coefficients of the operator A to be infinitely smooth. We also assume that the operator A is formally self-adjoint, see (4.1.1).

Let $A_1(x,\xi)$ be the principal symbol of the operator A. The principal symbol $A_1(x,\xi)$ is a 2 × 2 Hermitian matrix-function on the cotangent bundle T^*M , linear in every fibre T_x^*M . We assume that the operator A is elliptic in the sense described by (4.1.2).

We now make two additional assumptions:

Assumption 1: We assume the principal symbol to be trace-free.

Assumption 2: We assume the subprincipal symbol of the operator A to be zero (see Section 2.1.3 in [72] for background material pertaining to the subprincipal symbol).

The latter condition implies that our first order differential operator A is completely determined by its principal symbol. Namely, in local coordinates the full symbol $A(x,\xi)$ is expressed via the principal symbol $A_1(x,\xi)$ as

$$A(x,\xi) = A_1(x,\xi) - \frac{i}{2}(A_1)_{x^{\alpha}\xi_{\alpha}}(x)$$
(5.1.1)

where the subscripts x^{α} and ξ_{α} indicate partial derivatives and the repeated index α indicates summation over $\alpha = 1, 2, 3$.

We study the eigenvalue problem

$$Av = \lambda wv \tag{5.1.2}$$

where w(x) is a given infinitely smooth positive scalar weight function. Obviously, the problem (5.1.2) has the same spectrum as the problem

$$w^{-1/2}Aw^{-1/2}v = \lambda v \tag{5.1.3}$$

so it may appear that the weight function w(x) is redundant. We will, however, work with the eigenvalue problem (5.1.2) rather than with (5.1.3) because we want our problem to possess a gauge degree of freedom (5.5.1). This gauge degree of freedom will eventually manifest itself as the conformal invariance of the massless Dirac action, see Section 5.5 for details.

The problem (5.1.2) has a discrete spectrum accumulating to $\pm \infty$. Recall the definition of the counting function from Chapter 3

$$N(\lambda) := \sum_{0 < \lambda_k < \lambda} 1, \tag{5.1.4}$$

i.e. the number of eigenvalues λ_k of the problem (5.1.2), with account of multiplicities, between zero and a positive λ .

Theorem 8.4 from [22] states that under appropriate assumptions on periodic trajectories our counting function admits a two-term asymptotic expansion

$$N(\lambda) = a\lambda^3 + b\lambda^2 + o(\lambda^2) \tag{5.1.5}$$

as $\lambda \to +\infty$. If one wishes to reformulate the asymptotic formula (5.1.5) in such a way that it remains valid without assumptions on periodic trajectories, this can easily be achieved, say, by taking a convolution with a function from Schwartz space $\mathcal{S}(\mathbb{R})$; see Theorem 7.2 in [22] and Chapter 3 for further details.

The objective of Chapter 5 is to establish the geometric meaning of the coefficient b. The logic behind restricting our analysis to the case when the manifold is 3-dimensional and A is a 2×2 matrix differential operator with trace-free principal symbol and zero subprincipal symbol is that this is the simplest¹ eigenvalue problem for a system of partial differential equations. Hence, it is ideal for the purpose of establishing the geometric meaning of the coefficient b. In addition, we are able to build our analysis on that presented in Chapter 3 and Chapter 4: the problem considered here is a special case of that considered in Chapter 4, the extra assumption being the vanishing subprincipal symbol.

In order to establish the geometric meaning of the coefficient b we first need to identify the geometric objects encoded within our eigenvalue problem (5.1.2).

Geometric object 1: the *metric*. Observe that the determinant of the principal symbol is a negative definite quadratic form in the dual variable ξ ,

$$\det A_1(x,\xi) = -g^{\alpha\beta}\xi_\alpha\xi_\beta, \qquad (5.1.6)$$

¹The only way the eigenvalue problem (5.1.2) can be made simpler is by lowering the dimension of the manifold to two. Unfortunately, this raises geometric issues which we cannot currently resolve.

and the coefficients $g^{\alpha\beta}(x) = g^{\beta\alpha}(x)$, $\alpha, \beta = 1, 2, 3$, appearing in (5.1.6) can be interpreted as the components of a (contravariant) Riemannian metric.

Geometric object 2: the nonvanishing spinor field. The determinant of the principal symbol does not determine the principal symbol uniquely. In order to identify a further geometric object encoded within the principal symbol $A_1(x,\xi)$ we will now start varying this principal symbol, assuming the metric g, defined by formula (5.1.6), to be fixed (prescribed).

Let us fix a reference principal symbol $B_1(x,\xi)$ corresponding to the prescribed metric g and look at all principal symbols $A_1(x,\xi)$ which correspond to the same prescribed metric g and are sufficiently close to the reference principal symbol. Restricting our analysis to principal symbols which are close to the reference principal symbol allows us to avoid dealing with certain topological issues; this restriction will be dropped in Section 5.4. It turns out, see Section 5.2, that the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ are related as

$$A_1(x,\xi) = R(x) B_1(x,\xi) R^*(x), \qquad (5.1.7)$$

where

$$R: M \to \mathrm{SU}(2) \tag{5.1.8}$$

is a unique infinitely smooth special unitary matrix-function which is close to the identity matrix. Thus, special unitary matrix-functions R(x) provide a convenient parametrization of principal symbols with prescribed metric g.

Let *B* be the differential operator with principal symbol $B_1(x,\xi)$ and zero subprincipal symbol. It is important to emphasize that for the operators *A* and *B* themselves, as opposed to their principal symbols, we have, in general, the inequality $A \neq RBR^*$ because the substitution of (5.1.7) into (5.1.1) generates terms with derivatives of the matrix-function R(x). Hence, the transformation of operators $B \mapsto A$ specified by formula (5.1.7) does, in general, change the spectrum. The choice of reference principal symbol $B_1(x,\xi)$ in our construction is arbitrary, as long as this principal symbol corresponds to the prescribed metric g, i.e. as long as we have det $B_1(x,\xi) = -g^{\alpha\beta}(x) \xi_{\alpha}\xi_{\beta}$ for all $(x,\xi) \in T^*M$. It is natural to ask the question: what happens if we choose a different reference principal symbol $B_1(x,\xi)$? The freedom in choosing the reference principal symbol $B_1(x,\xi)$ is a gauge degree of freedom in our construction and our results are invariant under changes of the reference principal symbol. This issue is addressed in Section 5.6.

In order to work effectively with special unitary matrices we need to choose coordinates on the 3-dimensional Lie group SU(2). It is convenient to describe a 2×2 special unitary matrix by means of a (Weyl) spinor ζ , i.e. a pair of complex numbers ζ^a , a = 1, 2. The relationship between a matrix $R \in SU(2)$ and a nonzero spinor ζ is given by the formula

$$R = \frac{1}{\|\zeta\|} \begin{pmatrix} \overline{\zeta^1} & \overline{\zeta^2} \\ -\zeta^2 & \zeta^1 \end{pmatrix}, \qquad (5.1.9)$$

where the overline stands for complex conjugation and $\|\zeta\| := \sqrt{|\zeta^1|^2 + |\zeta^2|^2}$. Here $\overline{\zeta^a}$ is defined as follows: $\overline{\zeta^a} := \overline{\zeta}^{\dot{a}}$. Unlike in Chapter 2, we are able to avoid introducing dotted spinor indices using instead matrix notation throughout Chapter 5. This simplifies our calculations.

Formula (5.1.9) establishes a one-to-one correspondence between SU(2) matrices and nonzero spinors, modulo a rescaling of the spinor by an arbitrary positive real factor. We choose to specify the scaling of our spinor field $\zeta(x)$ in accordance with

$$\|\zeta(x)\| = w(x). \tag{5.1.10}$$

Remark 5.1.1. In Chapter 4 we chose to work with a Teleparallel connection (metric compatible affine connection with zero curvature) rather than with a spinor field. These are closely related objects: locally a Teleparallel connection is equivalent to a normalized ($\|\zeta(x)\| = 1$) spinor field modulo rigid rotations (5.7.4) of the latter. Geometric object 3: the topological charge. It is known, see Section 4.3, that the existence of a principal symbol implies that our manifold M is parallelizable. Parallelizability, in turn, implies orientability. Having chosen a particular orientation, we allow only changes of local coordinates x^{α} , $\alpha = 1, 2, 3$, which preserve orientation.

We define the topological charge as

$$\mathbf{c} := -\frac{i}{2}\sqrt{\det g_{\alpha\beta}} \operatorname{tr}((A_1)_{\xi_1}(A_1)_{\xi_2}(A_1)_{\xi_3}), \qquad (5.1.11)$$

with the subscripts ξ_{α} indicating partial derivatives. As explained in Section 4.3, the number **c** defined by formula (5.1.11) can take only two values, +1 or -1, and describes the orientation of the principal symbol relative to the chosen orientation of local coordinates.

We have identified three geometric objects encoded within the eigenvalue problem (5.1.2) — metric, nonvanishing spinor field and topological charge — defined in accordance with formulae (5.1.6)–(5.1.11). Consequently, one would expect the coefficients a and b from formula (5.1.5) to be expressed via these three geometric objects. This assertion is confirmed by the following theorem which is the main result of Chapter 5.

Theorem 5.1.1. The coefficients in the two-term asymptotics (5.1.5) are given by the formulae

$$a = \frac{1}{6\pi^2} \int_M \|\zeta\|^3 \sqrt{\det g_{\alpha\beta}} \, dx \,, \tag{5.1.12}$$

$$b = \frac{S(\zeta)}{2\pi^2},$$
 (5.1.13)

where $S(\zeta)$ is the massless Dirac action (5.A.1) with Pauli matrices

$$\sigma^{\alpha} := (B_1)_{\xi_{\alpha}}, \qquad \alpha = 1, 2, 3.$$
 (5.1.14)

Theorem 5.1.1 warrants the following remarks.

Firstly, recall that the *B* appearing in Theorem 5.1.1 is our reference operator which we need to describe all possible operators *A* with given metric *g*. It is natural to ask the question: what happens if we take A = B? In this case formula (5.1.13) holds with spinor field $\zeta^1(x) = w(x)$, $\zeta^2(x) = 0$. This in itself is a nontrivial result.

Secondly, the topological charge \mathbf{c} does not appear explicitly in Theorem 5.1.1. Nevertheless, it is implicitly present in our Pauli matrices (5.1.14). Indeed, formula (5.1.7) implies that the integer quantity

$$-\frac{i}{2}\sqrt{\det g_{\alpha\beta}} \operatorname{tr}((B_1)_{\xi_1}(B_1)_{\xi_2}(B_1)_{\xi_3})$$
(5.1.15)

has the same value as (5.1.11).

Thirdly, it is tempting to apply Theorem 5.1.1 in the case when the operator A is itself a massless Dirac operator. This cannot be done because a massless Dirac operator acts on spinors, the components of which are invariant under changes of local coordinates, rather than on pairs of half-densities. This impediment can be overcome by switching to a massless Dirac operator on half-densities, see formula (4.A.3) in Appendix 4.A. However, we cannot take A to be a massless Dirac operator on half-densities either because, according to Lemma 4.6.1 from Chapter 4, the latter has a nontrivial subprincipal symbol.

Finally, Theorem 5.1.1 provides a fresh perspective on the history of the subject of two-term spectral asymptotics for first order systems, see Section 3.6 of Chapter 3 for details. Namely, Theorem 5.1.1 shows that even in the simplest case the second asymptotic coefficient for a first order system has a highly nontrivial geometric meaning. At a formal level, the application of microlocal techniques does not require the use of advanced differential geometric concepts. However, the calculations involved are so complicated that it is hard to avoid mistakes without an understanding of the differential geometric content of the spectral problem. Chapter 5 is organized as follows. In Section 5.2 we explain the origins of formula (5.1.7) and in Section 5.3 we give the proof of Theorem 5.1.1. In Section 5.4 we introduce the concept of spin structure which allows us to drop the restriction that our principal symbol $A_1(x,\xi)$ is sufficiently close to the reference principal symbol $B_1(x,\xi)$. And in Sections 5.5–5.7 we show that our formula (5.1.13) is invariant under the action of certain gauge transformations.

5.2 Spinor representation of the principal symbol

Let $A_1(x,\xi)$ and $B_1(x,\xi)$ be a pair of trace-free Hermitian 2×2 principal symbols and let g be a prescribed Riemannian metric. Both $A_1(x,\xi)$ and $B_1(x,\xi)$ are assumed to be linear in ξ :

$$A_1(x,\xi) = A_1^{(\alpha)}(x)\,\xi_\alpha\,, \qquad (5.2.1)$$

$$B_1(x,\xi) = B_1^{(\alpha)}(x)\,\xi_\alpha\,, \qquad (5.2.2)$$

where $A_1^{(\alpha)}(x)$ and $B_1^{(\alpha)}(x)$, $\alpha = 1, 2, 3$, are some trace-free Hermitian 2×2 matrix-functions. The assumption that our principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ are linear in ξ means, of course, that we are dealing with differential operators as opposed to pseudodifferential operators. The principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ are assumed to satisfy

$$\det A_1(x,\xi) = \det B_1(x,\xi) = -g^{\alpha\beta}(x)\,\xi_\alpha\xi_\beta \tag{5.2.3}$$

for all $(x,\xi) \in T^*M$, and are also assumed to be sufficiently close in terms of the $C^{\infty}(M)$ topology applied to the matrix-functions $A_1^{(\alpha)}(x)$ and $B_1^{(\alpha)}(x)$, $\alpha = 1, 2, 3$.

Our task in this section is to show that there exists a unique infinitely smooth special unitary matrix-function (5.1.8) which is close to the identity matrix and

which relates the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ in accordance with formula (5.1.7).

We follow the convention of [23] and Chapter 4 in denoting the elements of the matrices $A_1^{(\alpha)}$ and $B_1^{(\alpha)}$ as $(A_1^{(\alpha)})_{\dot{a}b}$ and $(B_1^{(\alpha)})_{\dot{a}b}$ respectively, where the dotted index, running through the values $\dot{1}, \dot{2}$, enumerates the rows and the undotted index, running through the values 1, 2, enumerates the columns.

Put

$$e_1{}^{\alpha} := \operatorname{Re}(A_1^{(\alpha)})_{12}, \quad e_2{}^{\alpha} := -\operatorname{Im}(A_1^{(\alpha)})_{12}, \quad e_3{}^{\alpha} := \operatorname{Re}(A_1^{(\alpha)})_{11}, \quad (5.2.4)$$

$$f_1^{\alpha} := \operatorname{Re}(B_1^{(\alpha)})_{12}, \quad f_2^{\alpha} := -\operatorname{Im}(B_1^{(\alpha)})_{12}, \quad f_3^{\alpha} := \operatorname{Re}(B_1^{(\alpha)})_{11}.$$
 (5.2.5)

As explained in Section 4.3, formula (5.2.4) defines a *frame* — a triple of infinitely smooth real orthonormal vector fields $e_j(x)$, j = 1, 2, 3, on the manifold M and, moreover, the principal symbol $A_1(x,\xi)$ is equivalent to the frame e_j in the sense that the principal symbol uniquely determines the frame and the frame uniquely determines the principal symbol. Similarly, formula (5.2.5) defines a frame f_j which is equivalent to the principal symbol $B_1(x,\xi)$.

Condition (5.2.3) implies that the frames e_j and f_j are orthonormal with respect to the same metric. Hence, this pair of orthonormal frames is related as

$$e_j{}^{\alpha} = O_j{}^k f_k{}^{\alpha}, \tag{5.2.6}$$

where O(x) is a 3 × 3 orthogonal matrix-function with elements

$$O_j{}^k = \delta^{kl} g_{\alpha\beta} \, e_j{}^\alpha f_l{}^\beta.$$

As we assumed the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ to be close, the frames e_j and f_j are also close. Consequently, the matrix-function O(x) is close to the identity matrix and, hence, special unitary.

It is well-known that the Lie group SO(3) is locally (in a neighbourhood of the identity) isomorphic to the Lie group SU(2). According to formulae (4.A.15) and (4.A.2) from Appendix 4.A, a 3×3 special orthogonal matrix O is expressed via a 2×2 special unitary matrix R as

$$O_j{}^k = \frac{1}{2} \operatorname{tr}(s_j R s^k R^*), \qquad (5.2.7)$$

where

$$s^{1} := \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} = s_{1}, \quad s^{2} := \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix} = s_{2}, \quad s^{3} := \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} = s_{3}. \quad (5.2.8)$$

Formula (5.2.7) tells us that a 3×3 special orthogonal matrix is, effectively, the square of a 2×2 special unitary matrix. Formula (5.2.7) provides a local diffeomorphism between neighbourhoods of the identity in SO(3) and in SU(2).

A straightforward calculation shows that formulae (5.2.1), (5.2.2) and (5.2.4)–(5.2.8) imply formula (5.1.7).

Let us now define Pauli matrices $\sigma^{\alpha}(x)$ in accordance with formula (5.1.14). Of course, we have

$$\sigma^{\alpha} := B_1^{(\alpha)}(x), \qquad \alpha = 1, 2, 3,$$
(5.2.9)

where the $B_1^{(\alpha)}$ are the matrix-functions from formula (5.2.2). We could stick with the notation $B_1^{(\alpha)}$ but we choose to switch to σ^{α} because this is how Pauli matrices are traditionally denoted in the subject.

It is easy to see that formula (5.2.3) implies

$$\sigma^{\alpha}\sigma^{\beta} + \sigma^{\beta}\sigma^{\alpha} = 2Ig^{\alpha\beta}, \qquad (5.2.10)$$

where I is the 2 × 2 identity matrix. Formula (5.2.10) means that our σ^{α} satisfy the defining relation for Pauli matrices.

Formulae (5.2.6)–(5.2.8), (5.1.9), (5.2.5) and (5.2.9) allow us to express the frame e_j via the spinor field ζ and Pauli matrices σ^{α} . We took great care to choose coordinates on the Lie group SU(2) (i.e. structure of the matrix in the RHS of formula (5.1.9)) so that the resulting expressions agree with formulae (B.3), (B.4) and (B.1) from [23]. The only difference is in notation: the ϑ^j in Appendix B of [23] stands for $\vartheta^j{}_{\alpha} = \delta^{jk}g_{\alpha\beta} e_k{}^{\beta}$ (compare with formula (5.3.4)) and the spinor in Appendix B of [23] is denoted by ξ rather than by ζ . We do not denote the spinor by ξ in the current chapter because in microlocal analysis the letter ξ is traditionally reserved for denoting momentum.

The fact that our construction agrees with that in [23] will become important in the next section when we will make use of a particular formula from [23].

Remark 5.2.1. As explained in Section 4.3, the topological charge, initially defined in accordance with formula (5.1.11), can be equivalently rewritten in terms of frames as

$$\mathbf{c} = \operatorname{sgn} \det e_j{}^{\alpha} = \operatorname{sgn} \det f_j{}^{\alpha}. \tag{5.2.11}$$

The paper [23] was written under the assumption that

$$\mathbf{c} = +1,$$
 (5.2.12)

see formula (A.1) in [23]. This means that care is required when using the results of [23]. Namely, in the next section we will first prove Theorem 5.1.1 for the case (5.2.12) and then provide a separate argument explaining why formula (5.1.13)remains true in the case

$$\mathbf{c} = -1.$$
 (5.2.13)
5.3 Proof of Theorem 5.1.1

We prove Theorem 5.1.1 by examining the equivalent spectral problem (5.1.3). Note that transition from (5.1.2) to (5.1.3) is a special case of the gauge transformation (5.5.1). As explained in the beginning of Section 5.5, the gauge transformation (5.5.1) preserves the structure of our eigenvalue problem: the principal symbol of the operator $w^{-1/2}Aw^{-1/2}$ is trace-free and its subprincipal symbol is zero.

We now apply Theorem 4.1.1 from Chapter 4 to the eigenvalue problem (5.1.3).

Our formula (5.1.12) is an immediate consequence of formula (4.1.12) from Chapter 4 and our formulae (5.1.6) and (5.1.10). Here, of course, we use the fact that we are working in dimension three.

The proof of formula (5.1.13) is more delicate so we initially consider the case

$$w(x) = 1, \qquad \forall x \in M. \tag{5.3.1}$$

In this case, according to formulae (4.1.13) and (4.8.1) from Chapter 4, we have

$$b = \frac{3\mathbf{c}}{8\pi^2} \int_M *T^{\mathrm{ax}} \sqrt{\det g_{\alpha\beta}} \, dx \,, \tag{5.3.2}$$

where

$$* T^{ax} = \frac{\delta_{kl}}{3} \sqrt{\det g^{\alpha\beta}} \left[e^{k_{1}} \partial e^{l_{3}} / \partial x^{2} + e^{k_{2}} \partial e^{l_{1}} / \partial x^{3} + e^{k_{3}} \partial e^{l_{2}} / \partial x^{1} - e^{k_{1}} \partial e^{l_{2}} / \partial x^{3} - e^{k_{2}} \partial e^{l_{3}} / \partial x^{1} - e^{k_{3}} \partial e^{l_{1}} / \partial x^{2} \right], \quad (5.3.3)$$

$$e^{j}{}_{\alpha} = \delta^{jk} g_{\alpha\beta} \, e_{k}{}^{\beta}. \tag{5.3.4}$$

The real scalar field $*T^{ax}(x)$ has the geometric meaning of the Hodge dual of axial torsion of the Teleparallel connection, see Chapter 4 for details.

Let us now drop the assumption (5.3.1).

The introduction of a weight function is equivalent to a scaling of the principal symbol $A_1(x,\xi) \mapsto (w(x))^{-1}A_1(x,\xi)$, which, in view of formulae (5.2.4) and (5.1.6), leads to a scaling of the frame

$$e_j \mapsto w^{-1} e_j \tag{5.3.5}$$

and corresponding scaling of the metric

$$g^{\alpha\beta} \mapsto w^{-2} g^{\alpha\beta}.$$
 (5.3.6)

Substituting (5.3.5) and (5.3.6) into (5.3.4) and (5.3.3) we see that the integrand in formula (5.3.2) scales as

$$*T^{\mathrm{ax}}\sqrt{\det g_{\alpha\beta}} \mapsto w^2 * T^{\mathrm{ax}}\sqrt{\det g_{\alpha\beta}}$$
. (5.3.7)

Here the remarkable fact is that we do not get derivatives of the weight function because these cancel out due to the antisymmetric structure of the RHS of formula (5.3.3). In other words, axial torsion, defined by formulae (4.1.14) and (4.3.13) from Chapter 4, has the remarkable property that it scales in a covariant manner under scaling of the frame. Note that the full torsion tensor, defined by formula (4.3.13) from Chapter 4, does not possess such a covariance property.

Formula (5.3.7) tells us that in order to accommodate an arbitrary weight function w(x) we need to multiply the integrand in formula (5.3.2) by $(w(x))^2$, which gives us

$$b = \frac{3\mathbf{c}}{8\pi^2} \int_M w^2 * T^{\mathrm{ax}} \sqrt{\det g_{\alpha\beta}} \, dx \,. \tag{5.3.8}$$

Let us emphasize that the metric and torsion appearing in formula (5.3.8) are the original, unscaled metric and torsion determined by the original, unscaled principal symbol $A_1(x,\xi)$. The scaling argument has been incorporated into the factor $(w(x))^2$.

We now need to express the integrand in (5.3.8) in terms of the spinor field ζ .

We already have a expression for the weight function in terms of the spinor field, see formula (5.1.10). So we only need to express the Hodge dual of axial torsion in terms of the spinor field. Formulae (5.2.4), (5.2.1), (5.2.2), (5.2.9), (5.1.7) and (5.1.9) allow us to express the frame e_j via the spinor field ζ and Pauli matrices σ^{α} . Hence one needs to combine all these formulae to get explicit expressions for the vector fields e_j , j = 1, 2, 3, and substitute these into (5.3.4) and (5.3.3). This is a massive calculation. Fortunately, for the case (5.2.12) this calculation was carried out in Appendix B of [23]: in the notation of the current chapter formula (B.5) from [23] reads

$$* T^{ax} = \frac{4 \operatorname{Re}(\zeta^* W \zeta)}{3 \|\zeta\|^2}, \qquad (5.3.9)$$

where W is the massless Dirac operator (4.A.3).

Formulae (5.3.8), (5.1.10), (5.3.9) and (5.A.1) imply formula (5.1.13). This completes the proof of Theorem 5.1.1 for the case (5.2.12).

In order to prove formula (5.1.13) for the case (5.2.13), we invert coordinates $(x^{\alpha} \mapsto -x^{\alpha})$, which changes the sign of topological charge and allows us to use formula (5.1.13). We then invert coordinates again and use the facts that

- the integrand of the massless Dirac action (5.A.1) is invariant under inversion of coordinates and
- our spinor field ζ defined by formulae (5.1.7)−(5.1.10) is an anholonomic object, i.e. it does not depend on the choice of coordinates. □

5.4 Spin structure

In stating our results in Section 5.1 we assumed the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ to be sufficiently close. This was done in order to ensure that equation (5.1.7) could be resolved with respect to the special unitary matrix-function R(x). The restriction of closeness of principal symbols can be overcome by means of the introduction of the concept of spin structure.

Definition 5.4.1. We say that the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ have the same spin structure if there exists an infinitely smooth special unitary matrixfunction (5.1.8) such that we have (5.1.7).

Remark 5.4.1. Principal symbols with the same spin structure form an equivalence class.

The closeness of the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ is never used in the proof of Theorem 5.1.1. All that is needed for Theorem 5.1.1 to be true is for the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ to have the same spin structure, i.e. belong to the same equivalence class.

Hence, it would have been more logical to identify the spin structure as a separate geometric object from the very start, in Section 5.1, and avoid arguments relying on the closeness of the principal symbols. We chose not to proceed along this route in order to make the exposition in Section 5.1 as simple and clear as possible.

The only difference between the "local" setting (the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ are assumed to be close) and the "global" setting (the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ are assumed to have the same spin structure) is that we can no longer claim that the special unitary matrix-function R(x) appearing in formula (5.1.7) is unique. In the "local" setting uniqueness was achieved by requiring R(x) to be close to the identity matrix, whereas in the "global" setting R(x) is defined modulo sign (not surprising as SU(2) is the double cover of SO(3)). This sign indeterminacy does not affect formula (5.1.13) because the massless Dirac action is quadratic in the spinor field.

The number of different spin structures (i.e. number of equivalence classes of principal symbols) depends on the topology of the manifold. Say, the torus \mathbb{T}^3 admits eight different spin structures, whereas the sphere \mathbb{S}^3 admits a unique spin structure. See Appendices 4.A and 4.B in Chapter 4 and further bibliographic references therein for more details.

It may seem that our Definition 5.4.1 is different from the definition of spin structure in differential geometric literature. Indeed, differential geometers do not operate with concepts such as the principal symbol, using frames instead. However, it has been shown in Section 4.3 that a principal symbol is equivalent to a frame, so our "microlocal" definition of spin structure can be easily recast in terms of frames, bringing it in agreement with the traditional differential geometric one.

Here it is important to emphasize that we do not claim to have redefined the notion of spin structure for the most general case. We work in the very specific setting of dimension three.

5.5 Conformal invariance

Let us transform the differential operator A and weight w(x) as

$$A \mapsto e^{-\varphi/2} A e^{-\varphi/2}, \qquad w \mapsto e^{-\varphi} w,$$
 (5.5.1)

where $\varphi : M \to \mathbb{R}$ is an arbitrary infinitely smooth real-valued scalar function. The transformation (5.5.1) does not change the spectrum of our eigenvalue problem (5.1.2) and, moreover, preserves its structure: the principal symbol remains trace-free and the subprincipal symbol remains zero (the latter follows from the well-known formula for the subprincipal symbol of a composition of pseudodifferential operators, see formula (1.4) in [27]).

In this section we examine how the gauge transformation (5.5.1) works its way into scalings of the metric, Pauli matrices and spinor field.

Formulae (5.1.6) and (5.5.1) imply that the metric transforms as

$$g^{\alpha\beta} \mapsto e^{-2\varphi} g^{\alpha\beta} \tag{5.5.2}$$

which means that we are looking at a conformal scaling of the metric.

We scale the reference principal symbol $B_1(x,\xi)$ the same way as the principal symbol $A_1(x,\xi)$, i.e. as

$$B_1 \mapsto e^{-\varphi} B_1, \tag{5.5.3}$$

because this way we maintain the condition (5.2.3). Formulae (5.1.14) and (5.5.3) imply that the Pauli matrices scale as

$$\sigma^{\alpha} \mapsto e^{-\varphi} \sigma^{\alpha}. \tag{5.5.4}$$

Of course, the scaling of Pauli matrices (5.5.4) agrees with the scaling of the metric (5.5.2) in the sense that the scaled Pauli matrices and metric satisfy the identity (5.2.10).

Formulae (5.1.10) and (5.5.1) imply that the spinor field scales as

$$\zeta \mapsto e^{-\varphi}\zeta. \tag{5.5.5}$$

Let us now examine what happens to the massless Dirac action (5.A.1) under the transformations (5.5.2) (5.5.4) and (5.5.5).

We first look at the expression $W\zeta$. Examination of formulae (4.A.3) and (4.A.4) show that the expression $W\zeta$ transforms as

$$W\zeta \mapsto e^{-2\varphi}W\zeta. \tag{5.5.6}$$

We see that the expression $W\zeta$ scales in a covariant way, with "covariant" meaning that the derivatives of φ do not appear in the RHS of (5.5.6). Of course, the covariance of the massless Dirac operator under conformal scalings of the metric is a know differential geometric fact: see Theorem 4.3 in [28].

Formulae (5.5.5) and (5.5.6) imply

$$\operatorname{Re}(\zeta^* W \zeta) \mapsto e^{-3\varphi} \operatorname{Re}(\zeta^* W \zeta). \tag{5.5.7}$$

Formula (5.5.2) implies $g_{\alpha\beta} \mapsto e^{2\varphi}g_{\alpha\beta}$ and, as we are working in dimension 3, this, in turn, implies that the Riemannian density scales as

$$\sqrt{\det g_{\alpha\beta}} \mapsto e^{3\varphi} \sqrt{\det g_{\alpha\beta}}$$
. (5.5.8)

Substituting formulae (5.5.7) and (5.5.8) into formula (5.A.1) we see that our massless Dirac action is invariant under the transformations (5.5.2) (5.5.4) and (5.5.5). This is, of course, in agreement with Theorem 5.1.1: as the gauge transformation (5.5.1) does not change the spectrum of our eigenvalue problem (5.1.2), it does not change the second asymptotic coefficient (5.1.13) of the counting function.

5.6 SU(2) invariance

Let us transform the reference principal symbol $B_1(x,\xi)$ as

$$B_1 \mapsto QB_1Q^*, \tag{5.6.1}$$

where $Q: M \to SU(2)$ is an arbitrary infinitely smooth special unitary matrixfunction. Formulae (5.1.14) and (5.6.1) imply

$$\sigma^{\alpha} \mapsto Q \sigma^{\alpha} Q^*. \tag{5.6.2}$$

Also, formulae (5.1.7) and (5.6.1) imply $R \mapsto RQ^*$, which can be equivalently rewritten as

$$R^* \mapsto QR^*. \tag{5.6.3}$$

Examining the structure of the matrix R, see formula (5.1.9), we conclude that formula (5.6.3) is equivalent to the linear transformation of the spinor field

$$\zeta \mapsto Q\zeta. \tag{5.6.4}$$

Formulae (5.6.2), (5.6.4) and Property 4 from Appendix 4.A of Chapter 4 tell us that our massless Dirac action is invariant under the transformation (5.6.1). This is, of course, in agreement with Theorem 5.1.1: the choice of reference principal symbol does not affect the spectrum of our eigenvalue problem (5.1.2), hence, it does not affect the second asymptotic coefficient (5.1.13) of the counting function.

5.7 Invariance under rigid rotations

Let us transform the differential operator A as

$$A \mapsto \mathbf{Q}A\mathbf{Q}^* \tag{5.7.1}$$

where

$$\mathbf{Q} = \begin{pmatrix} \mathbf{Q}_{11} & \mathbf{Q}_{12} \\ \mathbf{Q}_{21} & \mathbf{Q}_{22} \end{pmatrix}$$
(5.7.2)

is a *constant* special unitary matrix. The transformation (5.7.1) does not change the spectrum of our eigenvalue problem (5.1.2) and, moreover, preserves its structure: the principal symbol remains trace-free and the subprincipal symbol remains zero. We refer to the transformation (5.7.1) as a *rigid rotation* because it describes a rotation of the frame (5.2.4), with this rotation being the same at all points of the manifold M.

The transformation (5.7.1) is equivalent to the following transformation of the special unitary matrix-function R(x) appearing in formula (5.1.7):

$$R \mapsto \mathbf{Q}R. \tag{5.7.3}$$

Formulae (5.1.9) and (5.1.10) give us a one-to-one correspondence between special unitary matrix-functions and weight functions on the one hand and nonvanishing spinor fields on the other. In terms of the spinor field the transformation (5.7.3) reads

$$\begin{pmatrix} \zeta^{1} \\ \zeta^{2} \end{pmatrix} \mapsto \begin{pmatrix} \mathbf{Q}_{21}\overline{\zeta^{2}} + \mathbf{Q}_{22}\zeta^{1} \\ -\mathbf{Q}_{21}\overline{\zeta^{1}} + \mathbf{Q}_{22}\zeta^{2} \end{pmatrix}.$$
 (5.7.4)

Note that, unlike (5.6.3), this transformation is **not** linear because of the complex conjugation. The transformation (5.7.4) can be written as a sum of linear and antilinear transformations:

$$\zeta \mapsto \mathbf{Q}_{22}\zeta - \mathbf{Q}_{21}\mathbf{C}(\zeta) \tag{5.7.5}$$

where

$$\begin{pmatrix} \zeta^1 \\ \zeta^2 \end{pmatrix} \mapsto \mathcal{C}(\zeta) := \begin{pmatrix} -\overline{\zeta^2} \\ \overline{\zeta^1} \end{pmatrix}$$
(5.7.6)

is the charge conjugation operator, see formula (4.A.9) in Appendix 4.A.

Let us show, by performing explicit calculations, that the massless Dirac action (5.A.1) is invariant under the transformation (5.7.5). Using the fact that the massless Dirac operator commutes with the charge conjugation operator, see Property 3 in Appendix 4.A, we get

$$(\mathbf{Q}_{22}\zeta - \mathbf{Q}_{21}\mathcal{C}(\zeta))^* W (\mathbf{Q}_{22}\zeta - \mathbf{Q}_{21}\mathcal{C}(\zeta))$$

= $|\mathbf{Q}_{22}|^2 \zeta^* W \zeta + |\mathbf{Q}_{21}|^2 (\mathcal{C}(\zeta))^* \mathcal{C}(W\zeta) - \overline{\mathbf{Q}_{22}} \mathbf{Q}_{21} \zeta^* \mathcal{C}(W\zeta) - \mathbf{Q}_{22} \overline{\mathbf{Q}_{21}} (\mathcal{C}(\zeta))^* W \zeta$
= $|\mathbf{Q}_{22}|^2 \zeta^* W \zeta + |\mathbf{Q}_{21}|^2 \overline{\zeta^* W \zeta} + \{\overline{\mathbf{Q}_{22}} \mathbf{Q}_{21} (\mathcal{C}(\zeta))^T \overline{W \zeta} - \mathbf{Q}_{22} \overline{\mathbf{Q}_{21}} (\mathcal{C}(\zeta))^* W \zeta\}.$

But the expression in the curly brackets is purely imaginary, so

$$\operatorname{Re}\left[\left(\mathbf{Q}_{22}\zeta - \mathbf{Q}_{21}C(\zeta)\right)^* W\left(\mathbf{Q}_{22}\zeta - \mathbf{Q}_{21}C(\zeta)\right)\right]$$
$$= |\mathbf{Q}_{22}|^2 \operatorname{Re}\left[\zeta^* W\zeta\right] + |\mathbf{Q}_{21}|^2 \operatorname{Re}\left[\overline{\zeta^* W\zeta}\right] = \left(|\mathbf{Q}_{22}|^2 + |\mathbf{Q}_{21}|^2\right) \operatorname{Re}\left[\zeta^* W\zeta\right] = \operatorname{Re}\left[\zeta^* W\zeta\right].$$

5.A The massless Dirac action

In this appendix we define, in a concise manner, the massless Dirac action. For more details concerning the massless Dirac operator, see Appendix 4.A.

In order to write down the massless Dirac action we need Pauli matrices, i.e. a triple of trace-free Hermitian 2×2 matrix-functions $\sigma^{\alpha}(x)$, $\alpha = 1, 2, 3$, satisfying the condition (5.2.10). In our case we have Pauli matrices $\sigma^{\alpha}(x)$ readily available: these are defined in accordance with formula (5.1.14), or, equivalently, in accordance with formulae (5.2.9) and (5.2.2). Covariant Pauli matrices are defined as $\sigma_{\alpha} := g_{\alpha\beta}\sigma^{\beta}$.

The massless Dirac operator is the matrix operator (4.A.3) which acts on 2component complex-valued spinor fields $\zeta = (\zeta^1 \ \zeta^2)^T$ (such spinors are called *Weyl spinors*).

We define the massless Dirac action as

$$S(\zeta) := \int_M \operatorname{Re}(\zeta^* W\zeta) \sqrt{\det g_{\alpha\beta}} \, dx \tag{5.A.1}$$

where the star indicates Hermitian conjugation. This is the variational functional corresponding to the operator (4.A.3). Here, of course, we use the fact that in view of the self-adjointness of the operator W we have

$$\int_M \zeta^* W \zeta \sqrt{\det g_{\alpha\beta}} \, dx = \int_M (W\zeta)^* \zeta \sqrt{\det g_{\alpha\beta}} \, dx = \int_M \operatorname{Re}(\zeta^* W\zeta) \sqrt{\det g_{\alpha\beta}} \, dx$$

5.B Example

In this appendix we consider an explicit example illustrating the use of Theorem 5.1.1. Consider the unit torus \mathbb{T}^3 parameterized by cyclic coordinates x^{α} , $\alpha = 1, 2, 3$, of period 2π . Let A be the differential operator with principal symbol

$$A_1(x,\xi) = \begin{pmatrix} \xi_3 & e^{2ix^3}(\xi_1 - i\xi_2) \\ e^{-2ix^3}(\xi_1 + i\xi_2) & -\xi_3 \end{pmatrix}$$
(5.B.1)

and zero subprincipal symbol. We examine below the eigenvalue problem (5.1.2) for this particular operator A and trivial weight function (5.3.1).

Substituting (5.B.1) into (5.1.6) we see that the above principal symbol generates the Euclidean metric

$$g^{\alpha\beta}(x) = \delta^{\alpha\beta}.$$
 (5.B.2)

Hence, as the reference principal symbol it is natural to take

$$B_1(x,\xi) = \begin{pmatrix} \xi_3 & \xi_1 - i\xi_2 \\ \xi_1 + i\xi_2 & -\xi_3 \end{pmatrix}.$$
 (5.B.3)

Substituting (5.B.3) into (5.1.14) we get standard Pauli matrices

$$\sigma^{1} = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \qquad \sigma^{2} = \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix}, \qquad \sigma^{3} = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}.$$
(5.B.4)

It is not a priori obvious that the principal symbols $A_1(x,\xi)$ and $B_1(x,\xi)$ have the same spin structure. The only way to establish that they do indeed have the same spin structure is to resolve equation (5.1.7) with respect to the special unitary matrix-function R(x). Straightforward calculations give

$$R(x) = \pm \begin{pmatrix} e^{ix^3} & 0\\ 0 & e^{-ix^3} \end{pmatrix}.$$
 (5.B.5)

Of course, the underlying reasons why in this particular case we do not encounter topological obstructions are that both principal symbols have the same (positive) topological charge and that the frame encoded in (5.B.1) makes an even number of turns (two turns) as x^3 runs from 0 to 2π . See Appendix 4.A for more details. Formulae (5.1.9), (5.1.10), (5.3.1) and (5.B.5) give us the following expression for the spinor field:

$$\zeta(x) = \pm \begin{pmatrix} e^{-ix^3} \\ 0 \end{pmatrix}.$$
 (5.B.6)

Substituting formulae (5.B.2), (5.B.4) and (5.B.6) into (4.A.3) and (5.A.1) we get $S(\zeta) = -(2\pi)^3$. Hence, Theorem 5.1.1 tells us that in our example the two-term asymptotics (5.1.5) takes the form

$$N(\lambda) = \frac{4}{3}\pi\lambda^3 - 4\pi\lambda^2 + o(\lambda^2)$$
(5.B.7)

as $\lambda \to +\infty$. Note that the nonperiodicity condition (see Definitions 8.3 and 8.4 in [22]) is fulfilled in our example, so, according to Theorem 8.4 from [22], the asymptotic formula (5.B.7) holds as it is, without mollification.

Observe now that in our example the spectrum of the operator A can be evaluated explicitly. Indeed, let B be the differential operator with principal symbol (5.B.3) and zero subprincipal symbol. In other words, let $B = B_1(x, -i\partial/\partial x)$. Put $\tilde{A} := RBR^*$, where R is the matrix-function (5.B.5) It is easy to check that the subprincipal symbol of the operator \tilde{A} is -I, where I is the identity matrix. Hence,

$$A = RBR^* + I. \tag{5.B.8}$$

But the operator RBR^* is unitarily equivalent to the operator B and the spectrum of B is known, see Appendix 4.B. Using (5.B.8), we conclude that the eigenvalues of our operator A are as follows.

- 1 is an eigenvalue of multiplicity two.
- For each m ∈ Z³ \ {0} we have the eigenvalue 1 + ||m|| and unique (up to rescaling) eigenfunction, with eigenfunctions corresponding to different m being linearly independent.

For each m ∈ Z³ \ {0} we have the eigenvalue 1 − ||m|| and unique (up to rescaling) eigenfunction, with eigenfunctions corresponding to different m being linearly independent.

Thus, $N(\lambda) - 1$ is the number of integer lattice points inside a 2-sphere of radius $\lambda - 1$ in \mathbb{R}^3 centred at the origin. Applying the result from [38] we get

$$N(\lambda) = \frac{4}{3}\pi\lambda^3 - 4\pi\lambda^2 + O_{\varepsilon}(\lambda^{21/16+\varepsilon})$$
(5.B.9)

as $\lambda \to +\infty$, with ε being an arbitrary positive number. The more advanced number theoretical result (5.B.9) agrees with our asymptotic formula (5.B.7).

Note that the calculations presented in this section remain unchanged if we replace everywhere $\xi_1 \mp i\xi_2$ by $\xi_1 \pm i\xi_2$. This is in agreement with the fact that the topological charge **c** does not appear in our formula (5.1.13).

Chapter 6

Spectral asymmetry of the massless Dirac operator

6.1 Introduction

Let M be a 3-dimensional connected compact oriented manifold without boundary equipped with a smooth Riemannian metric $g_{\alpha\beta}$, α , $\beta = 1, 2, 3$ being the tensor indices. Let W be the corresponding massless Dirac operator, see Appendix 4.A for definition. There are two basic examples when the spectrum of W can be calculated explicitly. The first is the unit torus \mathbb{T}^3 equipped with Euclidean metric. The second is the unit sphere \mathbb{S}^3 equipped with metric induced by the natural embedding of \mathbb{S}^3 in Euclidean space \mathbb{R}^4 . In both examples the spectrum turns out to be symmetric about zero, see Appendix 4.B for details. Physically, this means that in these two examples there is no difference between the properties of the particle (massless neutrino) and antiparticle (massless antineutrino).

As pointed out in [4, 5, 6, 7], for a general oriented Riemannian 3-manifold (M, g) there is no reason for the spectrum of the massless Dirac operator W to be symmetric. However, producing explicit examples of spectral asymmetry is a difficult task. To our knowledge, the only explicit example was constructed in [65], with the example based on the idea of choosing a 3-manifold with flat metric

but highly nontrivial topology. In this chapter we take a different route: we stick with the simplest possible topology (torus) and create spectral asymmetry by perturbing the metric.

Further on in this chapter we work on the unit torus \mathbb{T}^3 parameterized by cyclic coordinates x^{α} , $\alpha = 1, 2, 3$, of period 2π .

Suppose first that the metric is Euclidean. Then the massless Dirac operator corresponding to the standard spin structure (see formula (4.A.16)) reads

$$W = -i \begin{pmatrix} \frac{\partial}{\partial x^3} & \frac{\partial}{\partial x^1} - i\frac{\partial}{\partial x^2} \\ \frac{\partial}{\partial x^1} + i\frac{\partial}{\partial x^2} & -\frac{\partial}{\partial x^3} \end{pmatrix}.$$
 (6.1.1)

The operator (6.1.1) admits separation of variables, i.e. one can seek its eigenfunctions in the form $v(x) = ue^{im_{\alpha}x^{\alpha}}$, $m \in \mathbb{Z}^3$, $u \in \mathbb{C}^2$, $u \neq 0$, and calculate the eigenvalues and eigenfunctions explicitly. The spectrum of the operator (6.1.1) is as follows.

- Zero is an eigenvalue of multiplicity two.
- For each $m \in \mathbb{Z}^3 \setminus \{0\}$ we have the eigenvalue ||m|| and unique (up to rescaling) eigenfunction of the form $ue^{im_{\alpha}x^{\alpha}}$.
- For each $m \in \mathbb{Z}^3 \setminus \{0\}$ we have the eigenvalue -||m|| and unique (up to rescaling) eigenfunction of the form $ue^{im_{\alpha}x^{\alpha}}$.

We now perturb the metric, i.e. consider a metric $g_{\alpha\beta}(x;\epsilon)$ the components of which are smooth functions of coordinates x^{α} , $\alpha = 1, 2, 3$, and small real parameter ϵ , and which satisfies

$$g_{\alpha\beta}(x;0) = \delta_{\alpha\beta}. \tag{6.1.2}$$

One way of establishing spectral asymmetry of the perturbed problem is to compare the asymptotic distribution of large positive eigenvalues and large negative eigenvalues. As explained in Section 10 of [22], for a generic first order differential operator this approach allows one to establish spectral asymmetry. Unfortunately, the massless Dirac operator is very special in that the second asymptotic coefficient of its counting function is zero, see formula (4.1.17), so in the first two approximations in powers of λ its large positive eigenvalues are distributed the same way as its large negative eigenvalues. Therefore, in order to demonstrate spectral asymmetry of the perturbed problem, we will, instead of dealing with large eigenvalues, deal with small eigenvalues.

6.2 Main result

Let $W(\epsilon)$ be the massless Dirac operator corresponding to the metric $g_{\alpha\beta}(x;\epsilon)$. The difficulty with applying standard perturbation techniques to the operator $W(\epsilon)$ is that all its eigenvalues have even multiplicity, this being a consequence of the fact that the massless Dirac operator $W(\epsilon)$ commutes with the antilinear operator of charge conjugation

$$v = \begin{pmatrix} v_1 \\ v_2 \end{pmatrix} \mapsto \begin{pmatrix} -\overline{v_2} \\ \overline{v_1} \end{pmatrix} =: \mathcal{C}(v), \tag{6.2.1}$$

see Property 3 in Appendix 4.A. In order to overcome this difficulty we develop in Sections 6.3–6.5 a perturbation theory for the massless Dirac operator which accounts for this charge conjugation symmetry. We show that perturbation-wise the double eigenvalues of the massless Dirac operator can be treated as if they were simple eigenvalues: under perturbation a double eigenvalue remains¹ a double eigenvalue and all the usual formulae apply, with only one minor modification. The minor modification concerns the definition of the pseudoinverse of the unperturbed operator, see formulae (6.3.8)–(6.3.12). Namely, in the definition of

¹Here, of course, it is important that we don't have a magnetic field. A magnetic field would split up a double eigenvalue, see [28]. The fact that the massless Dirac operator and the charge conjugation operator do not commute in the presence of a magnetic covector potential is well known in theoretical physics: see, for example, formula (2.5) in [41].

the pseudoinverse we separate out a two-dimensional eigenspace rather than a one-dimensional eigenspace.

Given a function $f: \mathbb{T}^3 \to \mathbb{C}$, we denote by

$$\hat{f}(m) := \frac{1}{(2\pi)^3} \int_{\mathbb{T}^3} e^{-im_\alpha x^\alpha} f(x) \, dx \,, \qquad m \in \mathbb{Z}^3,$$
 (6.2.2)

its Fourier coefficients. Here $dx := dx^1 dx^2 dx^3$.

Let $\lambda_0(\varepsilon)$ be the eigenvalue of the massless Dirac operator with smallest modulus and let

$$h_{\alpha\beta}(x) := \left. \frac{\partial g_{\alpha\beta}}{\partial \epsilon} \right|_{\epsilon=0}.$$
(6.2.3)

Further on we raise and lower tensor indices using the Euclidean metric, which means that raising or lowering a tensor index doesn't change anything. A repeated tensor index always indicates summation over the values 1, 2, 3.

The following theorem is the main result of this chapter.

Theorem 6.2.1. We have

$$\lambda_0(\epsilon) = c \,\epsilon^2 + O(\epsilon^3) \quad as \quad \epsilon \to 0, \tag{6.2.4}$$

where the constant c is given by the formula

$$c = \frac{i}{16} \varepsilon_{\alpha\beta\gamma} \sum_{m \in \mathbb{Z}^3 \setminus \{0\}} \left(\delta_{\mu\nu} - \frac{m_{\mu}m_{\nu}}{\|m\|^2} \right) m_{\alpha} \,\hat{h}_{\beta\mu}(m) \,\overline{\hat{h}_{\gamma\nu}(m)} \,. \tag{6.2.5}$$

Here $\varepsilon_{\alpha\beta\gamma}$ is the totally antisymmetric quantity, $\varepsilon_{123} := +1$, and the overline stands for complex conjugation.

Theorem 6.2.1 warrants the following remarks.

• If the constant c defined by formula (6.2.5) is nonzero, then Theorem 6.2.1 tells us that for sufficiently small nonzero ϵ the spectrum of our massless Dirac operator is asymmetric about zero.

- Theorem 6.2.1 is in agreement with the established view [40, 9] that there are no topological obstructions preventing the shift of the zero eigenvalue of the massless Dirac operator.
- Theorem 6.2.1 is in agreement with the results of [2]. This chapter deals with the Dirac operator in the most general setting. When applied to the case of a compact oriented Riemannian 3-manifold (not necessarily a 3-torus with Euclidean metric) with specified spin structure the results of [2] tell us that if zero is an eigenvalue of the Dirac operator, then the metric can be perturbed so that the zero eigenvalue gets shifted. Furthermore, according to [1], the zero eigenvalue can be shifted by perturbing the metric on an arbitrarily small open set, which is also in agreement with our Theorem 6.2.1.

• Put

$$L_{\gamma\nu\beta\mu} := \frac{i\varepsilon_{\alpha\beta\gamma}}{(2\pi)^3} \sum_{m\in\mathbb{Z}^3\backslash\{0\}} \left(\delta_{\mu\nu} - \frac{m_{\mu}m_{\nu}}{\|m\|^2} \right) m_{\alpha} \int_{\mathbb{T}^3} e^{i(x-y)^{\alpha}m_{\alpha}} (\cdot) dy ,$$
$$P_{\gamma\nu\beta\mu} := \frac{1}{4} (L_{\gamma\nu\beta\mu} + L_{\nu\gamma\beta\mu} + L_{\gamma\nu\mu\beta} + L_{\nu\gamma\mu\beta}).$$

This gives us a first order pseudodifferential operator P acting in the vector space of rank two symmetric complex-valued tensor fields, $s_{\beta\mu} \mapsto P_{\gamma\nu\beta\mu}s_{\beta\mu}$. If we equip this vector space with the natural inner product (r, s) := $\int_{\mathbb{T}^3} r_{\alpha\beta} \overline{s_{\alpha\beta}} dx$ then it is easy to see that the operator P is formally selfadjoint and formula (6.2.5) can be rewritten as $c = \frac{1}{128\pi^3}(Ph, h)$, where h is defined in accordance with (6.2.3). This shows that our coefficient c has a nonlocal (global) nature, with the source of the nonlocality being the factor

$$\delta_{\mu\nu} - \frac{m_{\mu}m_{\nu}}{\|m\|^2} \tag{6.2.6}$$

in the symbol of the pseudodifferential operator P. In other words, formula (6.2.5) cannot be rewritten in terms of (linearized) local differential geometric quantities such as the curvature tensor and the Cotton tensor.

- The rank two tensor (6.2.6) can be identified with a linear map in \mathbb{R}^3 , $p_{\mu} \mapsto \left(\delta_{\mu\nu} - \frac{m_{\mu}m_{\nu}}{\|m\|^2}\right) p_{\nu}$. This linear map is an orthogonal projection: it projects onto the plane orthogonal to the covector (momentum) m.
- Suppose that we are looking at a conformal scaling of the Euclidean metric, $g_{\alpha\beta}(x;\epsilon) = e^{2\epsilon\varphi(x)}\delta_{\alpha\beta}$, where $\varphi : \mathbb{T}^3 \to \mathbb{R}$. Then $h_{\alpha\beta}(x) = 2\varphi(x)\delta_{\alpha\beta}$ and formula (6.2.5) becomes

$$c = \frac{i}{4} \varepsilon_{\alpha\beta\gamma} \sum_{m \in \mathbb{Z}^3 \setminus \{0\}} \left(\delta_{\mu\nu} - \frac{m_{\mu}m_{\nu}}{\|m\|^2} \right) m_{\alpha} \delta_{\beta\mu} \delta_{\gamma\nu} \left| \hat{\varphi}(m) \right|^2 .$$
 (6.2.7)

The expression in the RHS of (6.2.7) is zero because the summand in $\sum_{m \in \mathbb{Z}^3 \setminus \{0\}}$ is odd in m. (Another reason why the expression in the RHS of (6.2.7) is zero is that the summand is symmetric in β , γ .) This agrees with the well-known fact that the zero eigenvalue does not shift under a conformal scaling of the metric, see Theorem 4.3 in [28].

• Suppose that we replace the tensor $h_{\alpha\beta}(x)$ by the tensor $h_{\alpha\beta}(-x)$. Then $\hat{h}_{\alpha\beta}(m)$ is replaced by $-\hat{h}_{\alpha\beta}(-m)$ and, introducing a new summation index n := -m in formula (6.2.5), we see that the coefficient c changes sign. Physically, this means that formula (6.2.5) feels the difference between "left" and "right", as one would expect of a formula describing a fermion.

The proof of Theorem 6.2.1 is given in Section 6.6. In Section 6.7 we treat the special case when the metric $g_{\alpha\beta}(x;\epsilon)$ is a function of the coordinate x^1 only. In Sections 6.8 and 6.9 we present families of metrics for which the eigenvalue $\lambda_0(\epsilon)$ can be evaluated explicitly. Finally, in Section 6.10 we examine the eta invariant of our ϵ -dependent massless Dirac operator.

6.3 Perturbation process I: preliminaries

Let M be a 3-dimensional connected compact oriented manifold without boundary equipped with a smooth Riemannian metric $g_{\alpha\beta}(x)$, $\alpha, \beta = 1, 2, 3$ being the tensor indices and $x = (x^1, x^2, x^3)$ being local coordinates. The perturbation theory developed in this section and Sections 6.4–6.5 does not assume that the manifold is necessarily a 3-torus.

We perturb the metric in a smooth manner and denote the perturbed metric by $g_{\alpha\beta}(x;\epsilon)$, where ϵ is a small real parameter. Here we assume that $g_{\alpha\beta}(x;0)$ is the unperturbed metric described in the previous paragraph.

By $W_{1/2}(\epsilon)$ we denote the massless Dirac operator on half-densities corresponding to the metric $g_{\alpha\beta}(x;\epsilon)$, see Appendix 4.A for details. We choose to work with the massless Dirac operator on half-densities $W_{1/2}(\epsilon)$ rather than with the massless Dirac operator $W(\epsilon)$ because we do not want our Hilbert space to depend on ϵ . The difference between the operators $W(\epsilon)$ and $W_{1/2}(\epsilon)$ is explained in Appendix 4.A: compare formulae (4.A.3) and (4.A.19). The spectra of the operators $W(\epsilon)$ and $W_{1/2}(\epsilon)$ are the same.

The operator $W_{1/2}(\epsilon)$ is actually not a single operator, but an equivalence class of operators which differ by the transformation

$$W_{1/2}(\epsilon) \mapsto R W_{1/2}(\epsilon) R^*, \tag{6.3.1}$$

where $R(x; \epsilon)$ is an arbitrary smooth 2 × 2 special unitary matrix-function. See Property 4 in Appendix 4.A for a detailed discussion regarding the transformation (6.3.1), noting that the massless Dirac operator on half-densities $W_{1/2}(\epsilon)$ differs from the massless Dirac operator $W(\epsilon)$ only by "scalar" factors on the left and on the right — these "scalar" factors commute with matrix-functions $R(x; \epsilon)$ and $R^*(x; \epsilon)$. Obviously, the transformation (6.3.1) does not affect the spectrum. Later on, in Section 6.6, we will use this gauge degree of freedom to simplify calculations, see formula (6.6.11).

The operator $W_{1/2}(\epsilon)$ acts on 2-columns $v = \begin{pmatrix} v_1 \\ v_2 \end{pmatrix}$ of complex-valued halfdensities. Our Hilbert space is $L^2(M; \mathbb{C}^2)$, which is the vector space of 2-columns of square integrable half-densities equipped with inner product

$$\langle v, w \rangle := \int_M w^* v \, dx \,.$$
 (6.3.2)

The domain of the operator $W_{1/2}(\epsilon)$ is $H^1(M; \mathbb{C}^2)$, which is the Sobolev space of 2columns of half-densities that are square integrable together with their first partial derivatives. It is known that the operator $W_{1/2}(\epsilon) : H^1(M; \mathbb{C}^2) \to L^2(M; \mathbb{C}^2)$ is self-adjoint and that it has a discrete spectrum, with eigenvalues accumulating to $+\infty$ and $-\infty$. Note that here neither the Hilbert space nor the domain depend on ϵ . It is also known that the eigenfunctions of the operator $W_{1/2}(\epsilon)$ are infinitely smooth.

The antilinear operator of charge conjugation (6.2.1) maps any element of $L^2(M; \mathbb{C}^2)$ to an element of $L^2(M; \mathbb{C}^2)$ and any element of $H^1(M; \mathbb{C}^2)$ to an element of $H^1(M; \mathbb{C}^2)$. As the massless Dirac operator on half-densities $W_{1/2}(\epsilon)$ differs from the massless Dirac operator $W(\epsilon)$ only by real "scalar" factors on the left and on the right, it also commutes with the operator of charge conjugation:

$$C(W_{1/2}(\epsilon) v) = W_{1/2}(\epsilon) C(v),$$
 (6.3.3)

 $\forall v \in H^1(M; \mathbb{C}^2)$. Note that the operator of charge conjugation does not itself depend on ϵ .

Observe that formulae (6.2.1) and (6.3.2) imply the following useful identities:

$$C(C(v)) = -v, \qquad (6.3.4)$$

$$\langle v, \mathcal{C}(v) \rangle = 0, \tag{6.3.5}$$

$$\langle \mathbf{C}(v), \mathbf{C}(w) \rangle = \langle w, v \rangle.$$
 (6.3.6)

Let

$$W_{1/2}(\epsilon) = W_{1/2}^{(0)} + \epsilon W_{1/2}^{(1)} + \epsilon^2 W_{1/2}^{(2)} + \dots$$
(6.3.7)

be the asymptotic expansion of the partial differential operator $W_{1/2}(\epsilon)$ in powers of the small parameter ϵ . Obviously, the operators $W_{1/2}^{(k)}$, k = 0, 1, 2, ..., are formally self-adjoint first order differential operators which commute with the antilinear operator of charge conjugation (6.2.1).

Suppose that $\lambda^{(0)}$ is a double eigenvalue of the operator $W_{1/2}^{(0)}$. As explained in Appendix 4.A, eigenvalues of the massless Dirac operator have even multiplicity, so a double eigenvalue is the "simplest" eigenvalue one can get.

Remark 6.3.1. The spectrum of the massless Dirac operator on a 3-torus equipped with Euclidean metric was written down explicitly in Section 6.1. Examination of the relevant formulae shows that the only double eigenvalue is the eigenvalue zero as all others have multiplicity greater than or equal to six. However, in this section and Sections 6.4–6.5 we do not use the fact that $\lambda^{(0)} = 0$.

Let $v^{(0)}$ be a normalized, $||v^{(0)}|| = 1$, eigenfunction of the operator $W_{1/2}^{(0)}$ corresponding to the eigenvalue $\lambda^{(0)}$. Formula (6.3.3) and the fact that $\lambda^{(0)}$ is real imply that $C(v^{(0)})$ is also an eigenfunction of the operator $W_{1/2}^{(0)}$ corresponding to the eigenvalue $\lambda^{(0)}$. Formula (6.3.6) implies that $||C(v^{(0)})|| = 1$, and, moreover, in view of formula (6.3.5), the eigenfunctions $v^{(0)}$ and $C(v^{(0)})$ are orthogonal.

The argument presented in the previous paragraph shows that, when dealing with a double eigenvalue of the massless Dirac operator, it is sufficient to construct only one eigenfunction: the other one is obtained by charge conjugation. The argument is valid not only for the unperturbed operator $W_{1/2}^{(0)}$, but for the perturbed operator $W_{1/2}(\epsilon)$ as well, provided that ϵ is small enough (so that the multiplicity of the eigenvalue does not increase). Hence, in the perturbation process described in the next section we shall construct one eigenfunction only.

In the perturbation process that we will describe in the next section we will make use of the pseudoinverse of the unperturbed operator. This operator, which we denote by Q, is defined as follows. Consider the problem

$$(W_{1/2}^{(0)} - \lambda^{(0)})v = f \tag{6.3.8}$$

where $f \in L^2(M; \mathbb{C}^2)$ is given and $v \in H^1(M; \mathbb{C}^2)$ is to be found. Suppose that f satisfies the conditions

$$\langle f, v^{(0)} \rangle = \langle f, \mathcal{C}(v^{(0)}) \rangle = 0.$$
 (6.3.9)

Then the problem (6.3.8) can be resolved for v, however this solution is not unique. We achieve uniqueness by imposing the conditions

$$\langle v, v^{(0)} \rangle = \langle v, \mathcal{C}(v^{(0)}) \rangle = 0 \tag{6.3.10}$$

and define Q as the linear operator mapping f to v,

$$Q: f \mapsto v. \tag{6.3.11}$$

Thus, Q is a bounded linear operator acting in the orthogonal complement of the eigenspace of the operator $W_{1/2}^{(0)}$ corresponding to the eigenvalue $\lambda^{(0)}$. We extend this operator to the whole Hilbert space $L^2(M; \mathbb{C}^2)$ in accordance with

$$Qv^{(0)} = QC(v^{(0)}) = 0. (6.3.12)$$

It is clear from the above definition that the bounded linear operator Q is selfadjoint and commutes with the antilinear operator of charge conjugation (6.2.1). Note that our definition of the pseudoinverse Q of the unperturbed operator $W_{1/2}^{(0)} - \lambda^{(0)}$ is in agreement with Rellich's, see Chapter 2 Section 2 in [66].

Throughout our perturbation process we will have to deal with various formally self-adjoint linear operators which commute with the antilinear operator of charge conjugation (6.2.1). Such operators possess a special property which is the subject of the following lemma.

Lemma 6.3.1. Let $L : C^{\infty}(M; \mathbb{C}^2) \to C^{\infty}(M; \mathbb{C}^2)$ be a (possibly unbounded) formally self-adjoint linear operator which commutes with the antilinear operator of charge conjugation (6.2.1). Then for any $v \in C^{\infty}(M; \mathbb{C}^2)$ we have

$$\langle Lv, \mathcal{C}(v) \rangle = 0. \tag{6.3.13}$$

Proof Take arbitrary $v, w \in C^{\infty}(M; \mathbb{C}^2)$. Using formula (6.3.6) and the fact that L is formally self-adjoint and commutes with C, we get

$$\langle L C(w), C(v) \rangle = \langle C(Lw), C(v) \rangle = \langle v, Lw \rangle = \langle Lv, w \rangle.$$
(6.3.14)

For w = C(v) formula (6.3.14) reads

$$\langle L C(C(v)), C(v) \rangle = \langle Lv, C(v) \rangle.$$
 (6.3.15)

But in view of (6.3.4) formula (6.3.15) can be rewritten as

$$-\langle Lv, \mathcal{C}(v) \rangle = \langle Lv, \mathcal{C}(v) \rangle,$$

which gives us the required identity (6.3.13).

6.4 Perturbation process II: formal procedure

We now write down the formal perturbation process. A rigorous justification will be provided in the next section.

Further on in this section as well as in the two following sections (Sections 6.5 and 6.6) we write, for the sake of brevity, $A(\epsilon) = W_{1/2}(\epsilon)$ and $A^{(k)} = W_{1/2}^{(k)}$, $k = 0, 1, 2, \ldots$ In this new notation formula (6.3.7) reads

$$A(\epsilon) = A^{(0)} + \epsilon A^{(1)} + \epsilon^2 A^{(2)} + \dots$$
(6.4.1)

We need to solve the eigenvalue problem

$$A(\epsilon) v(\epsilon) = \lambda(\epsilon) v(\epsilon). \qquad (6.4.2)$$

We seek the eigenvalue and eigenfunction of the perturbed operator $A(\epsilon)$ in the form of asymptotic expansions

$$\lambda(\epsilon) = \lambda^{(0)} + \epsilon \lambda^{(1)} + \epsilon^2 \lambda^{(2)} + \dots, \qquad (6.4.3)$$

$$v(\epsilon) = v^{(0)} + \epsilon v^{(1)} + \epsilon^2 v^{(2)} + \dots$$
(6.4.4)

Note that we do not aim to preserve the normalization of our eigenfunction throughout the perturbation process.

Let us forget for a moment that we are dealing with a double eigenvalue and suppose that our eigenvalue is simple. Then the iterative procedure for the determination of $\lambda^{(k)}$ and $v^{(k)}$, k = 1, 2, ..., is well known, see Chapter 2 Section 2 in [66]. At the *k*th step we get the equation

$$(A^{(0)} - \lambda^{(0)})v^{(k)} = f^{(k)}, \qquad (6.4.5)$$

where

$$f^{(k)} := F^{(k)} v^{(0)}, (6.4.6)$$

and $F^{(k)}$ is some linear operator. The explicit formula for the operator $F^{(k)}$ appearing in equations (6.4.5), (6.4.6) is written as follows. Put

$$D(\epsilon) := (B(0) - B(\epsilon)) \left(I + \sum_{j=1}^{\infty} \left[Q \left(B(0) - B(\epsilon) \right) \right]^j \right), \tag{6.4.7}$$

where I is the identity operator, $B(\epsilon) := A(\epsilon) - \lambda(\epsilon)I$ and the infinite sum is understood as an asymptotic series. The operator $D(\epsilon)$ can be expanded in powers of the small parameter ϵ ,

$$D(\epsilon) = \sum_{k=1}^{\infty} \epsilon^k F^{(k)}, \qquad (6.4.8)$$

giving us the required $F^{(k)}$. The real number $\lambda^{(k)}$ is determined from the condition

$$\langle f^{(k)}, v^{(0)} \rangle = 0$$
 (6.4.9)

after which we resolve (6.4.5) by setting

$$v^{(k)} = Qf^{(k)}. (6.4.10)$$

We claim that the above process carries over to the case of a double eigenvalue that we are dealing with. Indeed, the difference between the cases of a simple eigenvalue and a double eigenvalue is that at the kth step of the iterative process in addition to condition (6.4.9) we need to satisfy the condition

$$\langle f^{(k)}, \mathcal{C}(v^{(0)}) \rangle = 0.$$
 (6.4.11)

The structure of the operator (6.4.7) is such that it is formally self-adjoint and commutes with the antilinear operator of charge conjugation (6.2.1), so the operator $F^{(k)}$ defined in accordance with formula (6.4.8) has the same properties and, hence, by Lemma 6.3.1, condition (6.4.11) is satisfied automatically and the asymptotic process continues as if the eigenvalue were simple.

We end this section by giving, for future reference, the explicit formulae for the coefficients $\lambda^{(1)}$ and $\lambda^{(2)}$ appearing in the asymptotic expansion (6.4.3):

$$\lambda^{(1)} = \langle A^{(1)} v^{(0)}, v^{(0)} \rangle, \tag{6.4.12}$$

$$\lambda^{(2)} = \langle A^{(2)} v^{(0)}, v^{(0)} \rangle - \langle (A^{(1)} - \lambda^{(1)}) Q (A^{(1)} - \lambda^{(1)}) v^{(0)}, v^{(0)} \rangle.$$
 (6.4.13)

6.5 Perturbation process III: justification

Recall that by $\lambda^{(0)} = \lambda(0)$ we denote a double eigenvalue of the unperturbed operator $A^{(0)} = A(0)$ (the unperturbed massless Dirac operator on half-densities). Let us choose a $\delta > 0$ such that $\lambda^{(0)}$ is the only eigenvalue of the operator $A^{(0)}$ on the interval $[\lambda^{(0)} - \delta, \lambda^{(0)} + \delta]$.

In order to justify our perturbation process we will need the following lemma.

Lemma 6.5.1. For sufficiently small ϵ the interval

$$(\lambda^{(0)} - \delta, \lambda^{(0)} + \delta) \tag{6.5.1}$$

contains exactly one double eigenvalue of the operator $A(\epsilon)$ and no other eigenvalues.

Proof Denote $C_{\delta} := \{\mu \in \mathbb{C} \mid |\mu - \lambda^{(0)}| = \delta\}$ (circle in the complex plane) and $D_{\delta} := \{\mu \in \mathbb{C} \mid |\mu - \lambda^{(0)}| < \delta\}$ (open disc in the complex plane). Put $R_{\mu}^{(0)} := (A^{(0)} - \mu I)^{-1}$. Clearly, for $\mu \in C_{\delta}$ the operator $R_{\mu}^{(0)}$ is well-defined and, moreover, is a bounded operator acting from $L^2(M; \mathbb{C}^2)$ to $H^1(M; \mathbb{C}^2)$. Furthermore, the norm of the operator $R_{\mu}^{(0)} : L^2(M; \mathbb{C}^2) \to H^1(M; \mathbb{C}^2)$ is bounded uniformly over $\mu \in C_{\delta}$.

Let us now define the operator

$$R_{\mu}(\epsilon) := \left(I + \sum_{j=1}^{\infty} \left[-R_{\mu}^{(0)}(A(\epsilon) - A^{(0)})\right]^{j}\right) R_{\mu}^{(0)}, \qquad (6.5.2)$$

where $\mu \in C_{\delta}$. The operator $A(\epsilon) - A^{(0)}$ is a bounded operator acting from $H^1(M; \mathbb{C}^2)$ to $L^2(M; \mathbb{C}^2)$ and the norm of the operator $A(\epsilon) - A^{(0)}$: $H^1(M; \mathbb{C}^2) \to L^2(M; \mathbb{C}^2)$ tends to zero as ϵ tends to zero. Hence, the series in (6.5.2) converges for sufficiently small ϵ . Furthermore, it is easy to see that

$$R_{\mu}(\epsilon) \to R_{\mu}^{(0)} \quad \text{as} \quad \epsilon \to 0$$
 (6.5.3)

in the sense of the operator norm $L^2(M; \mathbb{C}^2) \to H^1(M; \mathbb{C}^2)$ and this convergence is uniform over $\mu \in C_{\delta}$.

Acting onto (6.5.2) with the operator $A(\epsilon) - \mu I$ we see that $(A(\epsilon) - \mu I)R_{\mu}(\epsilon) = I$, so $R_{\mu}(\epsilon) = (A(\epsilon) - \mu I)^{-1}$. Put

$$E(\epsilon) := \frac{1}{2\pi i} \int_{C_{\delta}} R_{\mu}(\epsilon) \, d\mu \,. \tag{6.5.4}$$

The operator $E(\epsilon)$ is the orthogonal projection onto the span of eigenvectors of the operator $A(\epsilon)$ corresponding to eigenvalues on the interval (6.5.1). In particular, $E(0) = E^{(0)}$ is the orthogonal projection onto the span of eigenvectors of the operator $A^{(0)}$ corresponding to the double eigenvalue $\lambda^{(0)}$.

Formulae (6.5.3) and (6.5.4) imply

$$||E(\epsilon) - E^{(0)}||_{\text{op}} \to 0 \quad \text{as} \quad \epsilon \to 0, \tag{6.5.5}$$

where $\|\cdot\|_{\text{op}}$ stands for the operator norm in the Banach space of bounded linear operators $L^2(M; \mathbb{C}^2) \to L^2(M; \mathbb{C}^2)$. Formula (6.5.5) implies that for sufficiently small ϵ we have

$$||E(\epsilon) - E^{(0)}||_{\rm op} < 1.$$
(6.5.6)

Formula (6.5.6) and the fact that the orthogonal projections $E(\epsilon)$ and $E^{(0)}$ have finite rank imply that rank $E(\epsilon) = \operatorname{rank} E^{(0)} = 2$. Thus, the operator $A(\epsilon)$ has two eigenvalues, counted with multiplicities, on the interval (6.5.1). We know, see Property 3 in Appendix 4.A, that the eigenvalues of the operator $A(\epsilon)$ have even multiplicity, so we are looking at one double eigenvalue on the interval (6.5.1). \Box

Let $\lambda(\epsilon)$ be the unique double eigenvalue of the operator $A(\epsilon)$ from the interval (6.5.1). Denote by $\sigma(\epsilon)$ the spectrum of the operator $A(\epsilon)$ and, for a given $\mu \in \mathbb{R}$, denote dist $(\mu, \sigma(\epsilon)) = \min_{\nu \in \sigma(\epsilon)} |\mu - \nu|$. Obviously, without additional information

on μ and on $\sigma(\epsilon)$ we can only guarantee the inequality

$$\operatorname{dist}(\mu, \sigma(\epsilon)) \le |\mu - \lambda(\epsilon)|. \tag{6.5.7}$$

Choose an arbitrary natural k and denote

$$\tilde{\lambda}(\epsilon) = \lambda^{(0)} + \epsilon \lambda^{(1)} + \epsilon^2 \lambda^{(2)} + \ldots + \epsilon^k \lambda^{(k)}, \qquad (6.5.8)$$

$$\tilde{v}(\epsilon) = v^{(0)} + \epsilon v^{(1)} + \epsilon^2 v^{(2)} + \dots + \epsilon^k v^{(k)}, \qquad (6.5.9)$$

where the $\lambda^{(j)}$ and $v^{(j)}$, j = 0, 1, ..., k, are taken from (6.4.3) and (6.4.4). We have

$$\|\tilde{v}(\epsilon)\| = 1 + O(\epsilon),$$
 (6.5.10)

$$\|(A(\epsilon) - \tilde{\lambda}(\epsilon))\tilde{v}(\epsilon)\| = O(\epsilon^{k+1}), \qquad (6.5.11)$$

where $\|\cdot\|$ stands for the $L^2(M; \mathbb{C}^2)$ norm (see (6.3.2) for inner product). As our operator $A(\epsilon)$ is self-adjoint, formulae (6.5.10) and (6.5.11) imply

$$\operatorname{dist}(\tilde{\lambda}(\epsilon), \sigma(\epsilon)) \leq \frac{\|(A(\epsilon) - \tilde{\lambda}(\epsilon))\tilde{v}(\epsilon)\|}{\|\tilde{v}(\epsilon)\|} = O(\epsilon^{k+1}).$$
(6.5.12)

Formulae (6.5.8) and (6.5.12) and Lemma 6.5.1 imply that for sufficiently small ϵ ,

$$\operatorname{dist}(\tilde{\lambda}(\epsilon), \sigma(\epsilon)) = |\tilde{\lambda}(\epsilon) - \lambda(\epsilon)|, \qquad (6.5.13)$$

compare with (6.5.7). Combining formulae (6.5.12) and (6.5.13), we get $\lambda(\epsilon) = \tilde{\lambda}(\epsilon) + O(\epsilon^{k+1})$. This completes the justification of our perturbation process.

6.6 Proof of Theorem 6.2.1

The unperturbed massless Dirac operator on half-densities, which we denote by $A^{(0)}$, is given by the expression in the RHS of (6.1.1). The unperturbed eigenvalue,

 $\lambda^{(0)}$, is zero and the corresponding normalized eigenfunction is

$$v^{(0)} = \frac{1}{(2\pi)^{3/2}} \begin{pmatrix} 1\\ 0 \end{pmatrix}.$$
 (6.6.1)

The pseudoinverse Q of the operator $A^{(0)}$ is given by the formula

$$Q = \frac{1}{(2\pi)^3} \sum_{m \in \mathbb{Z}^3 \setminus \{0\}} e^{im_{\alpha}x^{\alpha}} \begin{pmatrix} m_3 & m_1 - im_2 \\ m_1 + im_2 & -m_3 \end{pmatrix}^{-1} \int_{\mathbb{T}^3} e^{-im_{\alpha}y^{\alpha}} (\cdot) dy$$
$$= \frac{1}{(2\pi)^3} \sum_{m \in \mathbb{Z}^3 \setminus \{0\}} \frac{e^{im_{\alpha}x^{\alpha}}}{\|m\|^2} \begin{pmatrix} m_3 & m_1 - im_2 \\ m_1 + im_2 & -m_3 \end{pmatrix} \int_{\mathbb{T}^3} e^{-im_{\alpha}y^{\alpha}} (\cdot) dy, \quad (6.6.2)$$

where $dy := dy^1 dy^2 dy^3$. The operator (6.6.2) is a self-adjoint pseudodifferential operator of order -1.

We have

$$\lambda(\epsilon) = \epsilon \lambda^{(1)} + \epsilon^2 \lambda^{(2)} + O(\epsilon^3), \qquad (6.6.3)$$

where the coefficients $\lambda^{(1)}$ and $\lambda^{(2)}$ are given by formulae (6.4.12) and (6.4.13) respectively. Thus, in order to prove Theorem 6.2.1 we need to write down explicitly the differential operators $A^{(1)}$ and $A^{(2)}$ appearing in the asymptotic expansion of the perturbed massless Dirac operator on half-densities,

$$A(\epsilon) = A^{(0)} + \epsilon A^{(1)} + \epsilon^2 A^{(2)} + O(\epsilon^3).$$
(6.6.4)

In what follows we use terminology from microlocal analysis. In particular, we use the notions of the *principal* and *subprincipal* symbols of a differential operator, see Subsection 3.1.3 in the current document or Subsection 2.1.3 in [72] for details.

Let L be a first order 2×2 matrix differential operator. We denote its principal and subprincipal symbols by $L_1(x,\xi)$ and $L_{sub}(x)$ respectively. Here $\xi = (\xi_1, \xi_2, \xi_3)$ is the variable dual to the position variable x. The subscript in $L_1(x,\xi)$ indicates the degree of homogeneity in ξ . A first order differential operator L is completely determined by its principal and subprincipal symbols. Indeed, the principal symbol has the form

$$L_1(x,\xi) = M^{(\alpha)}(x)\,\xi_\alpha\,, \tag{6.6.5}$$

where $M^{(\alpha)}(x)$ are matrix-functions depending only on the position variable x. It is easy to see that the differential operator L is given by the formula

$$L = -\frac{i}{2}M^{(\alpha)}(x)\frac{\partial}{\partial x^{\alpha}} - \frac{i}{2}\frac{\partial}{\partial x^{\alpha}}M^{(\alpha)}(x) + L_{\rm sub}(x).$$
(6.6.6)

Given a first order differential operator L, let us consider the expression $\langle Lv^{(0)}, v^{(0)} \rangle$, where $v^{(0)}$ is the constant column (6.6.1) and angular brackets indicate the inner product (6.3.2). Examination of formula (6.6.6) shows that

$$\langle Lv^{(0)}, v^{(0)} \rangle = \langle L_{\rm sub}v^{(0)}, v^{(0)} \rangle$$

because the terms coming from the principal symbol integrate to zero. Consequently formulae (6.4.12) and (6.4.13) simplify and now read

$$\lambda^{(1)} = \langle A_{\rm sub}^{(1)} v^{(0)}, v^{(0)} \rangle, \tag{6.6.7}$$

$$\lambda^{(2)} = \langle A_{\rm sub}^{(2)} v^{(0)}, v^{(0)} \rangle - \langle (A^{(1)} - \lambda^{(1)}) Q (A^{(1)} - \lambda^{(1)}) v^{(0)}, v^{(0)} \rangle.$$
 (6.6.8)

We see that for the purpose of proving Theorem 6.2.1 we do not need to know the full operator $A^{(2)}$, only its subprincipal symbol $A^{(2)}_{sub}$.

In order to write down explicitly the massless Dirac operator on half-densities $A(\epsilon)$ we need the concepts of *frame* and *coframe*. The differential geometric definition of coframe was given in Section 3 of [21]. However, as in the current chapter we are working in a specified coordinate system, we can adopt a somewhat simpler approach. For the purposes of the current chapter a coframe is a smooth

real-valued matrix-function $e^{j}_{\alpha}(x;\epsilon), j, \alpha = 1, 2, 3$, satisfying the conditions

$$g_{\alpha\beta}(x;\epsilon) = \delta_{jk} e^{j}{}_{\alpha}(x;\epsilon) e^{k}{}_{\beta}(x;\epsilon) , \qquad (6.6.9)$$

$$e^{j}{}_{\alpha}(x;0) = \delta^{j}{}_{\alpha}$$
. (6.6.10)

Here and further on when dealing with matrix-functions we use the convention that the first index (subscript or superscript) enumerates the rows and the second index (subscript or superscript) enumerates the columns. Say, in matrix notation the RHS of (6.6.9) reads as "product of coframe transposed and coframe".

Note that the reason we imposed condition (6.6.10) is so that our unperturbed operator has the form (6.1.1). See also formula (6.3.1) and associated discussion.

For a given metric $g_{\alpha\beta}(x;\epsilon)$ the coframe $e^{j}{}_{\alpha}(x;\epsilon)$ is not defined uniquely. We can multiply the matrix-function $e^{j}{}_{\alpha}(x;\epsilon)$ from the left by an arbitrary smooth 3×3 special orthogonal matrix-function $O(x;\epsilon)$ satisfying the condition O(x;0) = I, with I denoting the 3×3 identity matrix. This will give us a new coframe satisfying the defining relations (6.6.9) and (6.6.10). As explained in Appendix 4.A, this freedom in the choice of coframe is a gauge degree of freedom which does not affect the spectrum. In the current section we specify the gauge by requiring the matrix-function $e^{j}{}_{\alpha}(x;\epsilon)$ to be symmetric,

$$e^{j}{}_{\alpha}(x;\epsilon) = e^{\alpha}{}_{j}(x;\epsilon). \tag{6.6.11}$$

Condition (6.6.11) makes sense because we are working in a specified coordinate system. Looking ahead, let us point out the main advantage of the symmetric gauge (6.6.11): the asymptotic expansion of the subprincipal symbol of the massless Dirac operator on half-densities in powers of ϵ starts with a quadratic term and, moreover, the coefficient at ϵ^2 has an especially simple structure, see formulae (6.6.16) and (6.6.19).

In matrix notation condition (6.6.9) now reads "the symmetric positive matrix $g_{\alpha\beta}(x;\epsilon)$ is the square of the symmetric matrix $e^{j}{}_{\alpha}(x;\epsilon)$ ". Conversely, the symmetric matrix $e^{j}{}_{\alpha}(x;\epsilon)$ is the square root of the symmetric positive matrix $g_{\alpha\beta}(x;\epsilon)$. We choose the branch of the square root so that the matrix $e^{j}{}_{\alpha}(x;\epsilon)$ is positive.

According to formulae (6.1.2) and (6.2.3) we have

$$g_{\alpha\beta}(x;\epsilon) = \delta_{\alpha\beta} + \epsilon h_{\alpha\beta}(x) + O(\epsilon^2), \qquad (6.6.12)$$

hence, by Taylor's formula for $\sqrt{1+z}$,

$$e^{j}{}_{\alpha}(x;\epsilon) = \delta^{j}{}_{\alpha} + \frac{\epsilon}{2}h^{j}{}_{\alpha}(x) + O(\epsilon^{2}).$$
(6.6.13)

Here we follow the convention introduced in Section 6.2: we raise and lower indices in h using the Euclidean metric, which means that raising or lowering an index doesn't change anything. We also swap, when needed, tensor (Greek) indices for frame (Latin) indices, which is acceptable because we are working in a specified coordinate system.

The *frame* is the smooth real-valued matrix-function $e_j^{\alpha}(x;\epsilon)$, $j, \alpha = 1, 2, 3$, defined by the system of linear algebraic equations

$$e_j^{\ \alpha}(x;\epsilon) e_{\ \alpha}^k(x;\epsilon) = \delta_j^k. \tag{6.6.14}$$

Note the position of indices in formula (6.6.14). In matrix notation formula (6.6.14) reads as "the frame is the transpose of the inverse of the coframe". As we chose our coframe to be symmetric, our frame is also symmetric and is simply the inverse of the coframe. Formula (6.6.13) and Taylor's formula for $(1 + z)^{-1}$ imply

$$e_j^{\ \alpha}(x;\epsilon) = \delta_j^{\ \alpha} - \frac{\epsilon}{2}h_j^{\ \alpha}(x) + O(\epsilon^2). \tag{6.6.15}$$

According to formulae (4.6.1), (4.3.5) and (4.8.1) the subprincipal symbol of the massless Dirac operator on half-densities is

$$A_{\rm sub}(x;\epsilon) = \frac{3}{4} \left(*T^{\rm ax}(x;\epsilon) \right) I , \qquad (6.6.16)$$

where I is the 2 \times 2 identity matrix and $*T^{ax}(x;\epsilon)$ is the scalar function

$$*T^{ax} = \frac{\delta_{kl}}{3} \sqrt{\det g^{\alpha\beta}} \left[e^{k}_{1} \partial e^{l}_{3} / \partial x^{2} + e^{k}_{2} \partial e^{l}_{1} / \partial x^{3} + e^{k}_{3} \partial e^{l}_{2} / \partial x^{1} - e^{k}_{1} \partial e^{l}_{2} / \partial x^{3} - e^{k}_{2} \partial e^{l}_{3} / \partial x^{1} - e^{k}_{3} \partial e^{l}_{1} / \partial x^{2} \right]. \quad (6.6.17)$$

Note that formula (6.6.12) implies $g^{\alpha\beta}(x;\epsilon) = \delta_{\alpha\beta} - \epsilon h^{\alpha\beta}(x) + O(\epsilon^2)$, which, in turn, gives us

$$\sqrt{\det g^{\alpha\beta}(x;\epsilon)} = 1 - \frac{\epsilon}{2} \operatorname{tr} h(x) + O(\epsilon^2).$$
(6.6.18)

Substituting formulae (6.6.18) and (6.6.13) into (6.6.17) and using the symmetry condition (6.6.11), we get

$$*T^{\mathrm{ax}}(x;\epsilon) = \frac{\epsilon^2 \delta^{kl}}{12} \left[h_{k1} \frac{\partial h_{l3}}{\partial x^2} + h_{k2} \frac{\partial h_{l1}}{\partial x^3} + h_{k3} \frac{\partial h_{l2}}{\partial x^1} - h_{k1} \frac{\partial h_{l2}}{\partial x^3} - h_{k2} \frac{\partial h_{l3}}{\partial x^1} - h_{k3} \frac{\partial h_{l1}}{\partial x^2} \right] + O(\epsilon^3)$$
$$= -\frac{\epsilon^2}{12} \varepsilon_{\beta\gamma\delta} h_{\alpha\beta} \frac{\partial h_{\alpha\gamma}}{\partial x^{\delta}} + O(\epsilon^3) . \quad (6.6.19)$$

Formulae (6.6.16) and (6.6.19) imply

$$A_{\rm sub}^{(1)}(x) = 0, \tag{6.6.20}$$

$$A_{\rm sub}^{(2)}(x) = -\frac{1}{16} \varepsilon_{\beta\gamma\delta} h_{\alpha\beta} \frac{\partial h_{\alpha\gamma}}{\partial x^{\delta}} I. \qquad (6.6.21)$$

Substituting (6.6.20) into (6.6.7) we get $\lambda^{(1)} = 0$. Formulae (6.6.3) and (6.6.8) now simplify and read

$$\lambda(\epsilon) = c \,\epsilon^2 + O(\epsilon^3), \tag{6.6.22}$$

$$c = \lambda^{(2)} = \langle A_{\text{sub}}^{(2)} v^{(0)}, v^{(0)} \rangle - \langle A^{(1)} Q A^{(1)} v^{(0)}, v^{(0)} \rangle.$$
 (6.6.23)

In order to complete our calculation we now need only to write down the principal symbol $A_1^{(1)}(x,\xi)$ of the differential operator $A^{(1)}$.

According to formulae (4.A.1)–(4.A.3) and (4.A.19) the principal symbol of the massless Dirac operator on half-densities is

$$A_{1}(x,\xi;\epsilon) = \begin{pmatrix} e_{3}^{\alpha} & e_{1}^{\alpha} - ie_{2}^{\alpha} \\ e_{1}^{\alpha} + ie_{2}^{\alpha} & -e_{3}^{\alpha} \end{pmatrix} \xi_{\alpha} .$$
(6.6.24)

Formulae (6.6.24) and (6.6.15) imply

$$A_1^{(1)}(x,\xi) = -\frac{1}{2} \begin{pmatrix} h_3^{\alpha} & h_1^{\alpha} - ih_2^{\alpha} \\ h_1^{\alpha} + ih_2^{\alpha} & -h_3^{\alpha} \end{pmatrix} \xi_{\alpha}.$$
(6.6.25)

Formulae (6.6.5), (6.6.6), (6.6.20) and (6.6.25) allow us to write down the differential operator $A^{(1)}$ explicitly:

$$A^{(1)} = \frac{i}{4} \begin{pmatrix} h_3^{\alpha} & h_1^{\alpha} - ih_2^{\alpha} \\ h_1^{\alpha} + ih_2^{\alpha} & -h_3^{\alpha} \end{pmatrix} \frac{\partial}{\partial x^{\alpha}} + \frac{i}{4} \frac{\partial}{\partial x^{\alpha}} \begin{pmatrix} h_3^{\alpha} & h_1^{\alpha} - ih_2^{\alpha} \\ h_1^{\alpha} + ih_2^{\alpha} & -h_3^{\alpha} \end{pmatrix}.$$
 (6.6.26)

Substituting formulae (6.6.1), (6.6.2), (6.6.21) and (6.6.26) into (6.6.23) we arrive at (6.2.5). This completes the proof of Theorem 6.2.1.

6.7 Axisymmetric case

An important special case is when the metric $g_{\alpha\beta}(x;\epsilon)$ is a function of the coordinate x^1 only. In this case one can choose the coframe and frame so that they depend on the coordinate x^1 only and seek eigenfunctions in the form

$$v(x) = u(x^1)e^{i(m_2x^2 + m_3x^3)}, \qquad m_2, m_3 \in \mathbb{Z}.$$

We get separation of variables, i.e. the original eigenvalue problem for a partial differential operator reduces to an eigenvalue problem for an ordinary differential operator depending on the integers m_2 and m_3 as parameters. As we know the spectrum of the unperturbed operator (see Section 6.1) and as the eigenvalues of the original partial differential operator depend on the small parameter ϵ continuously, the eigenvalue with smallest modulus will come from the ordinary differential operator with $m_2 = m_3 = 0$. We call the case $m_2 = m_3 = 0$ the axisymmetric case.

The axisymmetric massless Dirac operator on half-densities reads

$$W_{1/2}(\epsilon) = -\frac{i}{2} \begin{pmatrix} e_3^1 & e_1^1 - ie_2^1 \\ e_1^1 + ie_2^1 & -e_3^1 \end{pmatrix} \frac{d}{dx^1} - \frac{i}{2} \frac{d}{dx^1} \begin{pmatrix} e_3^1 & e_1^1 - ie_2^1 \\ e_1^1 + ie_2^1 & -e_3^1 \end{pmatrix} + \frac{\delta_{jk}}{4\sqrt{\det g_{\alpha\beta}}} \left[e^{j}_3 \left(\frac{de^{k_2}}{dx^1} \right) - e^{j}_2 \left(\frac{de^{k_3}}{dx^1} \right) \right] I, \quad (6.7.1)$$

where I is the 2×2 identity matrix and

$$\sqrt{\det g_{\alpha\beta}} = \frac{1}{\sqrt{\det g^{\alpha\beta}}} = \det e^j{}_{\alpha} = \frac{1}{\det e_j{}^{\alpha}}.$$

Here $e^{j}{}_{\alpha}(x^{1};\epsilon)$ and $e_{j}{}^{\alpha}(x^{1};\epsilon)$ are the coframe and frame defined in accordance with formulae (6.6.9), (6.6.10) and (6.6.14).

Of course, for a given metric $g_{\alpha\beta}(x^1;\epsilon)$ the coframe $e^j{}_{\alpha}(x^1;\epsilon)$ and frame $e_j{}^{\alpha}(x^1;\epsilon)$ are not defined uniquely. We can multiply the matrix-functions $e^j{}_{\alpha}(x^1;\epsilon)$ and $e_j{}^{\alpha}(x^1;\epsilon)$ from the left by an arbitrary smooth 3×3 special orthogonal matrixfunction $O(x^1;\epsilon)$ satisfying the condition $O(x^1;0) = I$, with I denoting the 3×3 identity matrix. This will give us a new coframe and a new frame satisfying the defining relations (6.6.9), (6.6.10) and (6.6.14). Note that in writing down formula (6.7.1) we did not assume a particular choice of gauge, compare with (6.6.11).
In the axisymmetric case formula (6.2.5) also simplifies and reads now

$$c = -\frac{1}{8} \sum_{m_1 \in \mathbb{N}} m_1 \operatorname{tr} \left[\begin{pmatrix} \hat{h}_{22} & \hat{h}_{23} \\ \hat{h}_{32} & \hat{h}_{33} \end{pmatrix} \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix} \begin{pmatrix} \hat{h}_{22} & \hat{h}_{23} \\ \hat{h}_{32} & \hat{h}_{33} \end{pmatrix}^* \right], \quad (6.7.2)$$

where $\hat{h}_{\alpha\beta} = \hat{h}_{\alpha\beta}(m_1)$ and the star stands for Hermitian conjugation.

6.8 Example of quadratic dependence on ϵ

Consider the metric

$$g_{\alpha\beta}(x^{1};\epsilon) dx^{\alpha} dx^{\beta} = \left[dx^{1} \right]^{2} + \left[\left(1 + \epsilon \left(\cos x^{1} \right) \right) dx^{2} + \epsilon \left(\sin x^{1} \right) dx^{3} \right]^{2} + \left[\epsilon \left(\sin x^{1} \right) dx^{2} + \left(1 - \epsilon \left(\cos x^{1} \right) \right) dx^{3} \right]^{2}. \quad (6.8.1)$$

Then

$$e^{j}{}_{\alpha}(x^{1};\epsilon) = \delta^{j}{}_{\alpha} + \epsilon \begin{pmatrix} 0 & 0 & 0 \\ 0 & \cos x^{1} & \sin x^{1} \\ 0 & \sin x^{1} & -\cos x^{1} \end{pmatrix}$$
(6.8.2)

is a coframe associated with the metric (6.8.1), see formulae (6.6.9), (6.6.10), and

$$e_{j}^{\ \alpha}(x^{1};\epsilon) = \begin{pmatrix} 1 & 0 & 0 \\ 0 & \frac{1-\epsilon\cos x^{1}}{1-\epsilon^{2}} & -\frac{\epsilon\sin x^{1}}{1-\epsilon^{2}} \\ 0 & -\frac{\epsilon\sin x^{1}}{1-\epsilon^{2}} & \frac{1+\epsilon\cos x^{1}}{1-\epsilon^{2}} \end{pmatrix}$$
(6.8.3)

is the corresponding frame, see formula (6.6.14). Note that in writing formula (6.8.3) we used the fact that

$$\det e^{j}{}_{\alpha}(x;\epsilon) = 1 - \epsilon^{2} = \sqrt{\det g_{\alpha\beta}}.$$
(6.8.4)

Substituting formulae (6.8.2)–(6.8.4) into (6.7.1) we get

$$W(\epsilon) = -i \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \frac{d}{dx^1} - \frac{\epsilon^2}{2(1-\epsilon^2)} I.$$
 (6.8.5)

Note that in the LHS we dropped the subscript 1/2: as the Riemannian density is constant, see (6.8.4), there is no need to distinguish the massless Dirac operator $W(\epsilon)$ and the massless Dirac operator on half-densities $W_{1/2}(\epsilon)$.

It is easy to see that the eigenvalues of the ordinary differential operator (6.8.5) subject to the (boundary) condition of 2π -periodicity are

$$\lambda_n(\epsilon) = n - \frac{\epsilon^2}{2(1-\epsilon^2)}, \qquad n \in \mathbb{Z},$$

and that all eigenvalues have multiplicity two. In particular, the eigenvalue with smallest modulus is

$$\lambda_0(\epsilon) = -\frac{\epsilon^2}{2(1-\epsilon^2)} = -\frac{\epsilon^2}{2} + O(\epsilon^4) \quad \text{as} \quad \epsilon \to 0.$$
 (6.8.6)

Let us now test Theorem 6.2.1 by comparing the asymptotic formula from this theorem with formula (6.8.6). Substituting (6.8.1) into (6.2.3) we get

$$h_{\alpha\beta}(x^{1}) = 2 \begin{pmatrix} 0 & 0 & 0 \\ 0 & \cos x^{1} & \sin x^{1} \\ 0 & \sin x^{1} & -\cos x^{1} \end{pmatrix}.$$

Application of the Fourier transform (6.2.2) gives us

$$\hat{h}_{\alpha\beta}(m_1) = \begin{cases} \begin{pmatrix} 0 & 0 & 0 \\ 0 & 1 & -i \\ 0 & -i & -1 \end{pmatrix} & \text{for } m_1 = 1, \\ 0 & -i & -1 \end{pmatrix}$$
(6.8.7)

Substituting (6.8.7) into (6.7.2) we get $c = -\frac{1}{2}$, in agreement with (6.8.6).

6.9 Example of quartic dependence on ϵ

Consider the metric

$$g_{\alpha\beta}(x^{1};\epsilon) dx^{\alpha} dx^{\beta} = \left[dx^{1} + \epsilon \left(\cos x^{1} \right) dx^{2} + \epsilon \left(\sin x^{1} \right) dx^{3} \right]^{2} + \left[dx^{2} \right]^{2} + \left[dx^{3} \right]^{2}. \quad (6.9.1)$$

Then

$$e^{j}{}_{\alpha}(x^{1};\epsilon) = \delta^{j}{}_{\alpha} + \epsilon \begin{pmatrix} 0 & \cos x^{1} & \sin x^{1} \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}$$
(6.9.2)

is a coframe associated with the metric (6.9.1), see formulae (6.6.9), (6.6.10), and

$$e_{j}^{\alpha}(x^{1};\epsilon) = \delta_{j}^{\alpha} - \epsilon \begin{pmatrix} 0 & 0 & 0\\ \cos x^{1} & 0 & 0\\ \sin x^{1} & 0 & 0 \end{pmatrix}$$
(6.9.3)

is the corresponding frame, see formula (6.6.14). Note that in writing formula (6.9.3) we used the fact that

$$\det e^{j}{}_{\alpha}(x;\epsilon) = 1 = \sqrt{\det g_{\alpha\beta}}.$$
(6.9.4)

Substituting formulae (6.9.2)–(6.9.4) into (6.7.1) we get

$$W(\epsilon) = -i \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \frac{d}{dx^{1}} - \frac{\epsilon^{2}}{4}I + \frac{i\epsilon}{2} \begin{pmatrix} \sin x^{1} & -i\cos x^{1} \\ i\cos x^{1} & -\sin x^{1} \end{pmatrix} \frac{d}{dx^{1}} + \frac{i\epsilon}{2} \frac{d}{dx^{1}} \begin{pmatrix} \sin x^{1} & -i\cos x^{1} \\ i\cos x^{1} & -\sin x^{1} \end{pmatrix}.$$
 (6.9.5)

Note that in the LHS we dropped the subscript 1/2: as the Riemannian density is constant, see (6.9.4), there is no need to distinguish the massless Dirac operator $W(\epsilon)$ and the massless Dirac operator on half-densities $W_{1/2}(\epsilon)$.

We shall now rewrite the ordinary differential operator (6.9.5) in a somewhat more convenient form. To this end, let us introduce the special unitary matrix

$$R := \frac{1}{\sqrt{2}} \begin{pmatrix} 1 & 1\\ -1 & 1 \end{pmatrix}$$
(6.9.6)

and put

$$\tilde{W}(\epsilon) := R \ W(\epsilon) \ R^* \,, \tag{6.9.7}$$

compare with formula (6.3.1). Clearly, the operator $\tilde{W}(\epsilon)$ has the same spectrum as the operator $W(\epsilon)$. Substituting (6.9.5) and (6.9.6) into (6.9.7) we arrive at the following explicit formula for the ordinary differential operator $\tilde{W}(\epsilon)$:

$$\tilde{W}(\epsilon) = -i \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} \frac{d}{dx^{1}} - \frac{\epsilon^{2}}{4}I + \frac{i\epsilon}{2} \begin{pmatrix} 0 & -ie^{-ix^{1}} \\ ie^{ix^{1}} & 0 \end{pmatrix} \frac{d}{dx^{1}} + \frac{i\epsilon}{2} \frac{d}{dx^{1}} \begin{pmatrix} 0 & -ie^{-ix^{1}} \\ ie^{ix^{1}} & 0 \end{pmatrix}.$$
 (6.9.8)

The coefficients of the ordinary differential operator (6.9.8) are trigonometric polynomials and one would not normally expect the eigenfunctions to be trigonometric polynomials. However, the operator (6.9.8) has a special structure which ensures that the eigenfunctions are trigonometric polynomials. Namely, put

$$\lambda_n(\epsilon) = -\frac{1}{2} - \frac{\epsilon^2}{4} + \sqrt{1 + \epsilon^2} \left(n + \frac{1}{2} \right), \qquad n \in \mathbb{Z}, \tag{6.9.9}$$

$$v^{(n)}(x^{1};\epsilon) = \begin{pmatrix} \left(1 + \sqrt{1 + \epsilon^{2}}\right) e^{inx^{1}} \\ -i\epsilon e^{i(n+1)x^{1}} \end{pmatrix}, \qquad n \in \mathbb{Z}.$$
 (6.9.10)

It is easy to see that the column-functions (6.9.10) are eigenfunctions of the operator (6.9.8) corresponding to eigenvalues (6.9.9). Moreover, it is easy to

see that the charge conjugates, $C(v^{(n)}(x^1; \epsilon))$, of the column-functions (6.9.10) are eigenfunctions of the operator (6.9.8) corresponding to the same eigenvalues (6.9.9). This means that the numbers (6.9.9) are eigenvalues of the operator (6.9.8) of multiplicity at least two. Finally, it is easy to see that

$$span\{v^{(n)}(x^{1};\epsilon), C(v^{(n)}(x^{1};\epsilon)) \mid n \in \mathbb{Z}\} = span\{v^{(n)}(x^{1};0), C(v^{(n)}(x^{1};0)) \mid n \in \mathbb{Z}\}, (6.9.11)$$

where span S denotes the linear span, i.e. set of all finite linear combinations of elements of a given set S. Formula (6.9.11) implies that we haven't missed any eigenvalues, that is, that the list (6.9.9) contains *all* the eigenvalues of the operator (6.9.8) and that each of these eigenvalues has multiplicity two.

Remark 6.9.1. We do not fully understand the underlying reasons why the axisymmetric massless Dirac operator corresponding to the metric (6.9.1) admits an explicit evaluation of the eigenvalues and eigenfunctions. Somehow, this particular Dirac operator has properties similar to those of an integrable system.

The eigenvalue (6.9.9) with smallest modulus is

$$\lambda_0(\epsilon) = \frac{2\sqrt{1+\epsilon^2} - 2 - \epsilon^2}{4} = -\frac{\epsilon^4}{16} + O(\epsilon^6) \quad \text{as} \quad \epsilon \to 0.$$
 (6.9.12)

Let us now test Theorem 6.2.1 by comparing the asymptotic formula from this theorem with formula (6.9.12). Substituting (6.9.1) into (6.2.3) we get

$$h_{\alpha\beta}(x^{1}) = \begin{pmatrix} 0 & \cos x^{1} & \sin x^{1} \\ \cos x^{1} & 0 & 0 \\ \sin x^{1} & 0 & 0 \end{pmatrix}.$$

Application of the Fourier transform (6.2.2) gives us

$$\hat{h}_{\alpha\beta}(m_1) = \begin{cases} \begin{pmatrix} 0 & \frac{1}{2} & -\frac{i}{2} \\ \frac{1}{2} & 0 & 0 \\ -\frac{i}{2} & 0 & 0 \end{pmatrix} & \text{for } m_1 = 1, \\ & & \\ 0 & & \text{for } m_1 = 2, 3, \dots \end{cases}$$
(6.9.13)

Substituting (6.9.13) into (6.7.2) we get c = 0, in agreement with (6.9.12).

6.10 The eta invariant

Let H be a first order self-adjoint elliptic $m \times m$ matrix classical pseudodifferential operator acting on m-columns of complex-valued half-densities over a compact n-dimensional manifold M without boundary. Here ellipticity is understood as the nonvanishing of the determinant of the principal symbol of H, see Chapter 4. The *eta function* of H is defined as

$$\eta_H(s) := \sum \frac{\operatorname{sign} \lambda}{|\lambda|^s}, \qquad (6.10.1)$$

where summation is carried out over all nonzero eigenvalues λ of H, and $s \in \mathbb{C}$ is the independent variable. Asymptotic formulae for the counting function imply that the series (6.10.1) converges absolutely for Re s > n and defines a holomorphic function in this half-plane. It is known [7] that the eta function extends meromorphically to the whole *s*-plane. Moreover, it is known, see Theorem 4.5 in [7], that if the dimension *n* is odd, then the eta function is holomorphic at s = 0. This justifies, for odd *n*, the definition of the *eta invariant* as the real number $\eta_H(0)$. The eta invariant $\eta_H(0)$ is the traditional measure of spectral asymmetry of the operator *H*. If we have only a finite number of eigenvalues (i.e. if we are looking at an Hermitian matrix rather than a differential operator) then the eta invariant is an integer number: it is the number of positive eigenvalues minus the number of negative eigenvalues. However, in the case of a differential operator there is no reason for the eta invariant to be integer. The basic example [4] is that of the scalar ordinary differential operator $H(\epsilon) := -i\frac{d}{dx^1} + \epsilon$ acting on the unit circle parameterized by the cyclic coordinate x^1 of period 2π , with ϵ being a real parameter. It is known [4] that the eta invariant $\eta_{H(\epsilon)}(0)$ of this ordinary differential operator is the odd 1-periodic function defined by the formula $\eta_{H(\epsilon)}(0) = 1 - 2\epsilon$ for $\epsilon \in (0, 1)$. In particular, we have $\eta_{H(0)}(0) = 0$ and $\lim_{\epsilon \to 0^{\pm}} \eta_{H(\epsilon)}(0) = \pm 1$.

The current state of affairs (from an analyst's perspective) in the subject area of zeta/eta functions of elliptic operators is described in detail in the two papers [36, 37]. Let us highlight a few facts.

- The key results are Theorem 2.7 from [36] and Proposition 2.9 from [37]. Arguing along the lines of [7] one can recover from these results, in a rigorous analytic fashion, properties of the eta function.
- The eta function is holomorphic at s = 0 in any dimension n ∈ N (i.e. without the assumption of n being odd). This fact was proved by P. B. Gilkey [31].
- The seminal paper of R. T. Seeley [75] contained a small mistake: see page 482 in [36] or Remark 2.6 on page 39 in [37] for details.

The more recent survey papers [34, 35] provide an overview of the subject.

Let us denote our massless Dirac operator on half-densities by $A(\epsilon)$, where $\epsilon \in \mathbb{R}$ is the small parameter appearing in our metric $g_{\alpha\beta}(x;\epsilon)$. Theorem 6.2.1 implies the following corollary.

Corollary 6.10.1. Suppose that the coefficient c defined by formula (6.2.5) is nonzero. Then

$$\lim_{\epsilon \to 0} \eta_{A(\epsilon)}(0) = 2 \operatorname{sign} c.$$
(6.10.2)

Note that we have $\eta_{A(0)}(0) = 0$, so formula (6.10.2) implies that the function $\eta_{A(\epsilon)}(0)$ is discontinuous at $\epsilon = 0$.

Proof of Corollary 6.10.1 Put $f(\epsilon, t) := \text{Tr} \left[A(\epsilon) e^{-t(A(\epsilon))^2}\right]$, where t > 0 and Tr is the operator (as opposed to pointwise) trace². Having fixed ϵ , let us examine the behaviour of $f(\epsilon, t)$ as $t \to 0^+$. For a generic first order pseudodifferential operator $A(\epsilon)$ we would have $f(\epsilon, t) = O(t^{-2})$. However, as explained in Chapter II of [11], the Dirac operator in odd dimensions is very special and there are a lot of cancellations when one computes the asymptotic expansion for $f(\epsilon, t)$ as $t \to 0^+$. Namely, it was shown in [11] that

- $f(\epsilon, t) = O(\sqrt{t})$ as $t \to 0^+$,
- $\eta_{A(\epsilon)}(s)$ is holomorphic in the half-plane $\operatorname{Re} s > -2$, and

$$\eta_{A(\epsilon)}(s) = \frac{1}{\Gamma\left(\frac{s+1}{2}\right)} \int_0^{+\infty} t^{(s-1)/2} f(\epsilon, t) \, dt \quad \text{for} \quad \text{Re}\, s > -2 \,. \tag{6.10.3}$$

See also Section 1 in [82].

Formula (6.10.3) implies

$$\eta_{A(\epsilon)}(0) = \frac{1}{\sqrt{\pi}} \int_0^{+\infty} \frac{f(\epsilon, t)}{\sqrt{t}} dt , \qquad (6.10.4)$$

so in order to prove Corollary 6.10.1 we need to examine the behaviour of the integral (6.10.4) as $\epsilon \to 0$.

Let us denote by $\lambda_0(\epsilon)$ the eigenvalue of the operator $A(\epsilon)$ with smallest modulus and by $E_0(\epsilon)$ the orthogonal projection onto the corresponding 2-dimensional

² The paper [11] to which we are about to refer to actually deals with pointwise estimates, i.e. the trace in [11] is understood as the matrix trace of the integral kernel on the diagonal at a given point x of the manifold. We do not need pointwise estimates for the proof of Corollary 6.10.1.

eigenspace. Put

$$A_0(\epsilon) := \lambda_0(\epsilon) E_0(\epsilon), \qquad \tilde{A}(\epsilon) := A(\epsilon) - A_0(\epsilon),$$

$$f_0(\epsilon, t) := \operatorname{Tr} \left[A_0(\epsilon) e^{-t(A_0(\epsilon))^2} \right] = 2 \lambda_0(\epsilon) e^{-t(\lambda_0(\epsilon))^2},$$

$$\tilde{f}(\epsilon, t) := \operatorname{Tr} \left[\tilde{A}(\epsilon) e^{-t(\tilde{A}(\epsilon))^2} \right] = f(\epsilon, t) - f_0(\epsilon, t).$$

Then formula (6.10.4) can be rewritten as

$$\eta_{A(\epsilon)}(0) = \frac{1}{\sqrt{\pi}} \int_0^1 \frac{f(\epsilon, t)}{\sqrt{t}} dt + \frac{1}{\sqrt{\pi}} \int_1^{+\infty} \frac{\tilde{f}(\epsilon, t)}{\sqrt{t}} dt + \frac{2}{\sqrt{\pi}} \int_1^{+\infty} \frac{\lambda_0(\epsilon) e^{-t(\lambda_0(\epsilon))^2}}{\sqrt{t}} dt. \quad (6.10.5)$$

The three terms in the RHS of (6.10.5) are functions of the parameter ϵ and we shall now examine how they depend on ϵ .

The first term in the RHS of (6.10.5) is continuous at $\epsilon = 0$ because asymptotic formulae for $f(\epsilon, t)$ as $t \to 0^+$ are uniform in ϵ . This follows from the construction of heat kernel type asymptotics for $t \to 0^+$: the algorithm is straightforward and examination of this algorithm shows that the asymptotic coefficients and remainder term depend on additional parameters in a continuous fashion.

The second term in the RHS of (6.10.5) is continuous at $\epsilon = 0$ because the eigenvalues of the operator $\tilde{A}(\epsilon)$ depend on ϵ continuously and because all these eigenvalues, bar one double eigenvalue, are uniformly separated from zero. The double eigenvalue in question is identically zero as a function of ϵ and does not contribute to the second term in the RHS of (6.10.5).

Thus, the proof of formula (6.10.2) reduces to the proof of the statement

$$\frac{2}{\sqrt{\pi}} \lim_{\epsilon \to 0} \int_{1}^{+\infty} \frac{\lambda_0(\epsilon) e^{-t(\lambda_0(\epsilon))^2}}{\sqrt{t}} dt = 2 \operatorname{sign} c.$$
 (6.10.6)

But formula (6.10.6) is an immediate consequence of formula (6.2.4). \Box

dealing specifically with the Dirac operator on a 3-torus, though certain 2-torus bundles over a circle were examined in [3].

Bibliography

- B. Ammann, M. Dahl, and E. Humbert. Surgery and harmonic spinors. Advances In Mathematics, 220(2):523–539, 2009.
- [2] B. Ammann, M. Dahl, and E. Humbert. Harmonic spinors and local deformations of the metric. *Mathematical Research Letters*, 18(5):927–936, 2011.
- M. F. Atiyah, H. Donnelly, and I. M. Singer. Eta invariants, signature defects of cusps, and values of L-functions. *Annals Mathematics*, 118(1):131–177, 1983.
- [4] M. F. Atiyah, V. K. Patodi, and I. M. Singer. Spectral asymmetry and Riemannian geometry. Bull. London Math. Soc., 5:229–234, 1973.
- [5] M. F. Atiyah, V. K. Patodi, and I. M. Singer. Spectral asymmetry and Riemannian geometry I. Math. Proc. Camb. Phil. Soc., 77:43–69, 1975.
- [6] M. F. Atiyah, V. K. Patodi, and I. M. Singer. Spectral asymmetry and Riemannian geometry II. Math. Proc. Camb. Phil. Soc., 78:405–432, 1975.
- [7] M. F. Atiyah, V. K. Patodi, and I. M. Singer. Spectral asymmetry and Riemannian geometry III. Math. Proc. Camb. Phil. Soc., 79:71–99, 1976.
- [8] C. Bär. The Dirac operator on space forms of positive curvature. J. Math. Soc. Japan, 48(1):69–83, 1996.
- [9] C. Bär. On harmonic spinors. Acta Physica Polonica B, 29(4):859–869, 1998.
- [10] C. Bär. Dependence of the Dirac spectrum on the spin structure. Séminaires et Congrès, 4:17–33, 2000.

- [11] J. M. Bismut and D. S. Freed. The analysis of elliptic families. II. Dirac operators, eta-invariants, and the holonomy theorem. *Communications In Mathematical Physics*, 107(1):103–163, 1986.
- [12] C. G. Boehmer, R. J. Downes, and D. Vassiliev. Rotational elasticity. Quarterly J. Mechanics Appl. Mathematics, 64(4):415–439, 2011.
- [13] M. Brocato and G. Capriz. Gyrocontinua. Int. J. Solids Structures, 38(6-7):1089–1103, 2001.
- [14] I. L. Buchbinder and S. M. Kuzenko. Ideas and methods of supersymmetry and supergravity or a walk through superspace. Institute of Physics, 1998.
- [15] J. Burnett and D. Vassiliev. Weyl's Lagrangian in Teleparallel form. J. Math. Phys., 50(10):102501, 2009.
- [16] J. Burnett and D. Vassiliev. Modelling the electron with Cosserat elasticity. Mathematika, 58(2):349–370, 2012.
- [17] É. Cartan. Sur un généralisation de la notion de courbure de Riemann et les espaces á torsion. C.R. Acad. Sci., 174:593–595, 1922.
- [18] É. Cartan and A. Einstein. Letters on absolute parallelism, 1929–1932.
 Princeton University Press, Princeton, N.J., 1979. Original text with English translation by Jules Leroy and Jim Ritter, Edited by Robert Debever.
- [19] F. Chamizo, E. Cristobal, and A. Ubis. Lattice points in rational ellipsoids.
 J. Math. Analysis Applications, 350(1):283–289, 2009.
- [20] O. Chervova. The massless Dirac operator from the continuum mechanics and microlocal analysis perspectives. Ph.D. thesis, University College London, 2012.
- [21] O. Chervova, R. J. Downes, and D. Vassiliev. Spectral theoretic characterization of the massless Dirac operator. Available as a preprint: http://arxiv.org/abs/1209.3510, 2012.

- [22] O. Chervova, R. J. Downes, and D. Vassiliev. The spectral function of a first order system. J. Spectr. Theory, 3(3):317–360, 2013.
- [23] O. Chervova and D. Vassiliev. The stationary Weyl equation and Cosserat elasticity. J. Phys. A: Mathematical and Theoretical, 43(33):335203, 2010.
- [24] E. Cosserat and F. Cosserat. *Théorie des corps déformables*. Librairie Scientifique A. Hermann et fils, Paris, 1909. Reprinted by Cornell University Library.
- [25] R. J. Downes, M. Levitin, and D. Vassiliev. Spectral asymmetry of the massless Dirac operator on a 3-torus. Submitted for publication. Available as a preprint: http://arxiv.org/abs/1306.5689, 2013.
- [26] R. J. Downes and D. Vassiliev. Spectral theoretic characterization of the massless Dirac action. Available as a preprint by arrangement with authors, 2012.
- [27] J. J. Duistermaat and V. W. Guillemin. The spectrum of positive elliptic operators and periodic bicharacteristics. *Invent. Math.*, 29:39–79, 1975.
- [28] L. Erdos and J. P. Solovej. The kernel of Dirac operators on S³ and R³. Rev. In Math. Phys., 13(10):1247–1280, 2001.
- [29] J. L. Ericksen. Twist waves in liquid crystals. Quarterly J. Mechanics Appl. Mathematics, 21:463–465, 1968.
- [30] T. Friedrich. Dirac operators in Riemannian geometry. Amer. Math. Soc., Providence (RI), 2000.
- [31] P. B. Gilkey. The residue of the global eta-function at the origin. Advances In Mathematics, 40(3):290–307, 1981.
- [32] P. B. Gilkey. Asymptotic formulae in spectral geometry. Chapman and Hall/CRC Press, 2003.

- [33] P. B. Gilkey, J. V. Leahy, and J. Park. Spinors, spectral geometry, and Riemannian submersions. Seoul National University, Research Institute of Mathematics, Global Analysis Research Centre, Seoul, 1998.
- [34] G. Grubb. Analysis of invariants associated with spectral boundary problems for elliptic operators. Spectral Geometry of Manifolds With Boundary and Decomposition of Manifolds, 366:Danish Sci. Res. Council, PDE Network, 2005.
- [35] G. Grubb. Remarks on nonlocal trace expansion coefficients. In B. Booss-Bavnbek, S. Klimek, M. Lesch, and W. Zhang, editors, *Analysis, Geometry* and Topology of Elliptic Operators, pages 215–234. World Scientific, Singapore, 2006.
- [36] G. Grubb and R. T. Seeley. Weakly parametric pseudodifferential-operators and Atiyah-Patodi-Singer boundary-problems. *Inventiones Mathematicae*, 121(3):481–529, 1995.
- [37] G. Grubb and R. T. Seeley. Zeta and eta functions for Atiyah-Patodi-Singer operators. *Journal of Geometric Analysis*, 6(1):31–77, 1996.
- [38] D. R. Heath-Brown. Lattice points in the sphere. In K. Győry, H. Iwaniec, and J. Urbanowicz, editors, Number theory in progress: proceedings of the International Conference on Number Theory held in honor of the 60th birthday of Andrzej Schnizel, pages 883–892, Berlin, 1999. Walter de Gruyter.
- [39] F. W. Heyl and Y. N. Obukhov. Élie Cartan's torsion in geometry and in field theory, an essay. Ann. Fond. Louis de Broglie, 32(2-3):157–194, 2007.
- [40] N. Hitchin. Harmonic spinors. Adv. In Mathematics, 14(1):1–55, 1974.
- [41] M. Hortacsu, K. D. Rothe, and B. Schroer. Zero-energy eigenstates for the Dirac boundary-problem. *Nuclear Physics B*, 171(3):530–542, 1980.
- [42] V. Ivrii. On the second term of the spectral asymptotics for the Laplace-Beltrami operator on manifolds with boundary and for elliptic operators acting in fiberings. *Soviet Mathematics Doklady*, 21:300–302, 1980.

- [43] V. Ivrii. Accurate spectral asymptotics for elliptic operators that act in vector bundles. *Functional Analysis and Its Applications*, 16:101–108, 1982.
- [44] V. Ivrii. Precise spectral asymptotics for elliptic operators acting in fiberings over manifolds with boundary. Lecture Notes in Mathematics. Springer, 1984.
- [45] V. Ivrii. Microlocal analysis and precise spectral asymptotics. Springer, 1998.
- [46] J. Jeong and P. Neff. Existence, uniqueness and stability in linear Cosserat elasticity for weakest curvature conditions. *Mathematics Mechanics Solids*, 15(1):78–95, 2010.
- [47] I. Kamotski and M. Ruzhansky. Regularity properties, representation of solutions, and spectral asymptotics of systems with multiplicities. *Comm. Partial Differential Equations*, 32:1–35, 2007.
- [48] R. C. Kirby. The topology of 4-manifolds. Lecture Notes in Mathematics. Springer, 1989.
- [49] H. Kleinert. Towards a unified field-theory of defects and stresses. Lettere Al Nuovo Cimento, 35(2):41–45, 1982.
- [50] L. D. Landau and E. M. Lifshitz. Course of theoretical physics. Vol. 7. Pergamon Press, Oxford, third edition, 1986. Theory of elasticity, Translated from the Russian by J. B. Sykes and W. H. Reid.
- [51] M. Lazar. On the fundamentals of the three-dimensional translation gauge theory of dislocations. *Mathematics Mechanics Solids*, 16(3):253–264, 2011.
- [52] J. M. Lee. Introduction to smooth manifolds. Spinger, 2002.
- [53] M. Nakahara. Geometry, topology and physics. Institute or Physics Publishing, Bristol and Philadelphia, 1998.
- [54] D. Natroshvili, R. Gachechiladze, A. Gachechiladze, and I. G. Stratis. Transmission problems in the theory of elastic hemitropic materials. *Applicable Analysis*, 86(12):1463–1508, 2007.

- [55] D. Natroshvili, L. Giorgashvili, and I. G. Stratis. Representation formulae of general solutions in the theory of hemitropic elasticity. *Quarterly J. Mechanics Appl. Mathematics*, 59:451–474, 2006.
- [56] D. Natroshvili and I. G. Stratis. Mathematical problems of the theory of elasticity of chiral materials for Lipschitz domains. *Math. Methods In Appl. Sciences*, 29(4):445–478, 2006.
- [57] P. Neff. The Cosserat couple modulus for continuous solids is zero viz the linearized Cauchy-stress tensor is symmetric. Zamm-zeitschrift Fur Angewandte Mathematik Und Mechanik, 86(11):892–912, 2006.
- [58] P. Neff. Existence of minimizers for a finite-strain micromorphic elastic solid. Proc. Royal Soc. Edinburgh Section A-mathematics, 136:997–1012, 2006.
- [59] P. Neff. A geometrically exact planar Cosserat shell-model with microstructure: Existence of minimizers for zero Cosserat couple modulus. *Math. Mod*els and Methods in Appl. Sciences, 17(3):363–392, 2007.
- [60] P. Neff, K. Hong, and J. Jeong. The Reissner-Mindlin plate is the gamma limit of Cosserat elasticity. Math. Models and Methods in Appl. Sciences, 20(9):1553–1590, 2010.
- [61] P. Neff and J. Jeong. A new paradigm: the linear isotropic Cosserat model with conformally invariant curvature energy. Zamm-zeitschrift Fur Angewandte Mathematik Und Mechanik, 89(2):107–122, 2009.
- [62] P. Neff, J. Jeong, and A. Fischle. Stable identification of linear isotropic Cosserat parameters: bounded stiffness in bending and torsion implies conformal invariance of curvature. *Acta Mechanica*, 211(3-4):237–249, 2010.
- [63] P. Neff and I. Muench. Curl bounds grad on SO(3). Esaim-control Optimisation Calculus Variations, 14(1):148–159, 2008.
- [64] W. J. Nicoll. Global oscillatory integrals for solutions of hyperbolic systems. Ph.D. thesis, University of Sussex, 1998.

- [65] F. Pfaffle. The Dirac spectrum of Bieberbach manifolds. J. Geometry Phys., 35(4):367–385, 2000.
- [66] F. Rellich. Perturbation theory of eigenvalue problems. Courant Institute of Mathematical Sciences, New York University, New York, 1954.
- [67] G. V. Rozenblyum. Spectral asymptotic behavior of elliptic systems. Journal of Mathematical Sciences, 21:837–850, 1983.
- [68] Yu. Safarov. Exact asymptotics of the spectrum of a boundary value problem, and periodic billiards. *Mathematics of the USSR - Izvestiya*, 33:553–573, 1989.
- [69] Yu. Safarov. Non-classical two-term spectral asymptotics for self-adjoint elliptic operators. D.Sc. thesis, Leningrad Branch of the Steklov Mathematical Institute of the USSR Academy of Sciences, 1989. In Russian.
- [70] Yu. Safarov. Fourier Tauberian theorems and applications. J. Funct. Anal., 185:111–128, 2001.
- [71] Yu. Safarov. Distributions, Fourier transforms and microlocal analysis, 2012.
 Lecture notes. Available online: http://tinyurl.com/n5zslmb.
- [72] Yu. Safarov and D. Vassiliev. The asymptotic distribution of eigenvalues of partial differential operators. Amer. Math. Soc., Providence (RI), 1997, 1998.
- [73] T. Sauer. Field equations in Teleparallel space-time: Einstein's fernparallelismus approach toward unified field theory. Historia Math., 33(4):399–439, 2006.
- [74] A. Scorpan. The wild world of 4-manifolds. Amer. Math. Soc., Providence (RI), 2005.
- [75] R. T. Seeley. Complex powers of an elliptic operator. In Proc. Symp. Pure Math., pages 288–307. Amer. Math. Soc., Providence (RI), 1967.

- [76] E. Stiefel. Richtungsfelder und Fernparallelismus in n-dimensionalen Mannigfaltigkeiten. Commentarii Mathematici Helvetici, 8:305–353, 1935–1936.
- [77] S. Sulanke. Berechnung des Spektrums des Quadrates des Dirac-Operators auf der Sphäre und Untersuchungen zum ersten Eigenwert von D auf 5dimensionalen Räumen konstanter positiver Schnittkrümmung. Ph.D. thesis, Humboldt Universität zu Berlin, 1981.
- [78] A. Trautman. The Dirac operator on hypersurfaces. Acta Physica Polonica B, 26:1283–1310, 1995.
- [79] A. Unzicker and T. Case. Translation of Einstein's attempt of a unified field theory with Teleparallelism. Available as a preprint: http://arxiv.org/abs/physics/0503046, 2005.
- [80] D. Vassiliev. Two-term asymptotics of the spectrum of a boundary value problem under an interior reflection of general form. *Functional Analysis* and Its Applications, 18:267–277, 1984.
- [81] D. Vassiliev. Teleparallel model for the neutrino. *Phys. Rev. D*, 75(2):025006, 6, 2007.
- [82] K. P. Wojciechowski. The zeta-determinant and the additivity of the etainvariant on the smooth, self-adjoint Grassmannian. Communications In Mathematical Physics, 201(2):423–444, 1999.